

The use of manipulative materials in early place value instruction: a cognitive load perspective

Author: Vassar, Alexandra

Publication Date: 2017

DOI: https://doi.org/10.26190/unsworks/19943

License:

https://creativecommons.org/licenses/by-nc-nd/3.0/au/ Link to license to see what you are allowed to do with this resource.

Downloaded from http://hdl.handle.net/1959.4/58682 in https:// unsworks.unsw.edu.au on 2024-05-05

The use of manipulative materials in early place value instruction: a cognitive load perspective

Alexandra Vassar

A thesis in fulfilment of the requirements for the degree of Doctor of Philosophy



School of Education Faculty of Arts and Social Sciences The University of New South Wales, Australia

September 2017

PLEASE TYPE

THE UNIVERSITY OF NEW SOUTH WALES

Thesis/Dissertation Sheet

Surname or Family name: VASSAR	
First name: ALEXANDRA	Other name/s:
Abbreviation for degree as given in the University calendar: PhD	
School: Education	Faculty: Arts and Social Sciences
Title: The use of manipulative materials in early place value education: a cognitive load perspective.	

Abstract 350 words maximum: (PLEASE TYPE)

Teaching mathematical concepts is often accompanied by the use of worked examples, and the use of manipulative materials. Worked examples have been shown to be an effective method of instruction with novice learners, as shown by higher test performance and shorter acquisition times. Often worked examples are accompanied by illustrations of manipulative materials, and the physical use of such materials. One such example is the use of Multi-base Arithmetic Blocks (MAB) to teach place value. Whilst the use of worked examples alongside illustrations of MAB in instructional material is common, their efficacy has not been investigated in young students. Using the concepts of cognitive load theory, which investigates how the learner's limited working memory and vast long-term memory can be used to efficiently design educational material, this research examines the effects of using worked examples alongside MAB to teach place value to young students. Experiment 1 examined whether it was possible to facilitate a more efficient transition from the manipulative material to the abstract concept of place value. Using the abacus and MAB, and two methods of instruction, the results of Experiment 1 showed no significant differences in student performance between the methods of instruction, or the manipulative materials. Using worked examples in Experiment 2, no significant differences in the performance of students using either the abacus or MAB, were found. Experiment 3 reduced the level of the reading comprehension in the instructional material. The control group, with no access to MAB, performed better in the post-test than the group using MAB. Experiment 4 examined whether the use of MAB produced a redundancy effect, by providing identical information in three different formats. No significant differences were found due to the complexity of the instructional material. In Experiment 5, the expertise of students with respect to the experimental materials was decreased. The non-MAB group performed better in the post-test and the delayed test, than the group using MAB. Thus, despite the widespread use of MAB in primary school to teach place value, this research suggests that the blocks may produce a redundancy effect, leading to an increased extraneous load, and negatively affecting learning.

Declaration relating to disposition of project thesis/dissertation

I hereby grant to the University of New South Wales or its agents the right to archive and to make available my thesis or dissertation in whole or in part in the University libraries in all forms of media, now or here after known, subject to the provisions of the Copyright Act 1968. I retain all property rights, such as patent rights. I also retain the right to use in future works (such as articles or books) all or part of this thesis or dissertation.

I also authorise University Microfilms to use the 350 word abstract of my thesis in Dissertation Abstracts International (this is applicable to doctoral theses only).

25/09/2017

Signature

Witness Signature

Date

The University recognises that there may be exceptional circumstances requiring restrictions on copying or conditions on use. Requests for restriction for a period of up to 2 years must be made in writing. Requests for a longer period of restriction may be considered in exceptional circumstances and require the approval of the Dean of Graduate Research.

FOR OFFICE USE ONLY

Date of completion of requirements for Award:

COPYRIGHT STATEMENT

'I hereby grant the University of New South Wales or its agents the right to archive and to make available my thesis or dissertation in whole or part in the University libraries in all forms of media, now or here after known, subject to the provisions of the Copyright Act 1968. I retain all proprietary rights, such as patent rights. I also retain the right to use in future works (such as articles or books) all or part of this thesis or dissertation.

I also authorise University Microfilms to use the 350 word abstract of my thesis in Dissertation Abstract International (this is applicable to doctoral theses only).

I have either used no substantial portions of copyright material in my thesis or I have obtained permission to use copyright material; where permission has not been granted I have applied/will apply for a partial restriction of the digital copy of my thesis o

26.09.2017

Signed ...

Date

AUTHENTICITY STATEMENT

'I certify that the Library deposit digital copy is a direct equivalent of the final officially approved version of my thesis. No emendation of content has occurred and if there are any minor variations in formatting, they are the result of the conversion to distribute the terms of the conversion to distribute the terms of terms of the terms of ter

26.09.2017

Signed

.....

Date

ORIGINALITY STATEMENT

'I hereby declare that this submission is my own work and to the best of my knowledge it contains no materials previously published or written by another person, or substantial proportions of material which have been accepted for the award of any other degree or diploma at UNSW or any other educational institution, except where due acknowledgement is made in the thesis. Any contribution made to the research by others, with whom I have worked at UNSW or elsewhere, is explicitly acknowledged in the thesis. I also declare that the intellectual content of this thesis is the product of my own work, except to the extent that assistance from others in the project's design and conception or in style, presentation and linguistic expression is acknowledged.'

25/09/2017 Date

Acknowledgements

Writing this thesis has been an enormous and emotional journey. Firstly, I would like to offer a huge thank you to my wonderful supervisor, Professor Slava Kalyuga and my co-supervisor, Emeritus Professor John Sweller. Their support and guidance have been invaluable and their encouragement for me to persevere in the face of continual non-significance ensured that I did not mope, but pick myself up and keep going!

I would like to thank all the schools that were involved in the research, including all the children and their parents who supported this work. Thank you for all your assistance and support: Emanuel Primary School, Meriden, MLC Burwood, Newington, St Spyridon College, St Catherine's Primary School, Queenwood and Barker College Junior School.

I would like to thank all my wonderful friends, who have provided their support and cheered me on through what has been a very difficult patch whilst completing this thesis. Here is looking at you Noz, Alicia, Svet, Marina, Amy, Lil, Chloé, and Jess – thank you. In addition, my wonderful mother-in-law, who continually gave up her time to provide me with baby-sitting services and editing services! Thank you Bronwyn for all your support, for the constructive criticism and for the amazing job you have done of editing this beast of a thesis!

One of my biggest supports has been my husband through this endeavor, putting up with my constant discussion of place value, of how children learn maths and my heated arguments over the discrepancies and omissions of the current mathematics syllabus. Thank you for looking after our son, Benji, whilst I wrote, and for taking care of our furry children, Jax and Juno, whilst I continued to do more writing. Without him and his support, this journey would not have been possible. I am sad that my beloved grandmother, Liza, who was so proud of me for starting on this road to doing a doctorate, will not be here to see me complete this body of work. I am sure she is watching though and as always cheering me on.

Last but not least, I would like to thank my parents, Maya and Tony, for their frequent reality checks, and support along the way. From my early years at school, they have instilled in me a love of learning that cannot be rivalled. But most importantly, I would like to dedicate this thesis to my mother, who has bravely fought and continues fighting her own battle all the while supporting me in completing this body of work. Mum, I find you inspiring beyond words and thank you for everything you have taught me and continue to teach me.

Abstract

Teaching mathematical concepts is often accompanied by the use of worked examples, and the use of manipulative materials. Worked examples have been shown to be an effective method of instruction with novice learners, as shown by higher test performance and shorter acquisition times. Often worked examples are accompanied by illustrations of manipulative materials, and the physical use of such materials. One such example is the use of Multi-base Arithmetic Blocks (MAB) to teach place value. Whilst the use of worked examples alongside illustrations of MAB in instructional material is common, their efficacy has not been investigated in young students. Using the concepts of cognitive load theory, which investigates how the learner's limited working memory and vast long-term memory can be used to efficiently design educational material, this research examines the effects of using worked examples alongside MAB to teach place value to young students. Experiment 1 examined whether it was possible to facilitate a more efficient transition from the manipulative material to the abstract concept of place value. Using the abacus and MAB, and two methods of instruction, the results of Experiment 1 showed no significant differences in student performance between the methods of instruction, or the manipulative materials. Using worked examples in Experiment 2, no significant differences in the performance of students using either the abacus or MAB, were found. Experiment 3 reduced the level of the reading comprehension in the instructional material. The control group, with no access to MAB, performed better in the post-test than the group using MAB. Experiment 4 examined whether the use of MAB produced a redundancy effect, by providing identical information in three different formats. No significant differences were found due to the complexity of the instructional material. In Experiment 5, the expertise of students with respect to the experimental materials was decreased. The non-MAB group performed better

in the post-test and the delayed test, than the group using MAB. Thus, despite the widespread use of MAB in primary school to teach place value, this research suggests that the blocks may produce a redundancy effect, leading to an increased extraneous load, and negatively affecting learning.

Table of contents

ORI	GINALITY STATEMENT	111
AC	KNOWLEDGEMENTS	IV
ABS	STRACT	VI
TAE	BLE OF CONTENTS	VIII
1	INTRODUCTION	1
2	HUMAN COGNITIVE ARCHITECTURE	4
2.1	Chapter Overview	4
2.2	Memory and modeling memory	6
2.3	Working Memory	7
2.4	Chunking Theory	12
2.5	Working Memory and Mathematics	14
2.6	Long-Term Memory	16
2.7	Schemas	18
2.8	Schema Automation	21
2.9	Cognitive Architecture of Numerical Processing	22
2.	9.1 Modeling number cognition	22
2.10	Biologically Primary and Secondary Knowledge	26

2.10.1	Numerosity	29
2.10.2	Ordinality	30
2.10.3	Counting	31
2.10.4	Simple Arithmetic	34
2.10.5	Estimation	36
2.10.6	Geometry	37
2.10.7	Biologically primary skills and the base-10 system	39
2.11 \$	Summary of Chapter	40
3 CO	GNITIVE LOAD THEORY	43
3.1 li	ntroduction	43
3.2 0	Cognitive Load Types	43
3.2.1	Intrinsic Cognitive Load	45
3.2.2	Extraneous Cognitive Load	47
3.2.3	Germane Cognitive Load	48
3.3 C	Cognitive Load Effects	50
3.3.1	Worked Example Effect	50
3.3.2	Split-Attention Effect	53
3.3.3	Redundancy Effect	57
3.3.4	Modality Effect	62
3.3.5	The Expertise Reversal Effect	65
3.4 S	Summary of Chapter	66
4 AC	QUISITION OF THE HINDU-ARABIC NUMBER SYSTEM	69
4.1 li	ntroduction	69
4.2 F	Place Value	71
4.3 T	eaching Place Value	77
4.3.1	Use of manipulative materials	77

ix

	4.3.1.1 Base-10 blocks	80
	4.3.1.2 The Slavonic abacus	82
4.4	Summary of Chapter	85
5	RESEARCH IN THE FIELD	88
5.1	Research pertinent to this thesis	88
5.2	Rationale for study	90
5.3	An introduction to the experiments in the thesis	92
6	EXPERIMENT 1 - PLACE VALUE	95
6.1	Introduction	95
6	1.1 Participants	96
6	1.2 Materials and Procedure	96
6.2	Data Analysis	101
6.3	Results	101
6.4	Discussion	103
7	EXPERIMENT 2 – THE WORKED EXAMPLE AND PLACE VALUE	108
7.1	Introduction	108
7.2	Method	109
7.	2.1 Participants	109
7.	2.2 Materials and Procedure	109
7.3	Results	111
7.4	Discussion	113

8	EXPERIMENT 3 – THE WORKED EXAMPLE AND PLAC	E VALUE
(SIN	MPLIFIED COMPREHENSION)	115
8.1	Introduction	115
8.2	Method	116
8.2	.2.1 Participants	116
8.2	.2.2 Materials and Procedure	116
8.3	Results	118
8.4	Discussion	120
9	EXPERIMENT 4 – THE WORKED EXAMPLE, PLACE VA	LUE AND
MAN	NIPULATIVE MATERIAL REDUNDANCY	122
9.1	Introduction	122
9.2	Method	123
9.2	.2.1 Participants	123
9.2	.2.2 Materials and Procedure	124
9.3	Results	126
9.4	Discussion	127
10	EXPERIMENT 5 - THE WORKED EXAMPLE, PLACE V	ALUE AND
MAN	NIPULATIVE REDUNDANCY (INCREASED EXPERTISE	AND COMPLEXITY
OF	INSTRUCTIONAL MATERIAL)	129
10.1	1 Introduction	129
10.2	2 Method	130
10	0.2.1 Participants	130
10	0.2.2 Materials and Procedure	130

xi

10.3	Results	132
10.4	Discussion	133
11	GENERAL DISCUSSION	135
11.1	Summary of key findings	135
11.2	Theoretical Implications	140
11.3	Educational Implications	142
11.4	Limitations of Study and Future Research Directions	144
11.5	Conclusion	147
REFI	ERENCES	149
APPI	ENDICES	184
Арре	ndix A: Experimental materials used in Experiment 1	184
Pre	-test	184
Pos	st-test	186
Арре	ndix B: Experimental materials used in Experiment 2	189
Pre	-test	189
Acc	quisition phase – abacus	189
Acc	quisition phase – MAB	193
Acc	quisition phase – no visuals	197
Pos	st-test	199
Арре	ndix C: Experimental materials used in Experiment 3	201
Pre	-test	201
Acc	quisition phase – abacus	201
Acc	quisition phase – MAB	205
٨٥٥	nuisition phase – no visuals	210

Post-test	211
Appendix D: Experimental materials used in Experiment 4	213
Pre-test	213
Acquisition phase – MAB	213
Acquisition phase – no visuals	216
Post-test	217
Post-test transfer	218
Delayed test	219
Appendix E: Experimental materials used in Experiment 5	220
Pre-test	220
Acquisition phase – MAB	220
Acquisition phase – no visuals	225
Post-test	226
Post-test transfer	227
Delayed test	228

LIST OF FIGURES

Figure 1: Memory Model composed of three different stores first proposed by
Atkinson-Shiffrin (1968)7
Figure 2: The working memory model as proposed by Baddeley and Hitch (1974).8
Figure 3: Updated Model of Working Memory (Baddeley, 2000)9
Figure 4: Phonological Loop (Baddeley, 1998)10
Figure 5: The functions of numerical comprehension and numerical production are
independent to each other (McCloskey et al., 1985)22
Figure 6: Cognitive Architecture for numerical processing23
Figure 7: The triple-code model (Dehaene, 1992)
Figure 8: The series of slides presented to babies in a series of experiments
determining the basic ability to count (Starkey & Cooper, 1980). H1 and H2
represent the habituation slides, whereas PH is the post-habituation slide32
Figure 9: The setup with puppets to demonstrate the innate ability to do basic
addition and subtraction in infants (Wynn, 1992)
Figure 10: Cognitive Load Theory suggests three types of cognitive load, of which
intrinsic and extraneous are additive. Germany load is representative of the
working memory resources allocated to dealing with intrinsic load
Figure 11: The additive nature of cognitive load, demonstrating an overload on the
working memory (top), a decrease in extraneous load freeing up more
resources for intrinsic load processing (middle), and optimised germane load,
allowing for optimal learning with the presence of high intrinsic load to occur
(bottom) (van Merriënboer & Sweller, 2010)48
Figure 12: An algebra example of a worked-example effect, where the problem and
solution are presented together to the learner for optimal schema formation.51

Figure 13: An example of a conventional geometry worked example, where the
diagram and the accompanying set of statements are physically separated
(Tarmizi & Sweller, 1988)53
Figure 14: Worked example that does not require mental integration of spatially
separated material (Tarmizi & Sweller, 1988)55
Figure 15: Diagram only group (P. Chandler & Sweller, 1991)58
Figure 16: Conventional group: diagram and accompanying text spatially split (P.
Chandler & Sweller, 1991)58
Figure 17: Modified - diagram and text integrated (P. Chandler & Sweller, 1991).59
Figure 18: Four types of illustrations used in the experiment (Berends & van
Lieshout, 2009)61
Figure 19: The system groups its numbers by sets of ten
Figure 20: Representing the number 6424 using the base-10 number system70
Figure 21: MAB blocks
Figure 22: Slavonic abacus
Figure 23: Abacus showing the number 37 (with its complement to a 100 of 63
clearly and easily subitised)84
Figure 24: Example of mistakes made in the representation task, converting words
to numerals
Figure 25: Mistakes showing an inability to bridge the 100's when counting off the
decade in the post-test105
Figure 26: Common mistakes made in expanding a number
Figure 27 [.] Mistakes made by children in specifying place value in numbers 107

LIST OF TABLES

Table 1: Five stages of place value understanding, as described by Ross (1989)	77
Table 2: A 2×2 mixed factorial design in Experiment 1	.96
Table 3: Stages of place value understanding in the pre-test and post-test	.97
Table 4: Descriptive statistics for each stage of place value understanding 1	102
Table 5: Descriptive Statistics for the variables1	11
Table 6: Descriptive Statistics for the variables1	18
Table 7: Descriptive Statistics for the variables1	126
Table 8: Descriptive Statistics for the variables1	32

1 Introduction

Place value, defined as the numerical value that a digit has by virtue of its position in a number (A. Stevenson, 2010), is at the core of most basic arithmetic manipulations and procedures. Understanding place value is, therefore, a fundamental building block for the future mathematical understanding and performance of students (Kamii, 1986; Moeller, Pixner, Zuber, Kaufmann, & Nuerk, 2011; Nataraj & Thomas, 2007; Schmittau & Vagliardo, 2006).

Currently, the teaching of place value concepts is often accompanied by the use of manipulative materials, such as Multi-base Arithmetic Blocks (MAB blocks), which are designed to provide a concrete representation for the number bases. However, it remains unclear whether the use of such materials increases the extraneous load on the student by requiring the student not only to understand the blocks and how to manipulate them, but also to comprehend the number concepts they are trying to represent. Herein lies a problem of dual representation, with students being required to translate between the concrete blocks and the meaning of number bases, using their limited cognitive resources. Thus, it remains unclear whether the use of MAB or other manipulative materials in early mathematics education is an efficient way to assist students in understanding place value concepts, due to the implications of cognitive load theory (Sweller, 2011). Children's understanding of the base-10 number system emerges only with formal schooling and as a direct result of teaching. However, children do possess intuitive skills that involve the ability to bundle objects into sets.

The aim of this research is to draw on notions of evolutionary educational psychology (Geary, 2008a) and on cognitive load theory and the associated effects (Sweller, van Merriënboer, & Paas, 1998), in order to examine whether certain

manipulative materials might present a more intuitive representation of number concepts for children or whether the use of such materials may negatively affect learning due to the high level of processing required by the student. The fundamental argument of this thesis is that using manipulative materials as part of instructional material may lead to the redundancy effect, which occurs when identical information is presented to the learner simultaneously, but in different formats. This could result in a negative impact on learning. In exploring this argument, worked examples were used as an instructional method. These have been shown to improve performance and learning outcomes when used with novice learners. Worked examples are also commonly used in mathematics education, and, therefore, provide a familiar format to students. Additionally, using worked examples in the instructional material ensured a consistent method of intervention across the different student groups.

The second chapter presents a discussion of human cognitive architecture, with consideration given to working and long-term memory concepts. Cognitive number processing is also discussed, as pertinent to the subject matter of this thesis. The chapter concludes with a general overview of evolutionary biological primary and secondary skills, as proposed by Geary (2008a) in his contribution to the body of knowledge of cognitive load theory.

Chapter 3 provides a general overview of cognitive load theory, including a discussion of the three types of cognitive load and of the potential measurement mechanisms for each. The chapter concludes by discussing the cognitive load effects relevant to this thesis.

Chapter 4 discusses the acquisition of the Hindu-Arabic number system, by explaining the basis of the system and the instructional materials currently used in place value mathematics education.

The following chapter, Chapter 5, provides a brief overview of the literature relevant to this thesis. It presents a general overview of the series of experiments conducted as part of this thesis, which investigated the efficiency of teaching place value concepts with manipulative materials, from the perspective of cognitive load theory.

The experimental data and the results of these experiments are presented in Chapters 6-10. The results partially support the hypothesis and provide some evidence to question the use of MAB blocks as an aid in teaching place value concepts.

Chapter 11 summarises the major findings of the study, taking into consideration the concepts and effects of cognitive load theory. The chapter also considers the limitations of this study and proposes future research directions. Both theoretical and practical implications for teaching place value, with the assistance of manipulative materials, are also presented in this chapter.

2 Human Cognitive Architecture

2.1 Chapter Overview

The system of cognitive architecture is defined by Langley, Laird and Rogers (2009, p. 141), as something that specifies the "...underlying infrastructure for an intelligent system...". Similarly, human cognitive architecture refers to the structure and organisation of our cognitive resources that allows humans to integrate, process and use information (Sweller, 2011; Sweller et al., 1998). Human cognitive architecture is composed of memory structures that are fundamental to the human ability to think, learn and solve problems. Current research proposes two vital memory structures, working memory and long-term memory, as the basis of human cognitive architecture (Baddeley, 1998, 1999, 2000; Baddeley & Hitch, 1974). Understanding how these memory structures function can assist with designing the most effective instructional material.

Working memory is essential for conscious thought, but is only able to process a small number of elements at the one time, with the number often placed at 7 ± 2 (Miller, 1956). The limited working memory capacity can be improved through a process known as chunking, in which related elements are grouped together in longterm memory to form meaningful structures. Long-term memory can hold an unlimited number of elements on a permanent basis (Sweller, 2011). Storage in longterm memory occurs through the use of cognitive constructs, known as schemas, where related information is stored together for easy retrieval when required.

Cognitive load theory refers to the total amount of cognitive load required in working memory to complete a specific task. The processing capacity of each individual differs, based on the extent of their knowledge held in long-term memory in regards to a specific task. Learners who have more in-depth knowledge and understanding of a particular task are considered to be experts. Conversely, novices do not possess in-depth knowledge and understanding of the same task. This means that the cognitive load of an expert is often much lower than that of a novice when completing the same task. Catering for both expert and novice learners is an important consideration in the design of effective instructional material.

The theory has recently been expanded to take into consideration an evolutionary perspective of the development of human cognitive architecture (Geary, 2002, 2005, 2008a; Geary & Bjorklund, 2000; Muller, 2010; Paas & Sweller, 2011; Sweller, 2004, 2008). One of the primary goals of evolutionary educational psychology is to uncover inherent biases to learning, and to use knowledge of those biases for instructional design and reform. In his evolutionary framework, Geary (2008a) draws a clear line between two types of skills and knowledge, primary and secondary. Knowledge that is acquired effortlessly, and that humans have evolved to acquire naturally, is referred to as primary knowledge. This is in contrast to secondary knowledge, which requires specific instruction. Geary (1996) suggests six areas of primary mathematical knowledge: numerosity, ordinality, counting, estimation, basic arithmetic and geometry. He believes that building on an evolutionary knowledge can be more effective than using non-evolutionary instructional theories (Geary, 2007; Muller, 2010).

Currently, early primary school education is not making full use of children's primary spontaneous skills to facilitate the acquisition of secondary knowledge (Geary, 2008a; Muller, 2010). Educators should be more mindful of primary intuitive knowledge and how this knowledge may be exploited to help acquire secondary knowledge more effectively (Keil, 2008). The use of physical materials in the classroom could present one way of exploiting this primary knowledge. Physical materials, commonly referred to as manipulative materials, are widely used in the classroom to help students understand abstract mathematical concepts. Using such materials is inherently a primary skill. Children may use the materials intuitively in their play in order to learn. Mix, Huttenlocher and Levine (2002, p. 130) state that

"...working with concrete materials serves to help children bridge the gap between conventional symbols and their pre-existing concepts. This suggests that educators should place greater emphasis on the connections between symbols and experiences rather than simply providing the experiences themselves.". Whilst some manipulative materials may be a very useful tool in the classroom, they must be used correctly. For this to occur, children must have a basic understanding of the concept being taught and not simply play with the manipulative material (Smith, 2008). For this reason, research is still equivocal as to whether there is a clear and consistent benefit in using manipulative materials in the classroom rather than more traditional methods of teaching (Bartolini Bussi, 2011; Kaminski, Sloutsky, & Heckler, 2008; Sowell, 1989; Uttal, Scudder, & DeLoache, 1997; Wright, 2013).

2.2 Memory and modeling memory

An understanding of human cognitive architecture, composed of memory structures such as sensory memory, working memory and long-term memory, is essential in establishing the fundamentals of learning, problem solving and instructional design. A number of different memory models have been put forward. One of the initial propositions defining a memory model made up of multiple stores, was developed by Atkinson and Shiffrin (1968). It suggests a flow of information through the system in the following sequence (Figure 1). Incoming information enters the sensory register (sensory memory) and resides there for a brief period of time. From sensory memory, information flows onto the working memory store, where it can also only be kept for a short period of time. Through a process called rehearsal, information can then be moved into long-term memory. The long-term store is a permanent repository for information and cannot be easily accessed, although it allows for the retrieval of information from the long-term memory to the short-term memory, as required. Information that does not undergo the process of rehearsal is lost over a short period of time.



Figure 1: Memory Model composed of three different stores first proposed by Atkinson and Shiffrin (1968)

The Atkinson-Shiffrin model (R. C. Atkinson & Shiffrin, 1968) was one of the most influential memory models of its time, in part because it laid out a comprehensive view of information processing in the memory system. However, this is a fairly simplistic model of human cognitive architecture, and many improvements have been made to this model over the years. In particular, there has been a change in thinking about short-term memory function, and this shift is reflected in the use of the term working memory, instead of short-term memory, to describe what was once considered to be an extremely brief, unitary store.

2.3 Working Memory

Building on the initial research by Atkinson and Shiffner (1968), a working model of memory was proposed by Baddeley and Hitch (1974). It assumed that short-term memory is not only a unitary store, but it can also provide a workspace for complex cognitive activity. Baddeley and Hitch (1974) showed that working memory was composed of three basic parts: the central executive, the visuo-spatial sketch pad, and the phonological loop. The central executive is the driver of the overall system, assigning data to either of the two other components. It is also responsible for mental arithmetic and problem solving. The visuo-spatial sketchpad is where information presented in a visual or spatial format is stored and processed. Finally, the

phonological loop deals with spoken and written material, consisting of two further components: the phonological store, dealing with speech, and the articulatory control linked to speech production (Figure 2).



Figure 2: The working memory model as proposed by Baddeley and Hitch (1974)

Further research resulted in the addition of the fourth component to the working memory structure, the multimodal episodic buffer, which comprises a limited and temporary storage of information held in a multimodal code (Baddeley, 2000) (Figure 3). The episodic buffer is also controlled by the central executive, which is responsible for amalgamating information from multiple sources into a unitary episode. The buffer serves as a modeling space, separate from the long-term memory, but an important stage in long-term episodic learning (Baddeley, 2000).



Figure 3: Updated Model of Working Memory (Baddeley, 2000)

The central aspect of the working memory model developed by Baddeley and Hitch (1974) is the ability of working memory to act as more than just a temporary storage space. According to the model, working memory is a dynamic construct, a system consisting of two temporary storage facilities, the first to process visual information, and the second to process auditory information, and, in addition, a system to control these stores. The central executive unit is the system that controls both visual and auditory processing. This is the component that most strongly differentiates current thinking about working memory from historical views of shortterm memory. The central executive is responsible for deciding what information is stored and where the information should be stored. Additionally, it is responsible for any integration, coordination and cognitive manipulation of the information held in the separate stores. This component determines how to maximize the cognitive resources available, and how to suppress irrelevant information that would overuse the resources (Baddeley, 1998). The model further differentiates between the processing of phonological and visual information. Audio-based information directly enters the phonological store, a passive storage facility within the phonological loop, and is held there until it decays a couple of seconds later. Written material must first be converted to articulatory code before it can enter the phonological store, and can

also be stored there for up to two seconds. The second part of the phonological loop, the articulatory control, rehearses the information from the phonological store and circulates the information until it is remembered. The articulatory control is also responsible for converting written material into audio code and transferring it into the phonological store (Baddeley, 1999) (Figure 4).



Figure 4: Phonological Loop (Baddeley, 1998)

The visuo-spatial sketchpad, a second temporary memory store, allows visual and spatial information to be held so that it can be processed. Research has shown that the visuo-spatial sketchpad can operate simultaneously with the phonological loop, allowing both visual and auditory information to be processed without one affecting the efficacy of the other (Cornoldi & Vecchi, 2004). Sweller (2004) postulates against the existence of the central executive unit, and the lack of empirical evidence provides a strong indication against its existence. Sweller (Sweller, 2004) postulates that the central executive is not a general biological structure, but rather a specific learned structure, not available when processing novel information. He proposes that schemas, in long term memory, act as a central executive function for the working memory. If such a schema is not available, as is the case with novel information, learners attempt a series of random combinations followed by tests to establish the effectiveness of those combinations, which can act as a substitute for the central executive function (Sweller, 2003, 2004).

However, the model of working memory, consisting of two different storage buffers for visual and audio information, is strongly supported by dual-task studies, which require the concurrent performance of two tasks that could otherwise be performed separately (Baddeley & Hitch, 1974). In specific examples, verbal concurrent tasks affect the storage of verbal information, but not visuo-spatial information, whereas visuo-spatial tasks affect the storage of visuo-spatial information but not auditory information (Cocchini, Logie, Sala, MacPherson, & Baddeley, 2002; Olive, 2003).

Novel information enters working memory via sensory memory and is then processed in the working memory structure. However, the ability to process complex information is hindered by the limited capacity of working memory. Miller (1956) suggests that most adults can only store in their short-term memory between five and nine (seven plus or minus two) items that have not been previously learnt, and then only for a brief duration. This limit also applies to working memory, if information storage or maintenance only is involved; if processing is also involved then this limit could be further reduced. If information needs to be processed or manipulated in any manner, the number of elements that can be kept in working memory could be as few as two or three (Cowan, 2001). If not rehearsed, novel information can only be held in working memory for about 15-30 seconds (Driscoll, 2005). These limitations have important implications for instructional design and the way new information is

taught to students. It is important that new information is taught in ways that compensate for the limited capacity and duration of working memory. Instructional design that requires students to process many elements of complex novel information at the same time will lead to failure in the learning of that material (Sweller, 2002).

Information that has already been learnt and is stored in long-term memory is available for processing only after it is first retrieved and transferred temporarily into working memory. Unlike information entering from sensory memory, information retrieved from long-term memory is not hindered by the limited processing capacity of working memory. Since information in the long-term memory is stored in an organised form, there may be no limits to the amount of such information (Sweller, 2002). Consequently, there are further implications for instructional design based on whether students have prior knowledge of the material being taught. Students who have prior knowledge of new material that has already been organised in their longterm memory will be able to process new information very quickly, as opposed to those students who are hearing or seeing this information for the very first time.

2.4 Chunking Theory

When discussing storage limitations of short-term memory, Miller (1956) suggested seven plus or minus two as the maximum number of chunks, or groups, that a person can hold. Chunking, initially proposed by Miller (1956) and de Groot (1965), and theorized by Chase and Simon (1973), is an important mechanism in circumventing working memory limitations and increasing the ability to hold more information than a maximum of seven singular elements. Chunking involves combining together related pieces of information according to familiar patterns, in order to form a single chunk, which is then treated in working memory as one element. One of the most commonly used examples of chunking is the way phone numbers are grouped. A number such as 3412975837 might be hard to remember when each of the ten

numbers has to be remembered separately, but when chunked into three smaller singular units, such as 3412 975 837, it becomes possible and easier to remember, especially if some of the units contain familiar combinations of numbers (e.g., associated with some well-remembered dates, codes, et.).

More recent research has focussed on the mechanisms of chunking. This research has cast some doubt over whether chunks are stored in working memory or in our long-term memory (Gobet, 2000; Guida, Gobet, Tardieu, & Nicolas, 2012). Gobet (2000) proposed that some of the information encoded as chunks is stored in long-term memory. A study by Ericsson, Chase and Faloon (1980) found that an average undergraduate college student, who had been engaged in performing memory span tasks for more than 230 hours, was able to increase his memory span from 7 digits to 79 digits. The student utilised a mnemonic technique to chunk the numbers into familiar structures. The use of mnemonics allowed the student to relieve the burden placed on his working memory through an association with the already existing knowledge of sports records in his long-term memory. For example, 3492 was recorded by the student as "3 minutes and 49 point 2 seconds, near worldrecord mile time" (Ericsson et al., 1980). The study led Ericsson and his colleagues (1980) to develop their theory of skilled memory and to conclude that, whilst it is not possible to increase the capacity of working memory with extended practice, it is possible to increase the memory span with the use of chunking, based on the related associations in long-term memory.

To attain exceptional memory performance, prior knowledge must be used to encode and store items in long-term memory for later retrieval (Ericsson, Delaney, Weaver, & Mahadevan, 2004). With practice and rehearsal, the encoding and retrieval processes can be greatly sped up, causing dramatic improvements in memory performance. This is consistent with skilled memory theory and its generalization to long-term working memory (Ericsson et al., 2004). Recent studies have further confirmed that at least some of the information contained in chunks is stored in long-term memory, due to the absence of the effect of an interfering task (Ericsson & Kintsch, 1995; Gobet & Simon, 1996; Kintsch, 1998). The study by Ericsson et al. (2004) proposed that some chunking information was stored in the long-term memory. Their experiment looked at a person with superior memory abilities, and found that the subject's encoding techniques were a result of memorizing the first 10,000 digits of π as numerous 10-digit groups, consistent with the chunking information residing in long-term memory (Ericsson et al., 2004). In the case of a domain expert, a process meant to be occurring solely in working memory can also involve storage and retrieval of chunks from long-term memory (Guida et al., 2012). Such studies further demonstrate that some chunking information can be stored in long-term memory, thereby greatly improving memory skills by overcoming the limited capacity of working memory.

2.5 Working Memory and Mathematics

Considerable research conducted in the last decade suggests that working memory plays a significant role in mathematical cognition (Alloway, Gathercole, Willis, & Adams, 2004; Ashcraft & Krause, 2007; Berg, 2008; Bull & Scerif, 2001; Geary, 1990; Holmes & Adams, 2006; Passolunghi, Vercelloni, & Schadee, 2007; Siegel & Ryan, 1989; St Clair Thompson, Stevens, Hunt, & Bolder, 2010; Swanson & Sachse-Lee, 2001; van der Sluis, van der Leij, & de Jong, 2005; Zheng, Swanson, & Marcoulides, 2011). Raghubar, Barnes, and Hecht (2010) noted that "...the very nature of many mathematical tasks would seem to require or at least be supported by working memory...". Indeed, research has shown that children with a higher working memory capacity tend to learn early mathematics more efficiently (Hoard, Geary, Byrd-Craven, & Nugent, 2008; Passolunghi, Mammarella, & Altoè, 2008; Passolunghi et al., 2007; Passolunghi & Siegel, 2001, 2004).

Hecht (2002) proposed that children use different strategies to solve maths problems, some of the strategies drawing heavily on working memory resources. Research by Lee, Ng and Ng (2009) used a multitask approach to examine whether working memory is associated with generating a model of the problem and then solving the problem. They tested 255 students, all of them 11-years-old, using algebraic problem solving tasks, and found that differences in working memory capacity accounted for about a quarter of the variance in building a model of the problem and then solving that problem. Literacy, however, accounted for another 10% of variability in algebraic problem solving skills (K. Lee et al., 2009). This is consistent with other research by Zheng, Swanson and Marcoulides (2011) who proposed that whilst all working memory components play significant roles in children's ability to solve mathematics problems, reading skills can compensate for some of the gaps in working memory measures. Conversely, Swanson and Beebe-Frankenberger (2004), using a large sample of first, second and third graders, found that in the area of word problems, working memory predicted solution accuracy independent of reading ability. A review of literature by LeFevre et al. (2005) supports the notion that working memory is heavily involved in problem solving, when the problems reach sufficiently high levels of complexity.

There is, however, some debate as to the importance of the different components of working memory to the various mathematical skills (Ashcraft, 1992; Fürst & Hitch, 2000; Hecht, 2002; Raghubar et al., 2010). Meyer et al. (2010) suggested that the central executive and the phonological loop are vital in facilitating mathematical learning in the early stages (e.g., at Grade 2 level), whereas the visuo-spatial sketchpad plays an increasingly vital role in later stages of learning mathematical skills. Another study that is consistent with these results (Passolunghi et al., 2007) tested 170 children at the beginning and at the end of Year 1, approximately aged 6-7 years. From the results, the research similarly proposed that, in the early stages of mathematical learning at school, working memory and, specifically, the central executive unit, is a distinct and significant predictor of success. The study found no link to phonological ability as a predictor of mathematical success (Passolunghi et al., 2007).

In contrast to the above findings, other studies have found that school-aged children rely heavily on the phonological loop for subvocal rehearsal, ensuring accuracy during mental calculations involved in arithmetic problem solving (Logie, Gilhooly, & Wynn, 1994). Using a large sample of 8- to 11-year-olds, Adams and Hitch and Donlan (1998) found that the phonological loop is essential for the ability to solve arithmetic problems. The role of the visuo-spatial sketchpad is not as clear, although in more recent work, Holmes and Adams (2006) found that both the visuo-spatial sketchpad and the central executive, but not the phonological loop, were excellent predictors of curriculum-based mathematical success in early numerical development.

2.6 Long-Term Memory

Long-term memory is an organised store of information, altered through the acquisition of knowledge and it is widely accepted that an unlimited number of elements related in meaning can be stored together in long-term memory on a permanent basis (Baddeley, 1999; Ericsson & Kintsch, 1995; Sweller, 2004).

A series of experiments by De Groot (1965) examined the differences between chess grand masters and novice chess players. The chess grand masters were able to win almost all games, not because of their superior reasoning skills, but due to the fact that they were familiar with, and held in their long-term memory, the largest number of various chess board compositions from real game situations and their corresponding best moves. In this way, they were able to easily predict, with their prior experience, the best move from a large number of various board setups. A computer model built to replicate this process estimated that a chess grand master holds between 10,000 and 100,000 chess board configurations in his or her longterm memory (H. A. Simon & Gilmartin, 1973).

Later studies by Simon and Chase (1973) extended the experiment to include both real game positions and random positions. Their findings indicated that in random positions, chess grand masters performed on par with novice players, whilst in real game positions, chess grand masters performed significantly better in choosing their next move. This further reinforced De Groot's findings, linking longterm memory to higher-level cognitive activities such as problem solving and thinking (Chi, Glaser, & Rees, 1982; Sweller, Ayres, & Kalyuga, 2011). These findings have been replicated in areas other than chess, such as algebra (Sweller & Cooper, 1985), physics (Chi, Feltovich, & Glaser, 1981), and mechanics (Reif & Heller, 1982). Thus, it is widely accepted that expertise is heavily dependent on knowledge held in long-term memory (Sweller, 2002).

Tulving (1972) made one of the earliest distinctions between episodic, semantic and procedural long-term memories. Episodic memory is that part of longterm memory responsible for storing episodes in our lives associated with a certain time and place (Tulving, 1972). An example of episodic memory is recalling a city through memories of the last holiday taken there. Semantic memory is that part of long-term memory responsible for storing general information about the world, for example, basic knowledge of countries, continents and oceans, or knowledge required for the use of language. The above two types of knowledge are involved in conscious thought, if activated and retrieved into working memory. On the other hand, procedural memory does not require conscious thought and is that part of long-term memory responsible for motor skill memory, for example, the ability to ride a bicycle. The different types of memory differ greatly in the kind of prior experience involved: episodic and procedural memories require prior experience, whilst semantic memory does not (Greenberg & Verfaellie, 2010). Cohen (1980) was able to demonstrate a distinction between declarative and procedural knowledge in experiments with patients suffering from amnesia. Declarative knowledge is concerned with facts, and the recollection of this type of knowledge requires some degree of consciousness. Semantic and episodic memory structures are responsible for storing declarative knowledge. Conversely, procedural knowledge is based on motor skills and is automated; it does not require conscious thought for recollection. Whilst amnesiac patients have trouble in acquiring new episodic or semantic information after the onset of amnesia, their knowledge of procedural information remains largely unaffected, and they are able to acquire new procedural skills (N. J. Cohen, 1980).

2.7 Schemas

Information in long-term memory is organised and stored in domain-specific structures, known as schemas. In their research, Rumerlhart and Ortony (1977) identified four essential characteristics of schemas, which work together to establish the efficient building blocks of memory. The first characteristic is that schemas have variations in their values, based on the context and environment of a specific situation. To demonstrate this point, Rumelhard and Ortony (1977) used the example of GIVE as a schema with three variables: a giver, a gift and a recipient. On different occasions, different variables will take on different values, whilst the relationships internal to the GIVE schema will remain constant. Another example is COMPUTER as a schema, which can include the motherboard, the shell, the graphics card, memory and RAM. Each of these variables can vary largely in their dimensions or other characteristics, but the schema itself remains consistent in that these variables are present.

Secondly, schemas can embed one within the other (Rumelhart & Ortony, 1977). To use the previous example of a COMPUTER schema, one of the elements, such as the shell, can form a sub-schema consisting of plastic, screws, clips, etc. This sub-schema is embedded within the dominating schema of the COMPUTER. There are certain advantages to embedding schemas, primarily the ability to look at the 'big picture' without the need to delve deeper into the internal structures. A second advantage is that a deeper understanding can be achieved if reference is made to the internal structure of the lower-level elements. This means that the COMPUTER schema can remain a simple schema with just the main elements, and
without the need to examine every screw and clip holding elements in place, that is, until these lower-level structures need to be accessed.

The third characteristic is the ability of schemas to represent all levels of abstraction. This refers to the fact that schemas relate not only to concrete objects, such as the computer example used previously, but also to feelings and intangible concepts (Rumelhart & Ortony, 1977).

Finally, the fourth characteristic of schemas is their ability to represent actual knowledge rather than simply dictionary-based definitions and to "...represent knowledge in the kind of flexible way which reflects human tolerance for vagueness, imprecision, and quasi-inconsistencies..." (Rumelhart & Ortony, 1977, p. 111).

Schemas are engaged at all stages of cognitive processing. They not only allow the categorization of information based on similar elements, but also determine how that information will be used, thereby facilitating retrieval at a later stage when the information is required (Sweller et al., 1998). Basically, schemas are responsible for facilitating the flow of information through the cognitive processing system (Rumelhart, 1980) and are referred to as the "...building blocks of cognition..." (Rumelhart, 1980). Schemas help to reduce cognitive load by allowing a person to ignore any irrelevant information that would otherwise unnecessarily burden their working memory (Sweller & Chandler, 1994). For example, when one sees a dog, a schema allows the person to recognise the animal simply as a 'dog', despite there being many breeds of dogs. Because this information is stored in a schema in long-term memory, working memory does not need to be burdened with the different breed-specific elements presented in a single dog.

Although schemas are stored in the long-term memory, their construction occurs in working memory, where new material is first manipulated. Retrieval of relevant schemas from long-term memory depends on three factors, firstly, the strength of the information stored within the schema, secondly, the extent to which the incoming information matches with the information currently stored within the schema, and, finally, how recently and frequently the necessary schema has been used (Thorndyke & Hayes-Roth, 1977).

Schemas play an important role in increasing the capacity of working memory. Complex schemas, which store many elements of related information, can be treated in working memory as a single entity. This results, firstly, in an increased memory capacity and, secondly, in the ability to overcome the limitations of working memory that apply to new material with no associated schemas (Kirschner, Sweller, & Clark, 2006). Experts possess more complex and developed schemas in their long-term memory (van Merriënboer & Ayres, 2005) than non-experts, which is one of the factors differentiating a novice from an expert learner. This could be due to the fact that domain expertise has to be acquired slowly over years of deliberate practice (Ericsson, Krampe, & Tesch-Römer, 1993). As an example of this, a chess expert requires at least ten years of consistent and continuous practice to be able to store tens of thousands of board configurations in their long-term memory (H. A. Simon & Gilmartin, 1973). Another example is an algebraic schema, which allows a domain expert to easily transform $\frac{a}{b} = c$ into a = bc. Despite the fact that there are three elements and a set of relationships between them, this equation can be easily transformed by anyone with basic algebraic knowledge, due to the fact that the elements and relations are grouped into, and stored in, a single schema in long-term memory. Schema acquisition is directly related to the capacity of working memory and also to whether the cognitive load resulting from the material being learnt is in line with the cognitive capacity of the learner (Verhoeven, Schnotz, & Paas, 2009). Once acquired, schemas make processing in working memory a relatively simple task, because a schema acts as only one element, in contrast to the several elements present in the initial algebraic equation (Pollock, Chandler, & Sweller, 2002). The construction and automation of schemas are essential to the process of gradually transforming novices into experts (Verhoeven et al., 2009).

2.8 Schema Automation

Once a schema has been constructed, it needs to be practiced in order to promote automation (Kalyuga, Ayres, Chandler, & Sweller, 2003; van Merriënboer & Sweller, 2005). Schema automation occurs when information stored in schemas can be processed automatically with no conscious effort, bypassing the need to be processed within the working memory structure. To successfully achieve schema automation, a task requires much practice and repetition, and an automated schema can only develop if the task is consistent across a range of different problem situations (Schneider & Shiffrin, 1977; van Merriënboer & Sweller, 2005). An example of task consistency is the use of software applications, such as word processing tools, where the user makes use of the same functionality, located in the same area of the application, and produces the same result each time, allowing for maximized efficiency in task performance due to schema automation. In an early experiment looking at the effects of automation, Schneider and Shiffrin (1977) showed that the ability to differentiate letters from numbers is an automated skill in an adult. Approximately 2100 trials were required for participants to be able to differentiate between letters and letters as fast as they could differentiate between letters and numbers.

Automation allows familiar tasks to be performed quickly and efficiently without mistakes, and is an important element of instructional design. Efficient instructional design should, therefore, encourage both schema construction and schema automation (van Merriënboer & Sweller, 2005), as both these processes are essential for successful learning, "...Cognitive mechanisms of schema acquisition and transfer from consciously controlled to automatic processing are the major learning mechanisms and foundation of our intellectual ability and skilled performance..." (Kalyuga, 2010, p. 61).

2.9 Cognitive Architecture of Numerical Processing

Number competence in humans begins in early infancy and develops over time to encompass more complex number processing and manipulation skills (Geary, 1995). Various models of numerical cognition have been proposed, and have caused much debate (Jamie I. D. Campbell & Epp, 2004).

2.9.1 Modeling number cognition

McCloskey, Caramazza and Basili (1985) proposed a basic model of the cognitive architecture for numerical processing. The model examined three cognitive mechanisms involved in number processing: number comprehension, number production, and the execution of simple calculations. The model considered the functions of numerical comprehension and numerical production to be independent of each other (Figure 5). Basically, the model proposes a modular presentation of number cognition: comprehend numbers first, then perform number manipulations , and finally produce the result. An expanded model with all the components included is shown in Figure 6.



Figure 5: The functions of numerical comprehension and numerical production are independent to each other (McCloskey et al., 1985)



Figure 6: Cognitive Architecture for numerical processing (McCloskey et al., 1985)

Numeric comprehension and numeric production elements also distinguish between numbers spoken or written in alphabetical form, e.g. *one hundred and thirty-two* and numbers written in digital form, e.g. *132*. Within both forms of comprehension and production there is a distinction between lexical and syntactic processing components. Lexical processing involves the comprehension or production of individual elements in a number, such as the digit 3 or the word three (McCloskey et al., 1985). Syntactic processing involves the processing of the relationships amongst the different elements in order to produce the number as a whole. For example, using the number 132 again, lexical processing requires the understanding of the separate digits 1, 3, and 2 in the number, whereas syntactic processing ensures that the relationship between the positions of the digits determines the complete numeral. The model assumes a single 'abstract' semantic quantity code that is common to all

the components (Jamie I. D. Campbell & Epp, 2004). Therefore, as an example, when a person is presented with a mathematical problem, the input (digital or verbal) needs to be translated into an abstract code, then the problem needs to be solved, and finally, the abstract code translated back into an output code (digital or verbal) (García-Orza, León-Carrión, & Vega, 2003). Accordingly, in mathematics, difficulties with reading numbers aloud could be independent from actual errors in calculations. Thus, this model assumes that mathematical processes, such as number production, number estimation and memory for number facts, operate independently of the way in which numerals are represented (written, verbal or Arabic). However, Campbell & Epp (2004), in their study examining the encoding approach in Chinese-English bilinguals, have argued against this model and have proposed that there are indeed mechanisms for format specific number judgements and calculations (Jamie I. D. Campbell & Epp, 2004; Dehaene, 2004).

In contrast to the abstract code model (McCloskey et al., 1985), an encodingcomplex hypothesis has been proposed (Jamie I. D. Campbell, 1994). The hypothesis is based on empirical evidence of the existence of format-specific retrieval in number processing (Bernardo, 2001; Blankenberger & Vorberg, 1997; Jamie I. D. Campbell, 1994; Jamie I. D. Campbell & Epp, 2004; Sciama, Semenza, & Butterworth, 1999). According to this model, numerical skills are based on two factors: the modality of the representation, as opposed to abstract code, and the number processing tasks that are required and which involve common cognitive mechanisms. This means that number skills would be based on multiple forms of internal representation of number and realized in a variety of different ways (Thevenot & Barrouillet, 2006). Based on this hypothesis, another model, proposing a different view of number cognition, has been presented by Dehaene (1992), and is referred to as the triple-code model (Figure 7). The triple-code model proposes that number processing is based on three types of code: visual Arabic form, an auditory-verbal code, and an analog-magnitude representation (Dehaene, 2004). The triple-code model is, therefore, modular in respect to the representational code, in contrast to the abstract mode model, which is modular in respect to numerical function. The auditory-verbal code facilitates verbal number input and output, some counting processes, and the retrieval process for simple addition and multiplication facts. García et al. (2003) considers the multiplication tables to be the best examples of using verbal code for calculation, because the tables would be retrieved as automatic verbal associations, due to the rote learning of tables. The visual digital form mediates input and output, parity judgements and certain multi-digit operations, whereas the analog-magnitude form provides the basis for number comparisons and approximate calculation and estimation, and is also thought to contribute to subitising capabilities (Jamie I. D. Campbell, 1994; Dehaene, 1992). This model, unlike the abstract code model, assumes that the form in which the numbers are presented could affect numerical manipulation and calculation (Jamie I. D. Campbell & Epp, 2004). For example, extensive work with Arabic-based digits would develop an Arabic-specific inhibitory process, which would reduce reading-based interference in arithmetic (Jamie I. D. Campbell & Epp, 2004).



Figure 7: The triple-code model (Dehaene, 1992)

This research maintains consistent representations of number in its instructional material and uses only familiar systems of Arabic numerals and their associated place value. This was done to reduce any negatory preffects on numerical manipulation and calculation required within the proposed tasks.

2.10 Biologically Primary and Secondary Knowledge

Early numerical competence in humans has only recently become the subject of research in developmental psychology. Previously, this field was dominated by the theories of Piaget, who argued that young children do not possess any conceptual understanding of number and largely undertake activities such as counting by rote (Geary, 1994c). The notion of arithmetic in the first year of human life was unthinkable (Dehaene, 1997). Recently, David Geary (Geary, 1994a, 1994d, 1996, 2008a) proposed an evolutionary-based framework that accounted for early

numerical skills. This framework can assist in the understanding of biological and cultural influences on children's cognitive and academic development. Geary's theory proposes the presence in humans of an innate understanding, on a basic level, of psychology, biology and physics (folk domains), which ensured evolutionary human survival. Human folk domains comprise a set of innate motivations for learning, which can provide a framework for formal education to build upon. The ability to suppress the intuitive knowledge that comes with the human folk domain ensures that humans are able to create and learn evolutionary novel information (Geary, 2008a). This leads to the presence of two forms of knowledge: biologically primary knowledge, that is, evolutionary novel knowledge that is acquired beyond the intuitive innate motivations.

Biologically primary knowledge is the knowledge that we are born with, or that is easily acquired and which has undergone selection pressure. Biologically primary abilities are evolved abilities that allowed our ancestors to survive. It is this intuitive knowledge that allows humans to effortlessly learn how to speak, despite the high motor complexity of the task. This kind of knowledge requires minimal instruction and is thought to be acquired with very little effort.

In contrast, biologically secondary knowledge is acquired through explicit instruction and effort. This knowledge is harder to acquire and relies on the culture and environment of the child, and is introduced to deal with novel ecological problems that our ancestors never faced. It involves such skills as reading, writing, and higher mathematics. The cognitive processes associated with the acquisition of secondary knowledge rely heavily on conscious processing of information in working memory. Cognitive load theory, therefore, primarily deals with instructional aspects of the acquisition of secondary knowledge (Sweller, 2011).

This distinction between primary and secondary knowledge has direct consequences for instruction both in the classroom and outside. First, due to their evolved mechanisms, biologically primary abilities are universal and are typically the same in all normally developed humans across cultures and nationalities (Geary, 1995). The acquisition of biologically secondary abilities depends on the culture and motivation of the child. Whilst biologically primary skills are endogenous and do not require much external motivation, biologically secondary skills require external reinforcement. Children must often be coerced into practicing biologically secondary skills, such as mathematics (Geary, 1994d). Learning about children's extrinsic motivations can help to develop better strategies to teach evolutionary novel or biologically secondary knowledge.

The ability to rely on more primary forms of learning, intuitive to children, with subtle guidance from teachers, could prove to be effective in teaching novel concepts. It is, therefore, vital that we examine what these primary forms of learning are and how these can be used in the classroom. In mathematics education, of particular interest is the relationship between primary systems and the learning of the base-10 number system used globally, which Geary (2007) believes is essential for modern mathematics. Resnick (1984, p. 126) has stated that "...the initial introduction of the decimal system and positional notation system based on it is, by common agreement of educators, the most difficult and important instructional task in mathematics in the early school years...". One such difficulty in teaching place value occurs due to the conceptual nature of the number structure and the difficulty in knowing what students are thinking with regard to the numbers.

Biologically primary knowledge is composed of many different skills, in the fields of both numeracy and literacy. This research focusses only on numeracy skills. Geary (Geary, 1994a) suggested six potential categories for biologically primary mathematical abilities, clustered together in a coherent numerical domain. Each of these categories emerges early in childhood, and evidence shows that these biologically primary abilities are gender neutral (Geary, 1996; E. S. Spelke, 2005). These are:

 Numerosity – the ability to determine small sets of items without actually counting them;

- ii. Ordinality basic understanding of ordinal relationships in terms of more than or less than;
- iii. Counting there appears to be a preverbal counting system for very small subsets of numbers, up to perhaps three or four items;
- iv. Simple Arithmetic a sensitivity to increases or decreases in number groups, again this sensitivity is limited to small subsets
- v. Estimation inexact estimation of magnitude or size
- vi. Geometry implicit understanding of shapes and spatial relationships

The following sections will examine each of the mathematical categories for biologically primary skills in more detail.

2.10.1 Numerosity

Recent studies have provided sufficient evidence to show that children are indeed born with core cognitive skills in mathematics that provide some basic numerosity understanding (Dehaene, 1997; Geary, 2008a). Numerosity, or subitising, refers to the ability to recognise small quantities without the need to count. Subitising allows for the exact recognition of up to four items by infants and plays an important role in numerical development (Kroesbergen, Van Luit, Van Lieshout, Van Loosbroek, & de Rijt, 2009; Peucker & Weißhaupt, 2013). In fact, Baroody (1986) views subitising as a fundamental skill in children's further development of number concepts. Adults and some animals are able to discriminate and estimate numerosities larger than three or four (Dehaene, 1997; Xu & Arriaga, 2010). However, the accuracy of estimation is dependent on the geometrical pattern of the items to be subitised. For instance, the number of objects tends to be overestimated when they are consistently spread out, and underestimated when they are inconsistently distributed, a function of our visual system parsing the objects into small groups (Dehaene, 1997; Frith & Frit, 1972; Ginsburg, 1976). The ability to quickly and accurately determine small numerosities is evident in infants in their first week of life (Geary, 1996). By six months of age, and from at least four months of age, infants are able to perceive and represent small numbers of objects (S. E. Antell & Keating, 1983; Mandler & Shebo, 1982; Starkey & Cooper, 1980; Strauss & Curtis, 1981; Van Loosbroek & Smitsman, 1990).

In a number of experiments by van Loosbroek and Smitsman (1990), infants were visually habituated to a static display of a small number of items, from one to four. After habituation, infants were presented with a different display each varying in the number of items. Typically, collections would differ in their colour, density and configuration. In their study, van Loosbroek and Smitsman (1990) showed that infants would focus longer on displays containing different numbers of items, rather than on the new displays of the habituated number. These results indicate infants' sensitivity to changes in numerosity and an early perception of subitising small numerosities.

2.10.2 Ordinality

Ordinality describes the basic understanding of ordinal relationships, that is, discriminating between number relationships that are 'greater than' or 'less than'. Research has shown that basic ordinality is evident in 18-month old infants (Strauss & Curtis, 1981). In one of the first experiments testing only for ordinal relationships, Cooper (1984) habituated infants to successively presented pairs of displays. Each pair maintained a constant ordinal relationship, with varying absolute values between trials, the values ranging from 1 to 4. On habituation, infants were always shown a small number followed by a larger number, or vice versa. Then, during the experiment, infants were shown a series of relationships that were presented in one of three ways: the same way as the habituation, opposite to the habituation or with equal numerical relationships, but failed to look longer at the reversed relationship to the habituation. Conversely, 14- to 16-month-old infants looked longer at both the equal numerical relationships and at the reversed relationships to the habituation (R. G. Cooper, 1984). The results of this study could indicate a basic

understanding of ordinality in infants, in particular infants older than 12 months of age. However, the results were not controlled for surface area of displays and elements shown on displays (Brannon, 2002).

Two more recent studies have found evidence of basic ordinality at ten months of age (Feigenson, Carey, & Hauser, 2002) and at eleven months of age (Brannon, 2002). In their experiment, Feigenson et al. (2002) had 10- to 12-monthold infants choose between two quantities of crackers. The infants watched an experimenter sequentially hide two different quantities of crackers in either of two baskets, first using one cracker in one basket and two in the other, and then two crackers in one basket and three crackers in another basket. With these choices, infants always chose the larger quantity of crackers. However, when the numbers were increased in a 3 v 4 ratio, 2 v 4 ratio, and 3 v 6 ratios, the infants chose randomly among the quantities (Feigenson et al., 2002). This study has been reproduced with similar results (Siegler & Opfer, 2003; Xu & Arriaga, 2010), showing that infants possess an innate rudimentary understanding of ordinality, both in phylogeny and ontogeny (Brannon, 2002).

2.10.3 Counting

There appears to be an initial set of principles that allows children to count some small subsets of numbers, up to three or four, before knowledge of number words can develop. One of the first experiments that established that babies could recognise small numbers was conducted by Starkey and Cooper (1980). In these series of experiments, a number of slides were presented to a total of 72 babies, aged between 16 and 30 weeks. The slides contained either two or three dots, arranged in different patterns and with a variety of spreads (Figure 8).



Figure 8: The series of slides presented to babies in a series of experiments determining the basic ability to count (Starkey & Cooper, 1980). H1 and H2 represent the habituation slides, whereas PH is the post-habituation slide.

The babies were habituated with a slide containing the same number of dots until such time as each of the babies started losing interest in this repetitive stimulus. However, once a different slide with three dots were presented to the babies, the fixation time, 1.9 seconds prior to the switch, changed to 2.5 seconds on the first different slide, showing that the baby is able to detect a change in number, and therefore have a very basic principle of counting (Starkey & Cooper, 1980). A few years later, Antell and Keating (1983) demonstrated a similar result, but with newborns. Their experiments demonstrated that babies are able to discriminate between small numerosities even a few days after birth. Another series of experiments conducted by Strauss and Curtis (1981) used arrays of commonly observed objects as opposed to just dots to demonstrate that the ability to count to small numbers does not rely only on object recognition, but also on the numerosity of the image itself. The objects pictured in the photographs were all different,

photographed at different angles and from various distances. None of this variation in object representation modified the behavior of babies, who were able to notice the change in number. Using moving objects, van Loosbroek and Smitsman (1990) demonstrated that infants in the first few months of life were able to count objects even in a moving environment.

All of the above experiments relied heavily on the visual perception of babies. To investigate the ability of babies to discriminate between numerosities associated not just with changes in visual stimuli, a follow-up experiment by Starkey, Spelke and Gelman (1983) was designed to take into account audio sequences. In this experiment, 6-, 7- and 8-month old babies were provided with two visual stimuli from two separate projectors. One of the slides would show two common objects, whereas the second slide would show three objects randomly arranged. Simultaneously, the baby was presented with an audio sound of a drumbeat. The results of this study demonstrated that after habituation, the babies were more attentive to those slides that matched the numerosity of the drumbeats. This experiment implied that numerical representation in infants is not a result of visual or auditory perception (Dehaene, 1997). Two recent studies (Izard, Sann, Spelke, & Streri, 2009; Izard & Spelke, 2009), further confirmed the ability of infants to discriminate between small numerosities. Their experiment first introduced auditory sequences with a fixed number of syllables, followed by images of the same and different numbers of items. The results of the study showed that newborns focused their gaze consistently longer on displays that were in ratio 3:1 with the auditory representation during the habituation stage of the experiment (Izard et al., 2009). As in the previous study, babies were able to differentiate between varying numerosities, providing further evidence for abstract numerical representation early in life (Izard et al., 2009).

2.10.4 Simple Arithmetic

Some research suggests that 5-month old infants are aware of basic addition and subtraction, in particular of the effect of adding or subtracting one item from a small set of items. Wynn (1992) used puppets to demonstrate basic addition and subtraction. An infant was shown one puppet on a stage, a screen was then put up and a second puppet placed behind the screen, thereby producing a simple '1 + 1 = 2' equation. In one of the displayed outcomes ("impossible outcome"), infants were never shown the two puppets together, and when the screen was lowered, only one puppet could be seen on stage (Figure 9 demonstrates the setup used in the experiments).

Sequence of events 1+1 = 1 or 2

1. Object placed in case





3. Second object added









Sequence of events 2-1 = 1 or 2



Figure 9: The setup with puppets to demonstrate the innate ability to do basic addition and subtraction in infants (Wynn, 1992)

The time spent focusing on the stage was measured and compared for the possible outcome, i.e. 1 + 1 = 2 and for the impossible outcome, i.e. 1 + 1 = 1. The results indicated that infants spent on average a second longer looking at the impossible outcome (1 + 1 = 1) in comparison to the possible outcome (1 + 1 = 2). The experiment then looked at the subtraction component of simple arithmetic, using the same setup as for addition. It was found that infants spent as much as three seconds

longer looking at an impossible outcome (2 - 1 = 2), than the possible outcome (2 - 1 = 1). The study has been replicated with 2- and 3-year-olds (Houdé, 1997), and further evidence exists from other studies, demonstrating that infants are concerned not by the variability in the object being added or subtracted, but rather by the impossibility of some arithmetic operations (Kobayashi, Hiraki, Mugitani, & Hasegawa, 2004; Koechlin, 1997; McCrink & Wynn, 2004; T. J. Simon, Hespos, & Rochat, 1995). A study by Kobayashi (2004) used both auditory and visual modalities, and the results suggested that infants formed expectations about numerosity behind a screen based on both the visual stimuli presented and the auditory tones that they heard, thereby again replicating and further extending on Wynn's (1992) initial findings regarding innate arithmetical abilities.

2.10.5 Estimation

The hypothesized primary mathematical ability of estimation refers to the ability to approximate quantities, magnitudes or sizes. Estimation is a process that many children find difficult within the confines of the school curriculum, but this process is essential for future mathematical success (Siegler & Opfer, 2003). Current research suggests that the number representations found in human infants depend on a mechanism for estimating, but not determining, exact numerosities (Xu & Spelke, 2000). Prior to the development of language and number words, a language-independent representation of number magnitude is used for quantity manipulation and estimation (Dehaene, Spelke, Pinel, Stanescu, & Tsivkin, 1999; McCrink & Wynn, 2007). In fact, research using brain imaging evidence (Dehaene et al., 1999) suggested that estimation relies on non-verbal visuo-spatial cerebral networks, in line with infants' developing language skills.

Xu and Spelke (2000) attempted to directly test whether infants are capable of representing approximate numerosities, a direct function of estimation. Their study found that, provided the difference between two numerosities is sufficiently large, infants are able to distinguish between an 8- and a 16-element display, when all other variables of the display are controlled. However, when the difference between numerosities was reduced, infants could no longer discriminate between the two. This result is in line with prior research by Starkey and Cooper (1980), which also found that infants were unable to discriminate four from six elements. These findings also complement an experiment conducted by Wynn and Bloom (1999) studying infant ability to enumerate a collection of elements. This research was replicated using auditory stimuli as well, with similar results (Lipton & Spelke, 2004).

Further expanding on this work, a more recent study by McCrink and Wynn (2004) examined ratio abstraction in 6-month-old infants. Infants were presented with multiple examples of the same ratio during the habituation phase, using images of Pac-man and pellets. They were then presented either with new ratios or with the habituated ratio, and were successfully able to discriminate two ratios that differed by at least a factor of 2. The results of this study support the concept of an approximate magnitude-estimation system in infants (McCrink & Wynn, 2004). Sections *2.10.1* Numerosity and *2.10.3* Counting looked in more detail at an infant's ability to infer small numerical subsets and estimate quantities and magnitudes. The next section describes the ability to estimate sizes of objects.

2.10.6 Geometry

Core geometry knowledge refers to the implicit understanding of shapes and spatial relationships, and is one of the hypothesized primary mathematical abilities. Indeed, sensitivity to geometric information is essential for the development of mathematical knowledge, including identifying objects and establishing locations (Dehaene, Izard, Pica, & Spelke, 2006). Over the last few decades, research has found that human infants are sensitive to length, angle and direction in visual format (S. E. G. Antell & Caron, 1985; Bomba & Siqueland, 1983; Izard & Spelke, 2009; McGurk, 1972; Schwartz, Day, & Cohen, 1979; Slater, Mattock, Brown, & Bremner, 1991; E. Spelke, Lee, & Izard, 2010). Slater et al. (1991) investigated infant sensitivity to angles only a few hours after birth, and their results suggested that infants were indeed sensitive

to angle variations. A more recent study showed that 5½-month-old infants were capable of using geometric cues to differentiate different corners of an isosceles triangle, implying a basic understanding of angles (Lourenco & Huttenlocher, 2008). Similar experiments also demonstrated that infants were sensitive to variations in length (Newcombe, Huttenlocher, & Learmonth, 1999) and, specifically, could differentiate lines of different lengths (Slater, Mattock, & Brown, 1990). Sensitivity to length in infancy was observed even after objects had changed orientation or position (Izard & Spelke, 2009).

A number of experiments have also revealed that infants are capable of extracting object shape even from partly occluded objects (Kellman & Spelke, 1983; Slater, Morison, et al., 1990). Schwartz and Day (1979) demonstrated that infants could recognise differences between squares and rectangles, even when the orientation or the angular relationship of the shape had changed. Expanding on this work, Bomba and Siqueland (1983) used a series of dot patterns placed together to form shapes, to demonstrate that infants are also capable of distinguishing a triangle shape and treating it as a different object from a rectangle.

Baillargeon and Graber (1987) demonstrated that infants are able to distinguish between the sizes of different objects. In this experiment, infants were presented with short and tall objects, which moved behind a screen with a window in it, allowing infants to see the taller objects moving, but not the shorter objects. Results of the study showed that 5½-month-old infants looked significantly longer if the window in the screen showed a shorter object moving through, but not a taller object, indicating sensitivity to the size of the objects. Expanding on these experiments further, Baillargeon (1991; 1987) used a block hidden behind a screen, which was rotated in an arc formation in relation to the block. In a possible case, the screen was rotated only until it reached the top of the block, while in an impossible case, the screen was rotated unhindered by the block, either in a full 180° arc or partway through the top of the block. The results showed that, whereas 6½-month-old infants looked longer at both impossible events, 4½-month-old infants only

looked longer at one of the impossible events, the 180° arc rotation. This indicated a difference in the way height is distinguished by infants of different ages (Lourenco & Huttenlocher, 2008), consistent with earlier findings by Huttenlocher, Newcombe and Sandberg (1994) showing infant sensitivity to height.

2.10.7 Biologically primary skills and the base-10 system

A better way to design instructional material that involves secondary skills is to base such material on the foundation of children's intuitive primary skills. However, whether base-10 instructional material can be designed giving consideration to such intuitive skills has not yet been explored in depth. Geary (2007) believes that in the case of place value, or the base-10 system, the secondary mathematics is too far removed from the primary skills to be easily and intuitively learnt. Explicit instruction is required if these concepts are to be attained. Similarly, Sweller (Sweller, 2012), mentions that it is an error to assume that the difficulty in acquiring secondary knowledge can be alleviated or eliminated by using techniques that are appropriate for primary knowledge. It might therefore not be possible to eliminate the dificulties that children often face when learning the base-10 number system, but this issue does require further research. However, materials that are designed on the foundations of primary skills could provide a more efficient transition by learners to secondary skills.

There are some primary abilities that can potentially underlie the attainment of the base-10 system. For example, infants and young children implicitly organise collections of objects into sets; however, the sets are not clustered around the 10 set. One of the prime examples of this, often used by children as a basic learning strategy, is counting with fingers, grouped into two groups of 5. The primary understanding of the number line, and the decomposition of the number line into sets of ten, with the sets organised into clusters, may therefore facilitate the learning of the base-10 system.

Additionally, the ability to physically interact with the environment at large is a

natural primary ability, and could thus be helpful in acquiring secondary skills. Children are inherently motivated to engage in play activities, facilitating the development of primary skills (R J Sternberg & Ben-Zeev, 2012). However, it is not clear whether such activities would be sufficient for the acquisiton of secondary skills. In the arena of the base-10 system, manipulative materials are often used as physical objects that may allow for an easier translation to abstract mathematical concepts. Tools such as blocks and other counting objects are frequently used in both free play and organised play activities to establish deeper understanding of place value concepts. However, Geary (2008b) mentions that whilst free play is an important component of education, and is one of the most effective ways for building on pre-exisiting evolutionary based cognitive structures to enhance primary skills, it does have instructional limitations. Formal direct instruction is still the most effective approach in teaching complex secondary knowledge (Geary, 2008b; Sweller, 2012). The base-10 system is considered to be biologically secondary knowledge. Hence, it is a skill that requires specific instruction. However, the level of understanding of the concept of place value can vary greatly. One factor that can affect understanding and subsequent learning of place value is the first language of the student. Research shows that children from Asian language backgrounds find it easier to comprehend number concepts, compared to children from European-derived languages (Miura, Okamoto, Kim, Steere, & Fayol, 1993). This is largerly due to the direct correspondence between the number word for values greater than 10, and the place value of that number (Fuson, 1990; Miura & Okamoto, 1989; Miura et al., 1993).

Predominantly, however, children's knowledge of the base-10 system, globally, emerges in a formal school setting and only as a result of teaching practices that are designed to impart this knowledge to the child.

2.11 Summary of Chapter

This chapter provided a summary of human cognitive architecture, taking into

account the evolutionary framework proposed by Geary (Geary, 1996, 2000, 2008b). Human cognitive architecture is composed of two distinct memory constructs: working memory and long-term memory. Working memory is where conscious and effortful processing occurs. This structure is limited in its capacity and can only hold a small number of elements simultaneously (Miller, 1956). This is in contrast to longterm memory, which appears to be limitless in its capacity and is the basis of the human ability to solve complex problems. The limitless capacity of long-term memory is due to the acquisition and subsequent storage of cognitive constructs, known as schemas, in long-term memory. Schema acquisition and schema automation are critical for efficient and fluent learning. Their importance to learning should be considered in instructional design. The automation of schemas allows their efficient and effortless retrieval from long-term memory, without putting further burden on the limited capacity of working memory.

An evolutionary-based framework for learning was proposed by Geary (Geary, 1996, 2000, 2008b), drawing a distinction between primary skills, which are effortless to learn and can be acquired spontaneously, and secondary skills, which require effort and direct instruction for acquisition. A series of mathematical biologically primary skills has been hypothesized by Geary (1994a), and includes:

- i. Numerosity as the ability to determine small sets of items without actually counting them;
- ii. Ordinality, the basic understanding of ordinal relationships;
- iii. Counting, a pre-verbal counting system for small subsets of numbers;
- iv. Simple arithmetic, a sensitivity to basis increases or decreases in small numbers;
- v. Estimation, inexact estimation of magnitude or size; and
- vi. Geometry, a basic sensitivity to different shapes, angles and spatial relationships.

These primary skills are spontaneously learnt and often appear in infanthood. During formal schooling, primary skills are built upon to acquire secondary skills. It is vital

that children's spontaneous primary abilities are considered when designing instructional material, as this can lead to more efficient and fluent learning models.

Chapter 3 will discuss cognitive load theory and how it aims to improve instructional design, in order to encourage a more efficient facilitation of schema acquisition and automation (Sweller et al., 1998).

3 Cognitive Load Theory

3.1 Introduction

Cognitive Load Theory (CLT) is based on the assumption that human cognitive architecture consists of a limited working memory interacting with an unlimited long-term memory. The theory proposes that instructional design be developed on this basis, with consideration given to the implications of limited working memory (P. Chandler & Sweller, 1991; Sweller, 1989, 2002, 2004; Sweller et al., 2011).

One of the most important elements in the learning process is schema construction and automation, a process that occurs in the working memory. Accordingly, one of the main goals of instruction is to aid in the successful construction and automation of schemas in working memory. Work within the cognitive load framework has largely concentrated on the design of instructional methods that make efficient use of working memory (Paas, Renkl, & Sweller, 2003). Instructional methods developed on the basis of CLT have proven successful in a variety of different fields (Paas, Tuovinen, Tabbers, & Van Gerven, 2003; Sweller, 1999, 2004; van Merriënboer & Ayres, 2005). In contrast, instructional materials that do not take CLT into account have proven to be less efficient, requiring more time and mental effort for the learning process to take place.

3.2 Cognitive Load Types

The ease with which information can be processed in working memory depends on the cognitive load imposed by the task. Cognitive Load Theory proposes three different types of cognitive load: intrinsic, extraneous and germane cognitive load. Working memory may, therefore, be affected by intrinsic cognitive load, based on the nature of the material itself, extraneous cognitive load, based on the way in which the material is presented, and germane load, which is the load imposed by the actual learning process. Intrinsic and extraneous cognitive loads are considered to be additives. This means that if there is low intrinsic load, then a high extraneous load might not make a difference to the learning process, due to the overall cognitive load being within the boundaries of working memory capacity. When added together, they provide the overall cognitive load on the working memory during a task. In order for material to be efficiently learned, the amount of cognitive load cannot exceed the working memory capacity available for learning (Figure 10). This theory drives the development of alternative instructional formats and methods.



Total Working Memory Capacity

Figure 10: Cognitive Load Theory suggests three types of cognitive load, of which intrinsic and extraneous are additive. Germane load is representative of the working memory resources allocated to dealing with intrinsic load.

Research suggests that learner understanding and performance improve if extraneous load is reduced, in particular when intrinsic load is high (P. Chandler & Sweller, 1991; Kirschner et al., 2006; Richard E Mayer & Moreno, 2003; Sweller et al., 1998; van Merriënboer & Ayres, 2005; van Merriënboer & Sweller, 2005, 2010; Verhoeven et al., 2009).

The next section will examine the three types of cognitive load in more detail.

3.2.1 Intrinsic Cognitive Load

Intrinsic cognitive load is measured by the number of elements that must be processed simultaneously and their interconnectedness to each other, commonly referred to as element interactivity, within the context of the instructional material (Sweller et al., 1998; van Merriënboer & Sweller, 2005). Element interactivity depends on the difficulty of the material that needs to be learnt and the expertise of the learner (Gerjets & Scheiter, 2003; Kalyuga, 2011; Sweller et al., 1998). The inherent complexity of the material that needs to be learnt is, therefore, directly related to the amount of intrinsic cognitive load placed on the working memory.

Tasks that are low in element interactivity, that is, which do not contain many elements that need to be amalgamated, but rather elements that can be learnt in isolation to each other, impose a relatively low load on the working memory (Ayres, 2006). This is in contrast to high element interactivity material, requiring users to deal simultaneously with several complex elements that cannot be learnt in isolation. High element interactivity causes high cognitive load and can reduce the capacity of working memory available for optimal schema construction (van Merriënboer & Sweller, 2005, 2010).

Intrinsic cognitive load is hard to control. It cannot be altered by instructional intervention, without actually modifying the material that needs to be learnt, which is not a feasible solution in the majority of cases (Ayres, 2006; van Merriënboer & Sweller, 2010). However, intrinsic load is essential in learning and understanding instructional material, and, therefore, it is vital to accommodate this load without exceeding working memory capacity (Kalyuga, 2011).

Traditionally, intrinsic cognitive load has been regarded as independent of the specific instructional design methods (Kalyuga, 2011), making it difficult to manage. One strategy for managing intrinsic cognitive load is to manipulate learning material based on the individual learner's level of expertise. As expertise develops in a particular domain, it is possible to decrease the intrinsic cognitive load, due to the fact that the task interactions become incorporated into schemas and become

automated (Ayres, 2006). Gerjets, Scheiter and Catrambone (2006) argue that there are instructional-based manipulations that can be carried out to achieve a reduction in intrinsic cognitive load and suggest using a modular-example approach, focussing on the role of smaller individual meaningful units of the overall problem. Additionally, Kalyuga (2011) provides the example of achieving a reduction in intrinsic cognitive load by either substituting the original material with a simpler task, or by removing complex elements from the original material during the initial stages of learning.

Another strategy to reduce the amount of intrinsic cognitive load was first explored by Pollock et al. (2002), and involved teaching isolated elements to begin with, before combining the elements together to teach the complete concept. Clarke, Ayres and Sweller (2005) similarly demonstrated better information transfer in the learning of spreadsheet skills, by firstly teaching novice spreadsheet users how to use spreadsheets, thereby isolating those elements, and only then attempting mathematical tasks. Consistent with these results, Ayres (2006) examined how intrinsic cognitive load can be reduced in algebra. His results found that learners showed improvements in algebra performance when elements of the material were isolated and then learnt prior to the material being presented as a whole (Ayres, 2006). Furthermore, Ayres (2006) demonstrated that low-expertise learners benefit from an initial reduction in interactivity when learning mathematical tasks. Conversely, high-expertise learners benefit from an initial high-level interactivity (Ayres, 2006), a factor attributed to the decrease in germane cognitive load (See 3.2.3 Germane Cognitive Load for more information). Both these studies relied on improved schema acquisition through prior learning. Previous studies also demonstrated the efficiency of prior training (a pre-training method) in reducing intrinsic cognitive load, with learners being provided with specific prior knowledge before the intended materials were learnt (Clarke et al., 2005; Richard E Mayer, Mathias, & Wetzell, 2002; Richard E Mayer & Moreno, 2003).

In their research, Mayer and Moreno (2003) discussed a better information transfer when students first had to study the components making up a system before

being introduced to the system as a whole and its operation. Clarke et al. (2005) similarly found that the learning process was enhanced when students were provided with prior training in relevant elements before being introduced to the full element interactivity. These strategies do not directly affect the intrinsic cognitive load of a particular task, but they do manipulate the task to reduce its complexity in the initial learning stages.

3.2.2 Extraneous Cognitive Load

Cognitive load theory and many of the cognitive load effects were initially developed in an attempt to reduce or eliminate the amount of extraneous cognitive load in tasks with a high level of intrinsic cognitive load (Kalyuga, 2011). In contrast to intrinsic cognitive load, which is generated by the instructional material, extraneous cognitive load is generated by the manner in which the instructional material is presented to the learner. Extraneous cognitive load is detrimental to learning. It is caused by inappropriately designed instructional material that fails to take into consideration human cognitive architecture and the limited capacity of working memory, and is not essential in achieving instructional goals (Sweller & Chandler, 1994). For example, the presentation to learners of redundant information, the processing of which requires extra working memory resources, can impact the overall capacity available for processing the required instructional information.

Most cognitive load effects (see *3.3* Cognitive Load Effects for more information) arise from the need to reduce extraneous cognitive load, and have been demonstrated to exist only where high intrinsic cognitive load is present (R. Carlson, Chandler, & Sweller, 2003; Sweller, 2002). Instruction designed to reduce extraneous load has no effect on learning simple material, or material with low intrinsic load. However, for teaching complex material, it is essential that extraneous cognitive load is kept to a minimum, freeing up valuable memory resources to deal with intrinsic cognitive load, thereby inducing germane load to achieve the goals (van Merriënboer & Sweller, 2010) (Figure 11)



Figure 11: The additive nature of cognitive load, demonstrating an overload on the working memory (top), a decrease in extraneous load freeing up more resources for intrinsic load processing (middle), and optimised germane load, allowing for optimal learning with the presence of high intrinsic load to occur (bottom) (van Merriënboer & Sweller, 2010)

3.2.3 Germane Cognitive Load

The concept of germane cognitive load was added into cognitive load framework at a later time by Sweller et al. (1998). Germane cognitive load is considered to be an effective or positive load, as working memory resources are engaged in processing intrinsic cognitive load, leading directly to learning through schema acquisition and automation. Van Merriënboer and Sweller (2005, p. 152) suggest that "...Effective instructional methods encourage learners to invest free processing resources to schema construction and automation, evoking germane cognitive load..." Interestingly, it has been found that an increase in effort or motivation can also lead to an increase in cognitive resources allocated to a particular task, and this can directly translate to germane cognitive load, if the resources are being used in schema acquisition and automation (Paas, Renkl, et al., 2003; van Merriënboer & Sweller, 2005).

Cognitive load theory aims to optimise germane load in order to enhance the process of learning (Figure 11, bottom). A number of design guidelines for instructional material have been proposed, aiming to optimise germane load in

learning. Germane cognitive load was first proposed to help account for the effects in variability of instructional materials (Sweller et al., 1998). Variability in problem content and context, as applied to problem conditions, encourages learners to acquire and construct schema (van Merriënboer & Sweller, 2005). Problem variability ensures that when learners are confronted with similar problems, they are able to identify similar features and distinguish relevant information required to solve the problem. Initially, high variability increases cognitive load during practice, but thereafter it leads to schema construction and transfer of learning (van Merriënboer & Sweller, 2005). Another kind of possible variability known as contextual interference, refers to the interference resulting from practising various problem situations within the same context of practice. For example, if a series of problems performed one after another rely on exactly the same set of skills, then the contextual interference is low. Conversely, if a series of problems performed one after another rely on different skill sets, i.e. different problems presented to the learner in a random practice schedule, then interference is high. Van Merriënboer et al. (2002) have reported a number of studies that examine the role of context interference. Research suggests that a high context interference stimulates the construction of schema (van Merriënboer et al., 2002). However, for high context interference to be effective, learners are required to mentally integrate a large quantity of newly acquired knowledge during practice (van Merriënboer et al., 2002).

The use of self-explanations is another design guideline that aims to optimise germane load. This guideline states that worked examples should be enriched with prompts for self-explanations from learners. In their study of probability based problems, Renkl and Atkinson (2003) used worked examples that guided learners by asking them which probability rule was applied during each step within the solution. They found a strong effect on transfer test performance for learners who received self-explanation prompts in comparison to the learners who did not (Renkl & Atkinson, 2003), suggesting the efficiency of the self-explanation principle in inducing germane load.

49

The introduction of germane cognitive load into the cognitive load framework (Sweller et al., 1998) was not based on empirical evidence. It was used as a way to account for the intentional cognitive effort required for learning to take place, and the associated demands on working memory. More recently, it has been postulated that germane cognitive load is not an independent source of cognitive load, but rather pertains to the working memory resources associated with intrinsic cognitive load (Kalyuga, 2011; Sweller, 2010). This new conceptulisation aims to remove the germane resources', such as variability in practice or self-explanation, associated with allocating working memory resources to learning (Kalyuga, 2011; Leppink, Paas, Gog, Vleuten, & Merriënboer, 2014; Sweller, 2010).

3.3 Cognitive Load Effects

3.3.1 Worked Example Effect

The worked example effect was initially described in relation to cognitive load theory by Sweller and Cooper (1985), and it is probably one of the most recognised of the cognitive load theory effects (Sweller, 2011). The effect describes the ability of learners to learn more by studying a problem and its solution, as opposed to just being presented with the problem alone (Figure 12).

$$\frac{5x}{6} + 15 = 30$$
$$\frac{5x}{6} = 30 - 15$$
$$\frac{5x}{6} = 15$$
$$5x = 15 \times 6$$
$$5x = 90$$
$$x = \frac{90}{5}$$
$$x = 18$$

Figure 12: An algebra example of a worked-example effect, where the problem and solution are presented together to the learner for optimal schema formation

In their research, Sweller and Cooper (1985) compared two groups of students with a limited knowledge of algebraic transformations, i.e. problems that require students to change the subject of a formula. One group was involved with conventional problem solving techniques, whereas the second group was given a series of problems, with the first problem in each series having a worked example. The results of the study (Sweller & Cooper, 1985) suggested that, in a situation where the students were required to complete four pairs of similar problems, the worked example group required significantly less time than the conventional problem solving problem and made significantly fewer errors in the test phase, when both groups were subjected to a number of problems similar to those encountered in the acquisition phase (Sweller & Cooper, 1985). This initial research seemed to indicate the ability of worked examples to facilitate the acquisition of schemas (Sweller & Cooper, 1985).

Further research has shown that for novices, worked examples are more effective for learning and transfer, with less mental effort expended in the process of learning (R. K. Atkinson, Derry, Renkl, & Wortham, 2000; Rourke & Sweller, 2009; van Gog, Kester, & Paas, 2011; van Gog & Rummel, 2010). Conversely, learners with a higher level of expertise no longer required worked examples, and research has found that, for such learners, worked examples could be ineffective and even have a negative effect on learning when compared to problem solving (Kalyuga et al., 2003; Kalyuga, Chandler, Tuovinen, & Sweller, 2001). Conventional instruction that consists only of presenting the learner with the problem forces that learner to resort to problem solving strategies, such as means-ends analysis, where the learner is constantly searching for an operator to reduce the difference between the problems state and the goal state (Sweller, 1988). Despite being able to arrive at the correct solution to a problem using means-ends analysis, it remains a weak strategy, due to the high extraneous load imposed on working memory. The high extraneous load does not contribute to schema acquisition, and therefore reduces learning efficiency. The worked examples strategy allows the learners to focus their limited working memory on studying each step of the solution and thereby developing a schema to assist in solving similar problems in the future (Sweller & Cooper, 1985).

Another study showing the benefits of worked examples was conducted by Zhu and Simon (1987). In this study, the researchers found that a three-year mathematics curriculum could be completed in two years if a worked examples strategy was utilised, as opposed to conventional problem solving. Further to this, the study found that students in the worked examples groups performed slightly better in subsequent tests than the conventional problem solving group (Zhu & Simon, 1987).

Schema acquisition, using the worked-examples strategy, can go beyond the ability to solve the specific type of problem; it has been shown that general rules can be abstracted from examples, and parts of the solutions can be adapted to other problems (J. R. Anderson & Fincham, 1994; J. R. Anderson, Fincham, & Douglass, 1997; G. A. Cooper & Sweller, 1987; Rourke & Sweller, 2009). The worked examples effect can therefore be a powerful tool in improving the efficiency of instructional design, leading to improved learning.

3.3.2 Split-Attention Effect

Section 3.3.1 Worked Example Effect reviewed the benefits that worked examples can provide to the design of instructional material and to learning. However, many worked examples consist of two or more sources of mutually referring information (P. Chandler & Sweller, 1992). For a worked example to be effective, it needs to be presented in such a way as to take into consideration the limited capacity of working memory. One cognitive load phenomenon that needs to be of concern when presenting two or more sources of mutually referring information is the split-attention effect. Split-attention effect refers to the need to mentally integrate information, which is physically presented apart. A common example of the effect can be seen often in the area of geometry, where a diagram and a set of statements are presented to the learner (Figure 13).



Find $\angle ABC$ (1) $\angle ECD = \angle ACB = 68^{\circ}$ (vertically opposite angles) (2) $\angle BAC + \angle BCA + \angle CBA = 180^{\circ}$ (angle sum of a triangle) $45^{\circ} + 68^{\circ} + \angle CBA = 180^{\circ}$ $113^{\circ} + \angle CBA = 180^{\circ}$ $\angle CBA = 180^{\circ} - 113^{\circ}$ $\angle CBA = 67^{\circ}$

Figure 13: An example of a conventional geometry worked example, where the diagram and the accompanying set of statements are physically separated (Tarmizi & Sweller, 1988)

Mental integration of information presented in such a way requires the learner to search and match each statement with the diagram. In accordance with cognitive load theory, this process of searching and matching text to diagram has the same cognitive repercussions as searching for operators to solve a problem during meansends analysis (P. Chandler & Sweller, 1992). Both cases require a process that is not essential to learning, such as searching and matching, which uses up the limited cognitive capacity available for learning. For example, in a study of the effectiveness of worked examples, Tarmizi and Sweller (1988) investigated the use of worked examples in the area of circle geometry. The nature of content, in particular geometry or science-based content, often requires both a diagram and a set of descriptive statements, with both elements being essential for learning. Results showed that there was no significant difference in the time taken to solve the presented problems between the worked examples group and the control group (Tarmizi & Sweller, 1988). In fact, in the acquisition phase, students in the worked examples group spent the same amount of time on the question solutions as the conventional problemsolving group, and in the test phase the worked-examples group spent longer on the questions than the control group with equivalent scores on all questions rewritten (Tarmizi & Sweller, 1988). Other research has confirmed the hypothesis that worked examples requiring mental integration of physically separate information are no more effective than conventional problem-solving strategies (Berends & van Lieshout, 2009; Kalyuga, Chandler, & Sweller, 1999; Ward & Sweller, 1990; Yeung, Jin, & Sweller, 1997). Therefore, for optimal learning, the worked example used in Figure 13, should be presented without the need to mentally integrate physically separated information (Figure 14).


Figure 14: Worked example that does not require mental integration of spatially separated material (Tarmizi & Sweller, 1988)

However, to transition from a spatially separated format to a format that integrates both sources of information requires more than just a change in spatial proximity (P. Chandler & Sweller, 1992). Indeed, in order to integrate a set of statements into a diagram, the text needs to be segmented and the relevant segments positioned in close proximity to the related element in the diagram, thereby labeling elements of that diagram. In previous work, it has been demonstrated that segmentation of learning material facilitates learning (Clark & Mayer, 2011; Richard E Mayer, 2005). It would appear that the structure of the text is able to guide the learner's attention during reading. Segmentation of text is able to provide learners with information about which text elements hold meaning, thereby facilitating a more efficient learning process (Florax & Ploetzner, 2010). Florax and Ploetzner (2010) examined whether the split-attention effect in instructional material is the result of physical integration of text and image, or whether it is due to the labelling of the

image and segmentation of text to accomplish this physical integration. The results of the study replicated the effects of split-attention in retention of material, with the only significant effect on learning outcomes due to text segmentation (Florax & Ploetzner, 2010). This provides evidence for the positive effect of segmented material to learning.

Split-attention effect occurs when a combination of images and text are spatially separated. The effect has been replicated in several other situations, including computer manuals, where a user is required to read a paper manual to learn the function of particular menu options before looking at the screen to see the necessary menu options on the device itself (Sweller & Chandler, 1994). The users must split their attention between the device and/or program itself, and the manual and then mentally integrate these two sources, in order to learn how to use the new device or programme. Chandler and Sweller (1996) found that for instructional formats involving high element interactivity, a self-contained manual that physically integrated the necessary pieces of information was superior to other instructional formats involving continual interaction with the computer program.

Huff, Bauhoff and Schwan (2012), examined the split-attention effect in learning tasks requiring comparison and mental integration of two pictures. Building on previous work (Huff & Schwan, 2010), Huff et al. (2012) examined troubleshooting tasks, where learners are required to find an error in a mechanical device and compare it to a working reference device or a picture in a manual of a working reference device. The research used vexing images for the first group, presenting two pictures in such a way that learners were able to switch between the images just by moving their heads. This meant that learners in the 'vex mode' group were able to still see the second representation, just by moving their head but keeping their eyes fixed on the same point. The second group was given the images on two separate screens, requiring them to split their attention between the two screens, positioned directly next to each other. The results indicated that it was not the separation of the two images that was responsible for the split-attention effect, but

the re-orientation process after the visual focus was switched. This meant that in the troubleshooting tasks, the performance of the learners using the split screens was inferior to that of the learners using the 'vex mode' images.

The prevention of the split-attention effect is regarded as a basic design guideline in developing material with high element interactivity. The elimination of this effect results in more cognitive resources being available for schema acquisition and automation, thereby supporting the process of learning.

3.3.3 Redundancy Effect

Another basic design guideline in creating instructional material is the prevention of the redundancy effect. The redundancy effect occurs when the same information is presented more than once, or in different formats that are not required for learning. The effect has been demonstrated in a number of studies (Bobis, Sweller, & Cooper, 1993; P. Chandler & Sweller, 1991, 1996; Kalyuga et al., 1999; Sweller & Chandler, 1994). Chandler and Sweller (1991) found that learning can be improved by removing textual statements describing the contents of a diagram. The study involved thirty Year 9 students who had no previous knowledge of blood circulation around the heart, lungs and body. The students were split into a diagram-only group (Figure 15), a conventional group (Figure 16) or a modified group, where the diagram and textual statements were integrated (Figure 17).



Figure 15: Diagram only group (P. Chandler & Sweller, 1991)



Figure 16: Conventional group: diagram and accompanying text spatially split (P. Chandler & Sweller,

1991)



Figure 17: Modified - diagram and text integrated (P. Chandler & Sweller, 1991)

The diagram contained labels showing the different parts of the heart, and the arrows within the diagram indicated the direction of blood flow. The textual statements in the modified diagram of the heart reproduced the information that was already evident from the arrows and the labelling of the parts of the heart, and was, therefore, redundant. The study found that students presented with the diagram only learnt more than those students presented either with the conventional diagram and accompanying textual statements (spatially split), or the modified diagram and accompanying textual statements (spatially located together). In the conventional and modified diagrams, processing the redundant textual statements required extra cognitive resources to carry out the task, thereby leaving fewer cognitive resources

available for the essential learning process of schema acquisition and automation. Indeed, whether the text and diagram were integrated or separated had no positive influence on learning, since the textual elements in each experimental condition were redundant and the diagram could be used in isolation for learning.

Kalyuga et al. (1999) also demonstrated the redundancy effect using multimedia computer-based training material. In their research, thirty-four trade apprentices and trainees were separated into three groups for the purpose of learning how to use fusion diagram and solder. The first of the groups was provided with a visual explanatory text only, the second group with auditory instructions only and the third group was provided with both the visual and the auditory instructions, with the auditory instructions being a direct replica of the visual explanatory text. The auditory instructions only group outperformed the auditory and visual instruction group, due to the replicated material inhibiting learning through the increased cognitive load. The dual-mode presentation of the material resulted in the worst performance out of all three cases, evidence that the redundancy effect overrode all benefits of the dual-mode presentation (Kalyuga et al., 1999).

Berends and van Lieshout (2009) examined the effects of different types of illustrations in arithmetic word problems. Four different types of illustrations were used, the "Bare" contained no accompanying illustrations for the word problem, the "Useless" contained a graphic that did not in any way contribute to problem solving, the "Helpful" contained a pictorial representation of the problem and the "Essential" contained information necessary to solve the problem in conjunction with the word problem itself (Figure 18).



Figure 18: Four types of illustrations used in the experiment (Berends & van Lieshout, 2009)

The results indicated that it took longer to reach the correct solution to a word problem if the problem was accompanied by the 'Useless' illustration, than it did if the problem was produced without an illustration, or 'Bare'. However, in both cases the time take to reach a correct solution were not affected by the excessive demands on the working memory, due to the simple nature of the problems (Berends & van Lieshout, 2009). The 'Helpful' illustration contained redundant information, with the diagram being accompanied by a word problem repeating the already relevant information, yet the redundant information did not have a significant effect, with response times and accuracy remaining the same as those for the 'Useless' group. This contradicted the hypothesis of the study, which was that 'Bare' type of illustration would provide the best influence on the speed and accuracy of performance, followed by the 'Useless', 'Helpful', and 'Essential' types in that order. Such a hypothesis was based on the influence of the redundancy effect, where information that is duplicated leads to a higher extraneous load and therefore more cognitive resources being used on processing. The contradiction of such a hypothesis does not necessarily mean that illustrations are detrimental to learning, but rather, demonstrate that illustrations can slow down overall processing (Berends & van Lieshout, 2009).

It has been established that the level of expertise of the learner in a particular domain impacts the effect of redundancy on learning (Kalyuga, Chandler, & Sweller, 1998). Kalvuga et al. (1998) found that novice learners benefitted the most from an instructional design that included both a diagram and physically integrated text. However, as the level of expertise increased, the best results were then obtained from instructional material that included diagrams only, with all text having been eliminated (Kalyuga et al., 1998). Similar findings were obtained in the area of language comprehension and vocabulary learning (Yeung et al., 1997). Yeung et al. (1997) found that as language comprehension increased, the learner's ability to comprehend a passage decreased, when the passage included vocabulary definitions physically integrated into the text. This further validated the findings that with increasing expertise, some information becomes redundant and has a negative impact on learning. These findings are indicative of the process of schema acquisition – as schemas form, the inclusion of some explanatory text becomes redundant, and the replicated/redundant information uses up valuable cognitive resources on processing, thereby hindering the process of learning.

3.3.4 Modality Effect

Modality effect refers to the presentation of two pieces of information that cannot be learnt in isolation, in two different modalities, such as visually and orally, thereby facilitating mental integration of material. The effect can only be obtained if the two sources of information cannot be processed and learnt in isolation. The splitattention effect (*3.3.2* Split-Attention Effect) is evident when two sources of visually presented information are spatially separated and require mental integration. Both sources of information rely on the visuo-spatial sketchpad to process the information, thereby causing overload of scarce cognitive resources. However, there is evidence that presenting information in dual modalities can increase the effective capacity of working memory (Brünken, Plass, & Leutner, 2004; for full meta-analysis, see Ginns, 2005; Jeung, Chandler, & Sweller, 1997; Richard E Mayer, Bove, Bryman, Mars, & Tapangco, 1996; Richard E Mayer & Moreno, 1998; Mousavi, Low, & Sweller, 1995).

For instance, Mayer and Moreno (1998) demonstrated that students receiving auditory explanations concurrently with a computer animation outperformed students who received the same computer animation accompanied by textual statements. In fact, their research found that multimedia learners can integrate diagrammatical and textual information more efficiently if the text is delivered via audio (Richard E Mayer & Moreno, 1998).

Mousavi et al. (1995) examined the effects of dual instructional mode in the context of geometry education. In their research, Mousavi et al. (1995) used two groups, both groups working through geometry worked examples. One of the groups received a diagram with an auditory explanation, while the second group received a diagram accompanied by written text. The first experiment did not control for learning time, and allowed all the students to study the worked examples at their own pace, with the diagram/auditory explanation group taking a longer time to learn the geometry problem (due to listening to the narration from the beginning to the end). In the testing phase, the students in the diagram/auditory group consistently solved problems faster than the diagram/text group (Mousavi et al., 1995). In the second set of experiments, the learning time variable was controlled, but the modality effect remained (Mousavi et al., 1995), suggesting that dual modality instruction can lead to more effective learning.

In a series of experiments, Tindall-Ford, Chandler and Sweller (1997) also investigated the modality effect with worked examples within an adult learning setting, using elementary electrical engineering instructions. The results reinforced previous findings that, for learning, an audio narration with an illustration was superior to a purely visual illustration.

Harskamp, Mayer and Suhre (2007) examined the modality effect in the context of science education. The research used twenty-seven secondary school students, who received a web-based multimedia lesson in biology. Some of the students received a science lesson composed of a series of illustrations accompanied by written text, whilst the others received a series of illustrations accompanied by concurrent narration. Students who received the illustrations accompanied by concurrent narrations outperformed the students receiving the illustrations with concurrent text on subsequent transfer tasks (Harskamp et al., 2007). Additionally, the study found an interaction effect in the post-test scores between the learning times spent on each module and the modality of presentation. The illustration and concurrent narration group outperformed the illustration and concurrent text group once more, but only for students who required less time to learn (Harskamp et al., 2007). Mayer (2005) found that the modality effect is strongest in a situation where the images and words are presented quickly with no opportunity to replay the presentation. For those learners who move at a slower pace, replaying of the material is available, thereby allowing more time to rehearse the text and the images. This can lead to a higher expertise level in the domain, which would help to compensate for visually presented material.

Kalyuga, Chandler and Sweller (2000) demonstrated that the modality effect is only applicable for novice learners; as expertise starts to build, the advantages given by the modality effect virtually disappear and can even lead to detrimental effects to learning. In a set of experiments using instructions for industrial manufacturing machinery, novice learners benefitted the most from receiving instructions in visual format with concurrent auditory explanations. However, once expertise started to build, the modality effect reversed and the group with the visual only presentations started learning more efficiently. The auditory explanations

became redundant information (Kalyuga et al., 2000). This is referred to as the expertise reversal effect (Kalyuga et al., 2003). The reverse modality effect was demonstrated in a series of experiments by Leahy and Sweller (2011). Two experiments were conducted as part of the research. Both experiments tested the ability of primary school students to read graphs showing temperature variations throughout the day, using a visual only presentation and a visual presentation accompanied concurrently by audio explanations. In the first experiment, complex statements were used to explain the visual illustration, whilst in the second experiment, exactly the same information was presented as segmented text in a simpler format. Results of the first experiment indicated a reverse modality effect. whereas the second set of experiments indicated a conventional modality effect. It was hypothesized that the reverse modality effect was due to the use of complex instructions. The transitory nature and the length of the auditory explanations meant they could not be easily and effectively processed in the working memory. This contrasted with the written information presented, which was permanent and therefore could be more effectively processed in working memory.

3.3.5 The Expertise Reversal Effect

When designing instructional material, the expertise of the learner must be given careful consideration, as it may influence the effectiveness of this material (Kalyuga et al., 2003, 1998). The expertise referred to is the learner's prior experience and knowledge of the domain being taught. Thus, a novice learner is someone with very little or no prior knowledge of a specific domain. A novice learner lacks the schemas necessary to process complex information using limited working memory capacity and requires instructional guidance to help bridge the gap of the missing schemas and aid in schema construction. This is in contrast to an expert learner, who is able to activate relevant schemas, due to their prior knowledge and experience of a specific domain. Expert learners do not benefit from additional instructional guidance, and, in fact, such guidance can lead to "...an overlap between schema-

based and the redundant instruction-based components of guidance... " (Kalyuga et al., 2003, p. 24). As a direct result of such redundancy caused by the superfluous instructional guidance, the expert learner may be subjected to cognitive overload whilst attempting to deal with information that they already have schemas for. More experienced learners may benefit from the elimination of instruction-based guidance, thereby reducing cognitive load and increasing learning efficiency. This is known as the expertise-reversal effect, whereby instructional techniques that are highly effective for novice learners, can lose their effectiveness, or even have a negative effect on learning, when used with experts (Kalyuga et al., 2003). Numerous studies have provided evidence that the expertise-reversal effect can be extended to a variety of instructional methods and a wide array of learners (Bokosmaty, Sweller, & Kalyuga, 2014; Kalyuga et al., 2003; Leslie, Low, Jin, & Sweller, 2011). Accordingly, instructional material must be tailored to suit the learner's level of expertise (Kalyuga & Sweller, 2004).

3.4 Summary of Chapter

The goal of cognitive load theory is to use the knowledge of human cognitive architecture to develop effective instructional design. The theory utilises the knowledge of limited working memory, the limitless capacity of long-term memory and the knowledge of how schema acquisition and automation occur to achieve its goal. The theory was first proposed with regards to designing instructional material in the 1980s (Sweller, 1988), and since then has used aspects of human cognition to generate instructional effects that can assist in more efficient instructional design.

Cognitive load theory has identified three different types of load: intrinsic, extraneous and germane. Intrinsic load refers to that type of load found in the material itself, and is dependent on the complexity of the material that needs to be learnt. Intrinsic load is very hard to control without changing the very information that needs to be learnt. Extraneous load is the load that the majority of cognitive load

theory effects are concerned with. It is the load that is extraneous to the learning process and is encountered as a result of badly presented or designed instructional material. Extraneous load interferes with the process of schema acquisition and automation, thereby reducing the efficiency of learning. Germane load is the load produced by the learning process itself, complementing intrinsic and extraneous loads. It describes the working memory resources allocated to dealing with intrinsic load, and is considered to have a positive influence on learning. Accoding to the dual model of working memory that considers the additive property of intrinsic and extraneous loads, germane load is "...generated by intentional additional activities designed to further enhance schema acquisition beyond that associated with intrinsic load..." (Kalyuga, 2011, p. 9)

Some of the effects that have been established as a result of cognitive load theory include: the redundancy effect, the split-attention effect, the modality effect and the worked examples effect. Each of these effects should be considered in order to create the most efficient forms of instructional material. The worked-examples effect occurs when the learner is presented with a worked example of a problem and its solution, thereby freeing up limited working memory resources for learning. This is in contrast to using the more conventional means-ends analysis to solve a problem, where the user is presented with a problem and must successfully reach a solution, thereby using all available cognitive resources on processing potential steps for a solution. The worked examples effect is the most efficient when presented to novice learners, but has been found to hinder learning when expertise is built up in a particular domain.

Split-attention effect occurs when two independent but related sources of information are presented to the learner spatially separated, but still requiring mental integration. The learner must search and match statements, the processing required being similar to that needed for conventional means-ends analysis. The need to process redundant information uses up cognitive resources that would otherwise be

used for learning. This has the effect of hindering schema acquisition and automation, thereby preventing effective learning.

Using two different modalities to present information to a learner utilises the working memory more efficiently, because different modalities are linked to different areas of working memory. The modality effect is experienced by novice learners, studying two sources of material that cannot be understood in isolation to each other.

The expertise-reversal effect occurs both when instructional methods that are more effective for novice learners are not effective for expert learners, and also, when instructional methods effective for expert learners are found to be ineffective for novices.

The next chapter discusses the concept of number sense in young primaryaged children, and the design of instructional material that is used to teach number sense, specifically place value characteristics of the base-10 number system; this is then related to cognitive load theory and the cognitive load effects discussed thus far.

4 Acquisition of the Hindu-Arabic number system

4.1 Introduction

The Hindu-Arabic number system, commonly referred to as the base-10 number system, is the most universally used number system in the world. Indeed, its simplicity has been one of the main reasons why it is considered "...one of the greatest inventions of the human mind..." (Zhang & Norman, 1995, p. 272) There are three main features that help to clearly define the base-10 number system:

- The system is composed of *digits 0-9*, that can be used in combination to represent all possible numbers);
- 2) The system groups its numbers by sets of ten (Figure 19);



Figure 19: The system groups its numbers by sets of ten

 The system uses *place value* (based on powers of 10) to assign a value to a digit based on that digit's position within a number (Figure 20).

Another defining characteristic of the Hindu-Arabic number system is the use of the zero, which is vital to being able to represent a placeholder in the place value notation. For example, in the number 105, the 0 in the tens column indicates that there are no tens in this number, however, the zero still holds a place between the 1 and the 5 to indicate the total value of the number is in the hundreds.

A decimal point in a number signifies that the digits to the left are whole numbers, whereas the numbers to the right of the decimal point are fractional portions of a whole number. From any place in the system, the next position to the left is 10 times greater and the position to the right is one tenth greater than the previous place value.

These features make it possible to write all known numbers in an *expanded form*, as the sum of the digits multiplied by the place value of each of the digits. For example the number 6424 is represented in Figure 20.

Place Value	Thousands	Hundreds	Tens	Ones
Digit	6	4	2	4
Expanded Form	6×1000	4×100	2×10	4×1

Figure 20: Representing the number 6424 using the base-10 number system

Children's understanding of number grows over time. Very early number sense is developed with language, through the use of rhymes, songs and everyday language in a social environment. Children learn the Hindu-Arabic number system early by learning how to count objects and record the numerals. When children first construct number, they construct a system of ones, where 16 would be a collection of 16 ones (Chandler & Kamii, 2009). Early number counting is based on rote learning, the simple memorization of all the number words without an association with the meaning of each of the numbers (Kennedy, Tipps, & Johnson, 2007). Children at this stage are not yet able to think about tens and ones simultaneously (Chandler & Kamii, 2009).

Rational counting occurs later and involves children connecting actual quantities with the words. Counting is a complex cognitive task, requiring five key principles to come together:

- 1) The one-to-one principle objects can only be counted once for each item;
- The stable order principle numbers that are counted are arranged in a sequence that does not change with any number of counts;
- The cardinal principle the cardinal number is represented by the last number in a sequence and represents the number of objects in a set;

- The order-irrelevance principle the order in which objects are counted is irrelevant;
- The abstraction principle any collection of real or imagined objects can be counted.

Throughout the primary years of education, children start learning about place value notation, and how to manipulate large numbers with the help of that notation.

4.2 Place Value

Place value refers to the positional value of a digit within a set of digits and is one of the defining characteristics of the base-10 number system. The value of a digit as written depends entirely on the position of that digit within a string of digits. For example, whilst numbers 72 and 27 utilise the same digits, the position of the digits determines the actual value of the final number. The concept of positional relevance in building numbers is powerful, as it can be expanded or contracted to represent very large or very small numbers without difficulty. Before formal education commences, children think of numbers larger than 10 as a collection of ones, not as a collection of tens and ones, or of hundreds, tens, and ones, etc. (Mix, Prather, Smith, & Stockton, 2014). Formal schooling is needed in order to develop the concept of tens and ones and of place value in general (Fuson, 1990; Fuson et al., 1997; Rouder & Geary, 2014), which in turn provides children with the ability for more efficient number processing and manipulation in the future.

A lot of emphasis is currently being placed on teaching place value in Australian primary schools. The Australian Curriculum Board (National Curriculum Board, 2009, p. 8) describes the role of place value as important in order to "...develop deep understanding of whole numbers to build reasoning in fractions and decimals and develop their conceptual understanding of place value. With these understandings, students are able to develop proportional reasoning and flexibility with number through mental computation skills. These understandings extend students' number sense and statistical fluency...". The Australian Mathematics Syllabus expects students to develop an understanding of place value in Stage 1 (Years 1 and 2), "...applying an understanding of place value and the role of zero to read, write and order two-digit numbers; state the place value of digits in two-digit numbers...". At this stage, students are also expected to be able to count by tens on and off the decade, this ability made much simpler with a more concrete understanding of place value.

Using the latest available data from the Growing Up in Australia: The Longitudinal Study of Australian Children (LSAC), Daragonova and Ainley (2012) examined the teacher ratings of student performance in the area of place value understanding. An examination of the teacher ratings used in the LSAC study showed that students displayed difficulty grasping place value concepts. Results in early primary years (6-7 year olds, with an average age of 6 years and 10 months) showed only 31% of children demonstrated place value skills competently and consistently, while 29% displayed average competence. The remaining 40% of children were either not competent in the skill (7%), just beginning to demonstrate place value understanding (12%), or demonstrated place value skills below average competence (21%) (Daraganova & Ainley, 2012). Similarly, these statistics did not differ much in the middle primary school years (8-9 year olds, with an average age of 8 years and 10 months), where 37% of children showed an above average understanding of place value and 31% demonstrated average competence with place value. The remaining 32% displayed place value skills that were below average (Daraganova & Ainley, 2012).

The mimimum standards for numeracy in Year 3, as defined by the National Assessment Program (NAPLAN, 2016), state that students should be able to compare and order whole two-digit numbers. In addition, children should be able to use place value knowledge up to the hundreds to interpret representations of whole numbers. This demonstrates a very basic understading of place value. Such a level of knowledge does not cover three-digit and larger numbers, even though at that

level students are expected to know how to manipulate numbers well into the thousands. Studies show that children who use decomposition in multi-digit computation tasks, such as breaking up a number based on its positional properties, have overall improved mathematical performance (Geary, Hoard, Byrd-Craven, & Catherine DeSoto, 2004; Geary, Hoard, Nugent, & Bailey, 2013; Laski, Ermakova, & Vasilyeva, 2014). This type of number processing and manipulation requires students to have a much more in-depth understanding of place value. The results of the numeracy component of the National Assessment Program – Literacy and Numeracy (NAPLAN) indicate that 95.5% of students in Australia scored at or below the national minimum standard, a level that only requires children to order and compare two-digit numbers. This demonstrates a very basic understanding of place value. Ordering and comparing three-digit numbers is a Band 3 requirement in the NAPLAN, and only 62.5% of students in Australia were able to meet this level of achievement ("NAP - Results," n.d.).

Understanding place value concepts involves building connections between key ideas, including quantifying sets of objects by grouping them into 10s and treating those groups as units (Fuson, 1990; Steffe, Cobb, & von Glasersfeld, 1988), and using the structure of the written notation to capture this information about groupings. The introduction of multi-digit numbers in school could be difficult to understand even for relatively advanced students (Mix et al., 2014). However, understanding the positional properties or the place value feature of the Hindu-Arabic number system can greatly enhance learning subsequent secondary mathematics material and is essential for understanding and applying multi-digit numbers in further arithmetic and everyday life situations (Bailey, 2015; Chan & Ho, 2010; Chan, Au, & Tang, 2014; Dietrich, Huber, Dackermann, Moeller, & Fischer, 2016; Fuson, 1990; Ho & Cheng, 1997; Kamii, 1986; Mann, Moeller, Pixner, Kaufmann, & Nuerk, 2012; Moeller et al., 2011; Nataraj & Thomas, 2007; Ross, 1989).

Additionally, a significant correlation has been shown between children's place

value skills in early primary school and subsequent problem solving ability (Mix et al., 2014). Schmittau and Vagliardo (2006, p. 590) have emphasised the importance of understanding the system of place value, a concept which "...not only [does it] connect[s] to many important concepts... [but] it is also a prerequisite for any real understanding of the base-10 system ... ". Mix et al. (2014, p. 1306) refer to place value concepts as "...the gateway for conceptualizing large guantities and more complicated mathematical operations...". For example, when multi-digit addition or subtraction is started at school, prior to trading being introduced in the more complex equations, children are often taught to align their numbers to the right for both numerals that they are adding or subtracting. However, they are often not taught why there is a need to align those digits. It would be better for children to understand that each place in the base-10 system needs to be aligned underneath each other perfectly in order to complete the sum easily. This ensures that not only trading becomes easier once it is taught, but also that there is good understanding when decimal place value is introduced. The same rule will still apply to the decimal place value system: line up each of the ones, tens, hundreds etc columns to get the correct answer. Lining up numbers based on the place of each digit in the number would show a good conceptual understanding that the number 20, for example, does not simply refer to a set which is composed of 20 objects, but rather that it also has 2 tens and 0 ones. The understanding that multi-digit numbers represent place value groups of units, tens, hundreds, etc can further influence the sophistication of the problem-solving strategies that the child can use to solve complex arithmetic problems (Geary, 1994d).

Fuson (1990) proposed that the difficulty young children find with place value notation could be related to the vast differences in the spoken and written number systems. These differences make it difficult to see the relationship between written and spoken numbers, without explicit instruction. In fact, research has shown that the base-10 system is more obvious to Asian students, who develop a much better conceptual understanding of place value early in school (Bjorklund & Pellegrini,

2002; Fuson, 1990; Geary, 1994a, 1996; Moeller et al., 2011; Ngan Ng & Rao, 2010). This is due to language differences in the way numbers are spoken out loud. In particular, counting in an Asian language promotes an understanding of numbers that is directly related to the presentation of numbers in the traditional base-10 system (Miura et al., 1993). For example, in Japanese the teen numbers, such as 11, 12, 13 etc. are spoken as "ten-one", "ten-two", "ten-three" and so on. The number 30 is spoken as "three-ten". This means that Asian children have a better understanding of number concepts and have been shown to use more sophisticated techniques to solve complex mathematical problems earlier in their education (Geary, 1996; Ngan Ng & Rao, 2010). English and most European and Slavic languages experience the problem of the words not truthfully representing the positional properties of the base-10 system, making it very hard for speakers of those languages to understand the concept of place value and leading to further problems in mathematics. The earlier and more complete conceptual understanding of place value by Asian students is one of a number of factors including educational and social influences, often leading to higher mathematical achievement in these students, observed as early as the middle of the first year of schooling (Miura et al., 1993; H. W. Stevenson, Chen, & Lee, 1993).

Other studies have proposed that children simply lack the logical capacity at a young age to comprehend the complexity of the place value notation (Chandler & Kamii, 2009). However, research has indicated that children do not enter the early primary years without some understanding of numbers, or sensitivity to statistical patterns in multi-digit numerals and corresponding verbal names. This sensitivity and prior knowledge comes from the child's everyday environment, and includes such things as phone numbers, page numbers in books, and street addresses (Mix et al., 2014). Consistent with this idea of prior knowledge is a study by Barrouillet, Thevenot and Fayol (2010), which found that, prior to entering school, children may be able to differentiate between legal and illegal strings of lexical primitives, for example when comparing three hundred and forty nine to three nine forty hundred. A recent study

conducted by Byrge, Smith and Mix (2013) reinforced these findings by demonstrating that preschool children have the ability to write multi-digit numbers with some success, for the most part only making 'intelligent' errors. For example, the number 113 would be written as 10013, with the children writing one hundred and then followed by the number 13. Whilst not correct, it directly relates the spoken number to its written abstraction. Whilst Moeller et al. (2011) refer to such errors as precursors for later difficulties in mathematics, Byrge et al. (2013) suggests that these types of errors are only a precursor to later difficulties in mathematics if the expanded writing persists in the face of formal place value instruction. In fact, Byrge et al. (2013) have shown that expanded number writing, or 'intelligent' errors, may provide another important milestone in the development of a child's understanding of place value, as they are based on the children's initial and early understanding of multi-digit numbers. This type of understanding could then present a potential foundation upon which formal instruction of place value could begin.

Whilst the reasons for the difficulties experienced in learning place value concepts by children are many and varied, it is evident that the concept of place value itself requires instruction for mastery and is not intuitive knowledge. It could be highly beneficial to base any such formal instruction upon a child's intuitive knowledge and understanding of the place value notation (Byrge et al., 2013).

Various researchers have attempted to describe the progressive stages that apply to children's understanding of two-digit place value (Chan et al., 2014; Cobb, 1995; Fuson, 1990; Fuson et al., 1997; Fuson & Briars, 1990; Miura & Okamoto, 1989; Miura et al., 1993; Resnick, 1984; Steffe et al., 1988; Young-Loveridge, 2002). The authors have not always been in complete agreement on the stages involved in place value understanding. Ross (1989) and Fuson et al. (1997) agree on five stages of conceptual understanding of place value, with a similar approach to each of the stages: unitary multi-digit, decade and ones, sequence tens and ones, separate tens and ones, and finally an integrated sequence and separate tens. This thesis will use the five-stage conceptual understanding framework described by Ross (1989) to assess place value understanding in children, according to the following stages (Table 1).

Stage	Description	
Stage 1: Whole numerals	No meaning assigned to individual digits in a	
	number; number is seen as a whole.	
Stage 2: Positional	Meaning is assigned to individual digits in a	
properties	number; however, despite knowing that the digit on	
	the right is ones and the digit of the left is tens, the	
	meaning of these does not yet encompass	
	quantities.	
Stage 3: Face value	Interpret each digit as representing a number	
	indicated by its face value. Students do not yet	
	recognise that the number represented by the tens	
	column is a multiple of ten.	
Stage 4: Construction zone	Tentative knowledge that the number in the tens	
	column is not just a number but is a set of tens.	
Stage 5: Understanding	Students understand that the individual digits in	
	multi-digit numbers represent a part of the whole	
	quantity split into ones, tens, hundreds and so on.	

Table 1: Five stages of place value understanding, as described by Ross (1989)

4.3 Teaching Place Value

4.3.1 Use of manipulative materials

Manipulative materials in mathematics have long been used to help teach conceptual information that requires physical demonstration. Hynes (1986, p. 11) defines manipulative materials, as "...concrete models that incorporate mathematical concepts, appeal to several senses, and can be touched and moved

around by students...". However, a more recently proposed definition of manipulative materials also includes the concept of engagement with the materials being used, "...A manipulative material is an object that can be handled by an individual in a sensory manner during which conscious and unconscious mathematical thinking will be fostered..." (Swan & Marshall, 2010, p. 14). Manipulative materials can serve as a useful tool that can enhance learning and understanding. Early research suggests that the act of manipulation itself can aid the students in experiencing the patterns and relationships that are the focus of mathematics (Adler, 1966). For example, the NSW Mathematics K-10 Syllabus (Board of Studies Teaching and Educational Standards - NSW, n.d.) recommends using models, such as base 10 material, interlocking cubes and bundles of sticks to help teach place value (Australian Curriculum/Maths/Number and Algebra/028 (ACMNA028) Requirement: Group, partition and rearrange collections of up to 1000 in hundreds, tens and ones to facilitate more efficient counting). Using physical objects and physical actions to represent the world is intuitive, and there are many instances where this is demonstrated in the context of real-life (Marley & Carbonneau, 2014). For example, building a model of a house, using paddle-pop sticks and other material, is often the preferred method of explaining and demonstrating certain structures in architecture class.

Many studies have advocated the use of manipulative materials, such as base-10 blocks or bundling sticks, to teach place value concepts (Baroody, 1990; Fuson, 1990; Hiebert & Wearne, 1992; Nataraj & Thomas, 2007; Ross, 1989). However, there has also been other research that has suggested that manipulative materials could only be used at the children's own level of abstraction. That is, a child must first understand the concept before manipulative materials could be used to represent that concept (Fosnot & Dolk, 2001). In the case of place value, a child must first understand the basic principles of tens and ones to further benefit from the use of manipulative materials (Chandler & Kamii, 2009). Some of the research that opposes the use of manipulative materials in teaching the base-10 system includes the work of Kamii (2000; Kamii & Joseph, 2004) and Fosnot and Dolk (2001). Fosnot and Dolk (2001) consider the use of manipulative materials with the base-10 structure already built-in to be non-beneficial to the learning process. "...Building structure into manipulatives is not always beneficial by itself...abacus, Cuisenaire rods ... Base-10 blocks all have a base-10 structure built in. The problem with these materials is that while the structures in them are apparent to adults, they are not always apparent to children... If a child has not constructed the idea [of tens and ones], she does not see the rod as one ten; she sees it simply as a unit..." (Fosnot & Dolk, 2001, pp. 103–104).

Despite opposing views as to whether or not manipulative materials can aid in learning the place value notation of the base-10 system, one study showed that 81.9% of 820 teachers in over 250 schools in Western Australia used base-10 blocks as a teaching aid in the classroom, and considered them to be one of the three most important tools (Swan & Marshall, 2010). However, individual teacher interviews conducted in the same study found that whilst teachers believed that manipulative materials could assist in learning, they could not identify the exact elements of the materials that were responsible for the learning process (Swan & Marshall, 2010). This could potentially indicate a lack of training for the educators using the materials, or a gap in understanding how to teach children to translate effectively between the representational manipulative material and the abstract mathematical concepts.

To be effective, manipulative materials cannot be used in an erratic manner. The use of the materials must be accompanied by clear and relevant explanations and instruction from teachers, ensuring that the link between the manipulative material used and the mathematical concept being taught is explicit (Stein & Bovalino, 2001; Swan & Marshall, 2010).

This study will utilise two different manipulative materials, the abacus and base-10 blocks, as aids in the understanding of place value concepts.

4.3.1.1 Base-10 blocks

The pioneering work of Dienes (1961) related to the structured materials developed to support children as they learnt about place value, referred to as Multibase Arithmetic Blocks (MAB). The MAB are composed of units, small cubes of wood; longs, made up of as many units long as the base being used; flats, representing the square of the base being used; and blocks, representing the cube of the base being used (Figure 21).



Figure 21: MAB blocks

These blocks remain one of the most popular manipulative materials in the classroom (English & Halford, 1995; Marshall & Swan, 2013; Swan & Marshall, 2010). Base-10 blocks come in four standard sizes, depicting the ones, tens, hundreds and thousands columns. Arguably, base-10 blocks are good at relating the value of the number to the size of the block, making it apparent that as size increases so does the magnitude of the number. Base-10 blocks are good at showing children the magnitude of numbers, although it is hard to relate this back to the positional property of the base-10 system. The concept of place value is based on the position of the digit within a number. It is the position of that digit within the number that affects the digit's magnitude and not the other way around, as the base-10 blocks

may suggest. Base-10 blocks are arguably effective at demonstrating an initial understanding of trading, where ten single blocks can be swapped for a block of ten, and this can be useful in children's learning of place value (English & Halford, 1995). However, despite the common use of this manipulative, there have been contradictory findings in the literature about the effectiveness of base-10 blocks in teaching place value (Fuson, 1990; Resnick, 1984). A more recent study by Mix et al. (2014) explored the use of base-10 blocks to teach place value to 24 kindergarten students, with an average age of 4 years and 9 months. The children were divided into two equal groups, one of the groups using base-10 blocks to teach place value and the second group using symbols-only training for teaching place value. The study found that base-10 blocks were not particularly transparent to children. In fact, the results indicated that the potential correspondence between the differently sized blocks and the place value of the numbers that the blocks were meant to represent was not clear, with performance in block-based tasks consistently poor (Mix et al., 2014). Whilst training with the symbols-only method led to consistent improvements on certain tasks, base-10 block training did not lead to any improvements at all. The results go so far as to imply that perhaps the use of base-10 blocks requires prior knowledge of the written place value system, in order to be successful as a manipulative material intended for improving understanding of place value mix (Mix et al., 2014).

One of the most common ways to test understanding of place value in schools is to show children different combinations of base-10 blocks, arranged in order of magnitude, or canonical order, and asking them to represent the numbers that these blocks symbolize. The problem with this approach is that children commonly rote learn how to perform this type of task. They simply count up the different blocks and then put the two numbers together, whilst perhaps still not having a good concept of actual place value notation. Performing such a task correctly would appear to indicate that a child understands the concept of place value and the positional notation represented by digits in a number, although this might not actually be the case; the child may simply have learnt by rote how to put together base-10 blocks, without any real understanding of the magnitude of the digits within a number (Ross, 1989). Chan et al. (2014) proposed modifying this type of testing and using a method referred to as strategic counting. Using the strategic counting testing method, students would be given base-10 block representations arranged in non-canonical order, thereby requiring an understanding of base-10 properties in order to rearrange and group sets together to produce meaningful answers.

The contradictions in the efficiency of base-10 blocks as a training tool to teach place value could be attributed to a number of factors, incuding the child's first language, the quality of the teaching material accompanying the use of the manipulatives, teachers' understanding of place value, and the length of time spent on the topic.

4.3.1.2 The Slavonic abacus

Another manipulative, widely used in Asian countries and Eastern Europe, is the abacus. Throughout history, the abacus has played a part in showing the positional representation of numbers (Barbarin & Wasik, 2009). The simplified Slavonic abacus is composed of a hundred beads and allows counting up to one hundred (Figure 22).



Figure 22: Slavonic abacus

When the beads are positioned on the right, they are 'off' and are therefore not forming a number. Pushing beads to the left switches them 'on' and this is how numbers are represented on the abacus. There are ten rows altogether, with each row grouping beads into tens. This grouping in a row is then halved again with each half of the beads in different colours. This correlates to the number of fingers on each hand, and may therefore aid in subitising. Additionally, the first five rows will differ in colour to the last five rows, so as to allow easy subitising for each group of fifty. For example, the number 37 can be displayed on an abacus using the movement of beads in Figure 23.



Figure 23: Abacus showing the number 37 (with its complement to a 100 of 63 clearly and easily subitised)

The main advantage of using a Slavonic abacus is that it clearly encourages the grouping of numbers in sets of ten, one of the properties of the Hindu-Arabic number system. It also does not require a prior understanding of multi-digit numbers; it is enough for students to know the verbal sequence of numbers to be able to count the beads on the Slavonic abacus.

Additionally, each row of the Slavonic abacus is made up to two sets of five beads, each set in different colours. The use of the small number of beads in different colours and its direct correlation to the number of fingers on each hand can support the child's ability to subitise. Subitising, or the ability to automatically recognise small numerosities without counting is an innate ability, and is therefore considered to be a biologically primary skill (Geary, 1994e). Furthermore, our biological ability to make sets of numbers is also supported, as there are two sets of five on each row, and a set of ten on each wire. This also reflects clearly the base-10 property of the HinduArabic number system. The Slavonic abacus physically lends itself to counting in the base-10 setting, where each full line of an abacus is called a 'ten line' and a maximum of nine beads are allowed to be placed on a line. This setup helps to facilitate mental visualisation of a group of 'ten', as each row reresents a group of ten and allows to the child to count in a base-ten setting. The abacus can then facilitate the understanding of place value, by physically representing, for example, that the number seventy-six is made up of 7 tens, or seventy, and 6 ones. Such a visualisation can potentially improve schema formation for the base-10 property of number.

A recent longitudinal study examined whether using a mental abacus, a technique used to perform mathematical calculations using a mental image of an abacus, can improve students' mathematical abilities and lead to improvements in basic cognitive capacities, such as working memory (Barner, Alvarez, Sullivan, & Frank, 2016). Using a mental abacus relies primarily on the visuospatial working memory, as well as some motor procedures that are learnt during training with a physical abacus. The technique is based around the structure of the abacus, which is consistent with known working memory limits. The results of the study suggested that mental abacus training did not lead to consistent gains in the cognitive abilities of the students (Barner et al., 2016). However, mental abacus training did lead to measurable gains in ability to perform accurate mathematical computations. These gains emerged after a single year of training with a physical abacus, prior to the students learning the mental abacus technique. Towards the conclusion of this study, Barner et al. (2016) found that physical abacus expertise was significantly correlated with higher mathematical performance.

4.4 Summary of Chapter

Evolutionary educational psychology focusses on the ideas that relate evolutionary principles to educational foundations (Muller, 2010). The goals of evolutionary educational psychology are to determine what motivates learning and what motivational biases influence a child's ability to acquire evolutionary novel knowledge, as demanded by our system of schooling. Achieving these goals would result in more effective teaching methods and the ability to teach children in a way that is more naturally aligned to their intuitive skills and knowledge. Evolutionary educational psychologists claim that currently there is a great disparity between the way that we teach children and how children actually prefer to be taught (Muller, 2010).

Mathematics is an important area where educators need to be able to understand how children prefer to be taught, but, to date, we do not fully understand the mechanisms that influence children's mathematical learning (Geary, 1994b). Of particular interest are the motivational biases that can influence the way children come to an understanding of the abstract concept of number and its relationship to its symbolic representation, a concept which appears to be very difficult to grasp for children (Baroody, 1990; Fuson, 1992; Hiebert & Wearne, 1992). In particular, the positional property of the Hindu-Arabic number system, or the base-10 system, causes the most difficulty. Understanding the positional notation of our number system is essential to the development of number sense, and also forms the basis for four fundamental number operations (Jordan, Kaplan, Ramineni, & Locuniak, 2009). Chan et al. (2014, p. 78) refer to place value as "...crucial in learning arithmetic..." Understanding the base-10 system in early primary school years is essential, as research has shown that flawed understanding of the place-value positions of tens and units in first grade is related to difficulties in later arithmetic performance (Chan & Ho, 2010; Chan et al., 2014; Dietrich et al., 2016; Fuson, 1990; Ho & Cheng, 1997; Jordan et al., 2009; Moeller et al., 2011).

Place value errors are often syntactic, with children often using expanded notation to write down numbers, for example, the number 326 written as 300206. These mistakes then follow through to multi-digit addition and subtraction (Fuson, 1990). There are a number of different ideas relating to the conceptual structures in place value understanding. This thesis will be using the conceptual structures

proposed by Ross (1989) and will examine place value understanding in five different stages. The first stage has the children assign no meaning to the individual digits in a number, whilst the second stage sees meaning assigned to individual digits, although the number 23 is still seen as 23 ones. The third stage sees an understanding of the face value of digits, and the fourth stage sees the tentative knowledge about tens and ones. The final stage sees the understanding of a multi-digit number as individual digits split into ones, tens, hundreds and so on.

The use of manipulative materials is common in teaching concepts in mathematics, although there is some debate as to the efficiency of manipulative materials in teaching place value notation in particular. A very common manipulative used to teach place value concepts are base-10 blocks. Base-10 blocks are good at showing children the magnitude of numbers; however, it is often difficult to relate this back to the positional property of the base-10 system, as the blocks differ in size only. Another manipulative widely used in Asian and Eastern European countries is the abacus. The main advantage of the Slavonic abacus is that it clearly encourages grouping of numbers in sets of ten, one of the key properties of the Hindu-Arabic number system. It also does not require a prior understanding of multi-digit numbers; it is enough for students to know the verbal sequence of numbers to be able to count the beads on the Slavonic abacus.

This thesis will utilise base-10 blocks and a Slavonic abacus as manipulative materials to aid in teaching the concepts of place value. The next chapter looks at the literature pertinent to this thesis and provides a broad overview of the series of experiments conducted as part of this study.

5 Research in the field

5.1 Research pertinent to this thesis

Multi-digit numbers present a complex cognitive challenge to children, and the understanding of their functionality is only just starting to emerge (Nuerk, Moeller, Klein, Willmes, & Fischer, 2011). Interest in children's learning and processing of multi-digit numbers has increased significantly over the last few years (Mann, Moeller, Pixner, Kaufmann, & Nuerk, 2011). Not only is place value and its relationship to multi-digit numbers a difficult concept for young children to grasp, but the abstract mathematical idea of place value is also a challenging topic to teach (Bailey, 2015). Von Aster and Shalev (2007), in their research, present the idea that children are first introduced to multi-digit numbers as an expansion of single digit numbers, at a time when they start to acquire the compound number words of twodigit numbers. It is, therefore, vital that place value instruction starts in early education. However, it still leaves the question of how to best teach such concepts at a young age. Development of multi-digit number understanding and processing in children is vital, not simply to their ability to count, but also as an important stepping stone to other basic numerical tasks including transcoding, estimation, magnitude comparison and generalization to basic arithmetic operations (Nuerk et al., 2011).

Manipulative materials are concrete objects that are often used around the world to assist with learning and understanding new mathematics concepts. Such materials offer students the opportunity to explore concepts at both the visual and tactile level, through hands-on experience (McNeil & Jarvin, 2007). Research into the use of manipulatives in the classroom to teach mathematics concepts has shown mixed results. Some manipulatives have been found to help students under some conditions (Aburime, 2007; Mix et al., 2016), whilst other studies have shown there to be no benefit from using manipulative materials. Still others have shown that the

use of such manipulative materials can actually hinder learning under particular conditions (McNeil & Jarvin, 2007). Carbonneau, Marley and Selig (2013) provided a good meta-analysis of the current state of research into the use of manipulatives in the classroom in primary and high schools. The meta-analysis covered some very early studies by Fennema (1972) and similarly by Friedman (1978), who showed that manipulatives can lose their value after the first year of schooling. However, despite the mixed results of recent studies, the popularity of manipulative materials in the classroom remains strong. The use of manipulatives can pose a cognitive challenge to students' limited cognitive resources, in the form of dual representation, where a given manipulative not only needs to be represented as an object on its own, but also as a transparent symbol of whatever mathematical concept it is intended to explain (Uttal et al., 1997). Some early work by Boulton-Lewis (1998) found that when children interacted with manipulatives, their limited cognitive resources were all focused on representing and manipulating the objects, leaving no resources to establish actual understanding of concepts. It is, therefore, vital to continue research into how the use of manipulatives could affect students' limited cognitive resources and how this may impact learning and understanding in the mathematics domain.

Another common way in which novel information is presented in mathematics workbooks is with the help of worked examples. Worked examples present the learner with a problem and also the worked-out solution showing the steps that need to be taken to reach a goal state. This method of presentation means that a learner can dedicate all their cognitive resources to developing a schema for solving similar problems. Worked examples are particularly beneficial with novice learners in comparison to other standard problem solving techniques, provided the examples are well designed (van Loon-Hillen, van Gog, & Brand-Gruwel, 2012). For example, an early study by Tarmizi and Sweller (1988) found that if the worked example presented information in a split-attention format, the worked example effect was not

observed, and was only observed again when the same information sources were presented in an integrated manner.

Worked examples in mathematics have been widely researched (Chen, Kalyuga, & Sweller, 2015; G. A. Cooper & Sweller, 1987; Spanjers, Gog, & van Merriënboer, 2011; Sweller & Cooper, 1985; Takir, 2012; Zhu & Simon, 1987); however, these have been, for the most part, conducted with high school or early university students. Only a handful of known studies have focused on the efficiency of worked examples when used with primary school children, (Hu, Ginns, & Bobis, 2015; Mwangi & Sweller, 1998; van Loon-Hillen et al., 2012). Van Loon-Hillen et al. (2012) found that the only difference between groups that had access to worked examples and those groups that did not was in the acquisition time, the time taken to study the presented material. However, in the study conducted by van Loon-Hillen et al. (2012), students were subjected to a realistic maths teaching environment within their classroom curriculum, with access to worked examples, which they could chose to use or not. This made it very difficult to measure accurately the benefits to learning of the worked example, which was a limitation of that study (van Loon-Hillen et al., 2012). Previous research by Mwangi & Sweller (1998), and more recent work by Hu (2014; Hu et al., 2015), had also shown positive learning outcomes when the worked example was utilised in primary school mathematics instruction. The efficacy of using such worked examples was dependent on the structure and content of instructional material.

5.2 Rationale for study

NAPLAN results indicate that Australian children have great difficulty learning the characteristics of the base-10 number system (Daraganova & Ainley, 2012), in particular the concept of place value, which is essential for understanding expanded notation, multi-digit addition and subtraction, and other basic concepts. Children in
early primary years often do not understand that the number is made up of a series of quantifiable ones, tens and hundreds (and so on).

The goal of this research project is to investigate whether there is a more efficient way to teach the properties of the base-ten system to early primary schoolaged children, based on contemporary understanding of the characteristics and origins of human cognitive architecture. The research project looks at the role that manipulative materials, such as MAB blocks, and other base-10 materials, play in teaching place value concepts, from the viewpoint of cognitive load theory. This is done with the help of worked examples, which are used throughout the study to present novel concepts to students. It is envisaged that the use of the worked example strategy will aid in reducing the imposed extraneous load, and lead to a more consistent instructional design across the many groups of students tested. The main research question of this study therefore is:

Do base-10 manipulative materials represent a biologically primary skill that can be used to efficiently transition to secondary knowledge of the base-10 system, thereby improving children's understanding of number and place value?

The general hypothesis of this thesis is that the use of base-10 manipulative materials for teaching place value may create the redundancy effect, thereby minimising any positive effect that the use of such materials might have on learning. Whilst the use of base-10 materials may have its foundations in children's biologically primary skills, the fact that the use of such materials must be accompanied by explicit guidance, in order to provide any benefit to learning, may mean that such tools are not an efficient way to transfer between primary and secondary knowledge.

5.3 An introduction to the experiments in the thesis

The previous chapters have discussed the difficulty with teaching and learning the concepts of number and place value. The current study was designed with the intention of making a further contribution to this discussion, by investigating the ways in which place value is taught and the effect this has on student cognitive resources.

This thesis used five experiments to test the hypothesis that using base-10 manipulative material, and in particular MAB blocks, within the construct of a worked example, can lead to the redundancy effect and, therefore, to a lack of improvement in learning. This could be due to the fact that MAB blocks, one of the most commonly used manipulative materials for teaching place value in the Australian classroom, do not correspond to our primary biological knowledge. Such tools require children to learn the abstract connection between the meaning of the blocks and the actual number system. Learning both the concepts of the blocks and the number system represent biologically secondary knowledge and require effort and skill to learn. This is unlike the common and intuitive use of blocks to build or play with, which represent biologically primary knowledge. Thus, the use of the manipulative material requires additional learning, not only of the function and meaning of the manipulative material but also the relationship that the material has to the number system, which places an even higher load on children's limited cognitive resources.

Experiment 1 investigated whether a positive impact on learning the concept of place value could be achieved using different manipulative materials, and, specifically, whether a particular manipulative material could more neatly fit with our biologically primary knowledge, thereby providing a more intuitive way of grasping the concept of place value. Students were randomly assigned to one of four groups: groups 1 and 2 were given MAB blocks to assist with place value learning, with group 1 learning the five different stages of place value understanding gradually and group 2 learning only the fifth stage of place value understanding, without focusing on the preceding stages. Groups 3 and 4 were given Slavonic abaci to assist in learning, with group 3 learning the five stages of place value understanding gradually, and group 4 learning only the fifth stage without focusing on the preceding stages. Students spent the day undertaking place value related activities using the relevant manipulative material to assist them in learning. One of the hypotheses proposed that using the abacus, instead of the MAB blocks would lead to better performance in learning place value concepts, due to the more intuitive structure of the abacus.

Experiment 2 was designed to remove the real-life teaching bias found in Experiment 1, by utilizing the concept of worked example based learning, with the assistance of the same two manipulative materials used in Experiment 1, and the addition of a control group which was not provided with any manipulative materials to assist them in completing the set tasks. The aim of this experiment was to establish whether worked examples with certain manipulative materials were more beneficial than standard worked examples that did not use any manipulative materials.

Experiment 3 reduced the complexity of the worked examples by reducing the amount of literacy required in question comprehension and once again used worked examples with MAB blocks, Abacus or control groups (no manipulative materials) to see whether worked examples with particular manipulative materials are more beneficial to learning. It was argued that worked examples that do not utilise the images or the physical element of the manipulative materials would result in better performance in comparison to manipulative materials being used.

Experiment 4 once again utilised the concept of teaching/learning with worked examples, but reduced the sample size to only using one manipulative material, as no differences were found in Experiments 1, 2 or 3 for the different manipulative materials in learning the concepts of place value. This experiment was designed to test the hypothesis that the use of MAB blocks is in fact detrimental to learning the concepts of place value as it causes a Redundancy Effect both when presented on paper for the student to study in their own time and as a physical manipulative material. Experiment 5 used more advanced learners with more complex learning material to again test whether MAB blocks, often used to assist students in gaining the early concepts of place value, may in fact hinder learning, as they pose a Redundancy Effect both when presented in their physical form and also as images within the worked examples.

6 Experiment 1 - Place Value

6.1 Introduction

The series of experiments making up this thesis aimed to investigate whether the current methods of teaching place value with the aid of manipulative materials could be improved, based on cognitive load theory. Experiment 1 was designed to explore whether teaching place value could be improved with a more gradual approach to number understanding established first, and also whether teaching with MAB blocks as is the current standard, is superior to teaching with any other manipulative material, such as a Slavonic abacus.

It was predicted that the gradual transition using the abacus, where each stage of place value understanding was covered and practised individually before the following stage was introduced, would produce better results in genuine understanding of the place value topic than using the MAB blocks in an identical transition. Similarly, using an abacus to teach place value in an integrated fashion from the beginning, without first focussing on the separate stages, was predicted to produce better outcomes than using MAB blocks in an identical transition. This would reflect the gradual building of schemas during the intervention/acquisition phase, and thus lower levels of the inherent intrinsic cognitive load. It was hypothesized that the use of the Slavonic abacus would be more intuitive to the students' basic understanding of number construction due to its one-to-one correspondence with counting, i.e. one bead counts as one object. This is in contrast to MAB blocks, that do not represent number correspondence intuitively but attempt to do it visually through object size.

6.1.1 Participants

Experiment 1 was conducted with sixty Year 2 and remedial Year 3 students, aged 7-8, picked at random from two Sydney Eastern Suburbs tutoring centres. The children were a mix of both genders, and a range of private and public schools. The centre specialises in both remedial education and gifted students, so there was a range of students with different academic abilities. Students who chose to participate were randomly assigned to one of four groups:

- 1) Gradual transition group with the abacus as a tool (n = 15)
- 2) Gradual transition group with base-10 blocks as a tool (n = 15)
- 3) All-at-once transition with the abacus as a tool (n = 15)
- 4) All-at-once transition with base-10 blocks as a tool (n = 15)

6.1.2 Materials and Procedure

A 2 (manipulative materials: abacus vs. base-10 blocks) \times 2 (method of instruction: gradual transition vs. all-at-once transition) mixed factorial design was used with 15 participants in each of the groups (Table 2).

Table 2: A 2×2 mixed factorial design in Experiment 1

	Gradual Transition	All-at-once Transition
Abacus	15	15
Base-10 blocks	15	15

Children were invited to participate in the study throughout the Term 1 period, and were then tested in Term 2 of the school year, after they had covered basic number concepts at school but before they have had an opportunity to study place value concepts in depth. The study took place over four consecutive weekends, with each session lasting the whole day. All the materials used as part of this experimental setup can be found in Appendix A: Experimental materials used in Experiment 1. Prior to commencing teaching, children were given a pre-test, examining their prior knowledge and current understanding of place value concepts (up to 45 minutes in length). This was followed by one of four intervention strategies, with regular breaks given to the children. The day was structured to include five sessions of half an hour in length each with a 10-minute break between sessions and a longer lunch break. Each session involved one teacher, who was conducting the lesson, and a second teacher, to assist with class control and any administrative matters. The same two teachers were used for all sessions.

There were two types of manipulative materials used: the Slavonic abacus (one-hundred frame) and MAB blocks. The Slavonic abacus is made up of ten rows of ten beads each; with each group of five beads on any single row of the abacus in a different colour to ensure that efficient subitising is possible. The MAB blocks were familiar to the children, as they used them during their normal school time activities in developing their understanding of place value. Each set used in the experiment consisted of shorts (units), longs (tens) and flats (hundreds). When making numbers in the thousands, children were asked to put together the necessary number of flats to show understanding of number magnitude.

Children in two of the groups received a gradual instruction of the base-10 system, providing a segmented approach to learning based on each stage of learning place value, as described by Ross (1989). Accordingly, each of the five half-hour sessions covered each individual stage of understanding of place value (Table 3).

Stage	Description	Task to test	Example of
		understanding	question
Stage 1:	No meaning	Transcoding –	Numbers are read
Whole	assigned to	Question 1 (6 items)	out loud to
numerals	individual digits in a		students who are
	numeral, number is		required to write
	seen as a whole.		them down (up to

Table 3: Stages of	blace value	understanding	in the	pre-test and	post-test
	p				

			and including
			hundreds)
Stage 2:	Meaning is assigned	Number	12, 22, 32, ?
Positional	to individual digits in	Patterns/Counting -	
properties	a numeral, however,	Question 2 (4 items)	
	despite knowing that		
	digit on the right is		
	ones and digit of the		
	left is tens, the		
	meaning of these		
	does not yet		
	encompass		
	quantities.		
Stage 3: Face	Interpret each digit	Basic Maths	12 + 52 = ?
value	as representing a	operations –	74 – 31 = ?
	number indicated by	Question 3 (10	
	its face value.	items)	
	Students do not yet		
	recognise that the		
	number represented		
	by the tens column		
	is a multiple of ten.		
Stage 4:	Tentative	Representation of	Write the number
Construction	knowledge that the	numbers -	32 in words.
zone	number in the tens	Question 4 (7 items)	
	column is more than	and Question 5 (8	Write the number
	just a number but is	items)	'one hundred and
	a set of tens.		three' in numbers.

Stage 5:	Students	Expanded Notation -	Write the following
Understanding	understand that the	Question 6 (4 items)	in expanded
	individual digits in		notation:
	multi-digit numbers	Face Value –	341 =? + ? + ?
	represent a part of	Question 7 (4 items)	
	the whole quantity		What is the face
	split into ones, tens,	Place Value -	value of the tens in
	hundreds and so on.	Question 8 (3 items)	the number 652?
			What is the total
			value of the tens in
			the number 349?

The first session focussed on transcoding numbers, starting from smaller and building up to greater numbers. The students practised saying numbers out loud, covering numbers from one-digit to three- or four- digits. The use of zero as a placeholder was also discussed and used. The relevant manipulative materials, either the Slavonic abacus or the MAB blocks were also used as part of this activity to aid in understanding. The second session discussed the different place value positions in a number, starting on the right and from the lowest possible unit and building up to larger three digit numbers. The next session focussed on using numbers up to three digits and on being able to specify face value for a required digit within a number. In the fourth session, the children began to construct numbers, and discussed what makes up a two-, three- or greater digit number. Starting from a one-digit number and building up to a three-digit number, the students participated in activities that involved deconstructing the numbers into ones, tens and hundreds. Finally, the last session brought all these concepts together, with the children being

able to deconstruct numbers and easily understand the difference between face value of numbers and total value of numbers.

Children in the remaining two groups received an all-at-once lesson, which covered the last stage of understanding without focussing separately on the preceding stages. Their material was delivered also in five half-an-hour sessions; however, the lesson was continuous in determining place value in specific numbers. A discussion of numbers starting with one-digit numbers and continuing up to threedigit numbers was undertaken, using place value to write and record numbers, but without specifically discussing the total values of digits within a two- or more digit number. For example, the children in the gradual groups worked on transcoding numbers in the first session, discussing place value positions in the second session, specifying face value in the third session, constructing numbers in the fourth session and finally working with deconstructing and constructing numbers in the last session by specifying what place value each digit represented. This is in contrast to the children in the all-at-once group, who worked on construction and deconstruction of numbers with one digit in the first session, two-digit in the second session and threedigit numbers in the third session. The fourth and fifth sessions were used to work through different number deconstruction, with children specifying what place value each digit in a number represented.

After the lesson, children were tested again to establish any benefits they might have received from either of the interventions in understanding the base-10 system. The post-test was composed of ten questions, and used similar questions to the ones used in each of the interventions, and in the pre-test. The questions in the test were divided according to particular areas of understanding they measured, based on the five stages of place value understanding (Ross, 1989). Each test item was marked as either correct or incorrect, with no partial marks being given. The stage of place value understanding was deduced from the results of each test.

It was predicted that the gradual transition, where each stage of place value understanding was covered and practiced individually before the following stage was introduced would produce better results in genuine understanding of the place value topic than if the place value was simply taught in an integrated way from the beginning without focussing on the separate stages first.

6.2 Data Analysis

To check the approximate normality of distribution, values of skewness and kurtosis were examined. These values were converted to z-scores by dividing them by their standard error, if the resulting score was greater than 1.96, a potential violation of the independent groups t-test assumption of normality was evident (Field, 2007). The Shapiro-Wilks test of normality was also used to test for parametric distribution assumptions. Where normality assumption was not violated, the independent groups t-test assumption of equality of variances was examined using Levene's test. Non-significant results of this test meant that analysis with a standard independent groups t-test suitable for heterogenous variances within the SPSS software package. The experimental data reported as part of this thesis combined tests of significance, controlling the Type 1 error rate at 0.05, with estimates of the standardised mean difference effect size (d). This thesis uses the benchmarks for effect size magnitude as defined by Cohen (1988), "small, d = .2,", "medium, d = .5," and "large, d = .8 and above".

6.3 Results

The scores received by each of the students in the pre-test were measured and analysed to understand base-line knowledge in all the groups, and to ensure that there were no significant differences in the cohorts of students between the Abacus (n = 30, M = 79.97, SD = 12.11) and MAB blocks (n = 30, M = 76.93, SD = 15.62), F(1, 59) = .714, p = .402.

The independent variables in this experiment were the types of materials and the method of instruction, while the dependent variables under analysis were total post-test scores. Table 4 shows the means and standard deviations of each of the variables.

		Pre-test %	Post-test %
Abacus			
(N = 15)	Gradual	77.87 (12.94)	81.13 (14.03)
(N = 15)	All-at-once	82.07 (11.27)	82.67 (14.13)
(N = 30)	Total	79.97 (12.11)	81.9 (13.85)
MAB			
(N = 15)	Gradual	73.40 (20.61)	78.87 (18.46)
(N = 15)	All-at-once	80.47 (7.33)	84.0 (6.89)
(N = 30)	Total	76.93 (15.62)	81.43 (13.94)
Total			
(N = 30)	Gradual	75.63 (17.06)	80.0 (16.15)
(N = 30)	All-at-once	81.27 (9.38)	83.3 (10.94)
(N = 60)	Total	78.45 (13.94)	81.67 (13.78)

Table 4: Descriptive statistics for place value understanding

Using a two-way ANCOVA with the pre-test total scores used as a covariate, to take into account variations in children's understanding of concepts prior to the instructional session, it was found that there were no statistically significant differences in the post-test mean marks between the abacus or the MAB blocks groups, F(1, 59) = 1.146, p = .289, $\eta p^2 = .020$ (small effect size) – i.e. no significant main effect of the type of materials. Similarly, there was no statistically significant difference in post-test total mean marks between the gradual transition and the all-

at-once transition instructional methods, F(1, 59) = .527, p = .471, $\eta p^2 = .009$ (small effect size) – i.e., no significant main effect of the instructional method.

Additionally, there was no significant interaction between the manipulative material (abacus v. MAB blocks) and the method of instruction (gradual v. all-at-once), F(1, 59) = .088, p = .768, $\eta p^2 = .002$ (small effect size).

A brief examination of any potential gender effects, using the independent samples t-test, indicated that there were no statistical differences in the performance of males and females in either the pre-test scores, t(58) = 1.000, p = .322, or the post-test scores, t(58) = 0.647, p = .520 in the overall marks across all four strategies of teaching.

No other statistically significant differences were detected between the different materials or instructional methods used, or any interactions between the materials and methods.

6.4 Discussion

Experiment 1 was designed to test the hypothesis that teaching place value with the aid of MAB blocks would be inferior to teaching place value with a Slavonic abacus. It was hypothesized that because blocks do not directly translate to the numbers that they represent, it would be more beneficial to use an abacus. When using MAB blocks, the student is expected to decide which block is bigger based on the size of the block. No direct counting is involved, and choosing the correct block size can be easily learnt by rote. In contrast, each bead of the abacus is the same size and is counted as one, with beads arranged in groups of ten. This could reflect a more dirrect correlation between the physical object and counting. Additionally, the experiment attempted to establish whether a gradual teaching method (Ross 1989), would be more beneficial to students than an all-at-once transition.

The results of this experiment did not support the hypothesis. There were no significant differences in the use of MAB blocks or the abacus to teach the concepts

of place value and there were no significant differences in the methods of teaching. Additionally, no interactions between the method of teaching and the materials used were found. This result could be due to the children's familiarity with MAB blocks, a material that they often use within the setting of their normal classroom. Observationally, children were able to use the MAB blocks in such a way that showed that they had simply "memorised the answer", that is, they knew how to use the blocks to construct the necessary numbers, without actually making the connection between the blocks and numbers, and showing no inherent understanding of the concept of place value. Some issues were also observed with children using the Slavonic abacus. The abacus was used to help the children count out numbers: however, all counting was mostly done in ones and the logical tens mapping of each line was not used by the students for the most part. This lack of connection between the meaning of each number and the deconstruction of each number into its relevant base-10 values was evident in the common mistakes made by the children in their post-tests. A small number of children had trouble writing numbers such as "one hundred and two" and would write these numbers as "12". This was repeated in the representation task in both the pre- and post-tests, where children did not include the zero placeholders when converting words to numerals (Figure 24). This demonstrated a very early stage of place value understanding, where no meaning was assigned to the actual values that each digit within a particular number represented.



Figure 24: Example of mistakes made in the representation task, converting words to numerals

An inability to bridge the 100s when counting off the decade was another common mistake made by more than half (58.3%) of all the children tested (Figure 25) in their post-test results. This mistake indicated stages 1 or 2 in the level of understanding; however, this was specifically observed with three-digit numbers only.



Figure 25: Mistakes showing an inability to bridge the 100's when counting off the decade in the post-test

One of the most common mistakes made was in specifying expanded notation of a number. A large majority of children in their post-test, specifically 78.3% of all children tested, were not able to complete this question, or could not complete it correctly. The mistakes made while attempting this question showed a lack of understanding of how numbers are built, and that each digit within a number represented a given quantity based on its location (Figure 26).



Figure 26: Common mistakes made in expanding a number

Stage 5 of understanding, where children are easily able to determine what place value is, expand a particular number based on place values for both single and multi-digit numbers, was only attained by 8% of the children tested in the post-test. The remainder of the children mostly left this section blank or attempted to complete this section with mistakes, which showed a lack of understanding of the concept (Figure 27).



7) What is the number of tens in 342? 8) What is the number of hundreds in 587? 9) What is the number of ones in 324? 324

Figure 27: Mistakes made by children in specifying place value in numbers

All these mistakes were made despite the fact that, throughout the intervention, children were able to successfully use MAB blocks and the Slavonic abacus to construct numbers. Thus, observationally, whilst children understood what the different MAB blocks represented, they were not actually associating the blocks with the numerical values. Similarly, the children were not exploting the base-10 characteristics of the Slavonic abacus. The children had simply learnt how to build the numbers correctly using the blocks or to count out the numbers using individual beads, without any actual understanding of the quantities that the numbers and blocks or the lines of beads represented. It appeared that for those children who could easily construct numbers, and who had a good understanding of place value, the representative MAB block sizes actually correlated easily to the numerical value desired. Similarly, children with a good understanding of place value were able to count in tens on the Slavonic abacus by using a whole line of beads at once. However, for those children who had not yet made number associations, MAB blocks or the Slavonic abacus, did not appear to help bridge the gap between the physical objects and their meaning. Children were able to construct numbers more quickly using the Slavonic abacus, as each bead represented a single count, an association that was easier to make for the children and requiring fewer of the limited cognitive resources. This was in contrast to the MAB blocks, where patterns in the structure of the blocks and the association between the blocks and the numbers, were difficult to construct using three different sized blocks.

7 Experiment 2 – The Worked Example and Place Value

7.1 Introduction

The results of the first experiment showed no significant differences in student performance between the gradual or all-at-once transition or the two different manipulative materials used (abacus vs. MAB blocks). The lack of any differences could be attributed to the real-time method of instruction in the classroom. Whilst great care was taken to ensure consistency across all the groups with the same teachers and the same material presented to the students during instruction, there were still differences that could not be controlled, due to the real-time nature of classroom instruction. These included differences in questions posed by children, and the differing personalities of the children themselves, with some children more willing to ask questions to improve their understanding as necessary. In order to achieve a more consistent approach in the acquisition stage of the experimental setup, Experiment 2 based its instructional materials on the worked example effect, which states that providing worked examples is a more efficient way to teach novice students, in comparison to straight problem solving (G. A. Cooper & Sweller, 1987; Sweller, 1988).

The children in Experiment 1 demonstrated a lack of place value knowledge, in particular, when attempting number decomposition. The students scored an average mark of 60.9% in post-test scores and an average mark of 55.6% in pretest scores in this area. Experiment 2 thus focussed on expanded notation (number decomposition), an area that children had little or no prior knowledge in, and was therefore the most likely area to produce the worked example effect.

The aim of Experiment 2 was to test the hypothesis that a worked example using the students' primary skills of movement and grouping knowledge and representation of number, when using an abacus, would be superior to that of using the MAB blocks as the manipulative material, in learning the secondary skills of place value and number knowledge, particularly when focussing on problems involving expanded notation and face value.

7.2 Method

7.2.1 Participants

Experiment 2 was conducted with forty-eight Year 2 students, from a selection of three private schools in the Eastern Suburbs, and Northern Suburbs of Sydney. My School website (www.myschool.edu.au) was used to better understand the backgrounds of each of the schools. School 1, located in metropolitan Sydney, had a student population of 2044, with 17% of students having a language background other than English. My School classified 83% of the students as being in the top quartile of the Index of Community Socio-Educational Advantaged. It is therefore considered to be a very advantaged school. School 2, also categorised as 'metropolitan', had a student population of 753, with 83% of students having a language background other than English. My School classified 36% of the students as being in the top quartile of the Index of Community Socio-Educational Advantaged. It is therefore considered to be a less advantaged school. The last school, School 3, also a metropolitan school, had a student population of 861, with 2% of students having a language background other than English. My School does not report the data for the Index of Community Socio-Educational Advantaged for this school. Students were all aged approximately 7-8 years, and were randomly assigned to three groups of sixteen students each: Abacus worked example, MAB blocks worked example, No visuals worked example (control).

7.2.2 Materials and Procedure

Students were tested in groups of three, with one student using an Abacus worked example and having access to an abacus, the second student using an MAB blocks

worked example and having access to MAB blocks to aid them, and the third student in each test group using the No visuals worked example with no access to any of the manipulative materials. All the materials used as part of this experimental setup can be found in Appendix B: Experimental materials used in Experiment 2. Prior to commencing the testing, students were given a pre-test to establish their current understanding of place value. The pre-test was composed of eight questions, with guestions focussed on the areas of expanded notation, face value and total value of place value. Following this, students were tested in two phases: the acquisition phase and the test phase. In the acquisition phase, students were presented with four sets of place value questions, each set consisting of a worked example followed by a similar question for the students to solve. Each student was given four minutes for each set of questions: two minutes to study the worked example and a further two minutes to solve the similar question. Therefore, for completing all four sets of problems, each student needed 16 minutes in total. Students in all of the three groups were given the same amount of time in the acquisition phase of the experiment. Students were asked to follow the worked example solution when solving the similar question, with the worked example available to students at all times in the acquisition phase. Students were allowed to have as many attempts as needed at a particular question, as long as their total time per question did not exceed two minutes.

The acquisition phase was followed by the test phase, in which a common test was used to examine the effect, if any, of the acquisition phase on learning the concept of place value. The test phase was composed of seven questions, without any worked examples but similar to the problems used in the acquisition phase. Each student was given a maximum of one minute to solve each of the 7 questions, for a total of 7 minutes. If a question remained unsolved after 1 minute, the child was guided to the next question. The time taken by each student to solve the questions was measured. Each correct answer was worth 1 mark, resulting in a total possible

score of 7 marks. Children did not have access to the worked examples during the testing phase.

7.3 Results

The time taken to complete the pre-test and the score received by each of the students in the pre-test were measured and analysed to understand base-line knowledge in all the groups, and to ensure that there were no significant differences in the cohorts of students between the Abacus (n = 16, M = 33.04, SD = 20.68), MAB blocks (n = 16, M = 34.82, SD = 22.72), and Control (n = 16, M = 33.93, SD = 26.53) groups, F(2, 45) = 0.023, p = .977.

In addition, in the acquisition phase, the variables under analysis were the learning times, and also the learning scores. In the test phase, the variables under analysis were the testing time and the total test scores. The total learning time was another dependent variable used in analysis, and measured the amount of time each student spent studying the worked examples in the acquisition task. Table 5 shows the means and the standard deviations of each of the variables.

	Abacus	MAB blocks	Control	Total
	(N=16)	(N=16)	(N=16)	(N=48)
Pre-Test Time (mins)	3.64 (1.21)	3.37 (0.92)	3.22 (1.10)	3.41 (1.07)
Pre-Test Score %	33.04	34.82 (22.72)	33.93 (26.53)	33.93 (22.95)
	(20.68)			
Learning Time (mins)	12.29 (1.83)	12.63 (1.30)	8.80 (0.72)	11.24 (2.20)
Acquisition Phase	68.75 (1.06)	60.94 (1.26)	64.06 (0.96)	64.58 (1.09)
Score %				
Testing Phase Time	2.7 (1.38)	2.57 (1.42)	2.55 (1.36)	2.61 (1.36)
(mins)				
Testing Phase Score %	50.0 (31.73)	47.32 (32.04)	58.93 (22.05)	52.08 (28.80)

A one-way between groups analysis of variance was conducted to explore the impact of the manipulative material used on place value learning, as measured by the testing phase scores, testing times, learning times and acquisition phase scores. Using the pre-test mark as a covariate, there were no significant differences in the mean testing phase scores between the abacus (n = 16, M = 50.0, SD = 31.73), MAB blocks (n = 16, M = 47.32, SD = 32.04), and Control (n = 16, M = 58.93, SD = 22.05) groups, F(2,44) = 1.135, p = .331, $\eta p^2 = .049$.

There were also no significant differences in the mean acquisition phase scores between the Abacus (n = 16, M = 68.75, SD = 1.06), MAB blocks (n = 16, M = 60.94, SD = 1.26), and Control (n = 16, M = 64.06, SD = 0.96) groups, F(2,44) = 0.046, p = .955, $\eta p^2 = .002$.

No significant differences were also found in the mean testing phase time across the three different groups, Abacus (n = 16, M = 2.7, SD = 1.38), MAB blocks (n = 16, M = 2.57, SD = 1.42), and Control (n = 16, M = 2.55, SD = 1.36) groups, F (2,44) = 0.364, p = .697, $\eta p^2 = .016$.

However, there was a statistically significant difference in the mean learning times, that is, the time taken during acquisition to study the worked examples, between the three groups of Abacus (n = 16, M = 12.29, SD = 1.83), MAB blocks (n = 16, M = 12.63, SD = 1.30), and Control (n = 16, M = 8.80, SD = 0.72) groups, F (2,44) = 40.124, p < 0.0005, $\eta p^2 = .646$. A Tukey post-hoc test showed that the increase in learning times from 8.8 ± 0.72 in the control group to 12.63 ± 1.3 in the MAB blocks group was statistically significant (p < 0.0005); d = 8.45. Additionally, the increase in learning times from the control group (8.8 ± 0.72) to 12.29 ± 1.83 in the Abacus group was also statistically significant (p < 0.0005); d = 6.12. This meant that the students in the control group took less time to study and learn the worked examples, than the students in those groups using manipulative materials.

7.4 Discussion

Informed by the lack of significant findings in Experiment 1, Experiment 2 modified the mode of instructional guidance by using worked examples. Additionally, based on the results in Experiment 1 pre- and post-test scores, the test materials in Experiment 2 were modified to include only face value and expanded notation questions, which were the areas that the learners had the most difficulting in grasping. The procedure of Experiment 2 was subsequently also changed to accommodate the inability of such young students to focus for a prolonged period of time on the same task. It was hypothesized that these changes would result in a significant difference in the use of materials when teaching place value, and, in particular, that using the abacus would be more intuitive to students than the MAB blocks in place value problem solving.

The results of the experiment did not support the hypothesis. There were no significant differences found in the use of either MAB blocks or abacus to support place value instructional material, as indicated by the post-test scores in the testing phase. Throughout the experiment, it was observed that learners were having great

difficulty comprehending the highly text-based instructions given to them for each of the conditions. Some of the students even required explanations of the English terms used in the questions. The difficulty reading the textual information prior to even attempting the mathematical question, possibly resulted in an increase in extraneous load and fewer cognitive resources remaining to cope with the presented mathematical problem. Whilst there were no significant differences between the two manipulative materials, the finding that students took less time to study and learn the worked examples in the control group, in comparison to both the MAB blocks and the Abacus group could be easily explained. The students in the control group were only required to contend with the text-based information, as opposed to the combination of text, pictures and physical material. Therefore the worked examples within the instructional material presented to the learner, even without visuals or manipulative materials, presented such a high cognitive load, that the worked examples were not able to act as substitute schemas to aid in construction of students' own schemas (Kalyuga et al., 2001), and therefore were not effective in producing any improvements in learning.

8 Experiment 3 – The Worked Example and Place Value (Simplified comprehension)

8.1 Introduction

The results of Experiment 2 indicated that there were no significant differences in the performance of students using different manipulative materials. This result can be attributed to the high level of reading comprehension required to understand the mathematical problems presented as worked examples. An assumption based on the results and observations made in Experiment 2 is that the worked examples provided to the students were too complex and were causing a cognitive overload prior to the children attempting to understand the concept of place value.

Therefore, Experiment 3 aimed to reduce the reading comprehension required for the children by presenting the place value problems in a more readable manner, with simplified instructions and explanations. This was expected to help reduce the extraneous load on the students, thereby allowing their limited cognitive resources to be used in understanding and familiarising themselves with the place value worked example.

Additionally, students were tested in groups of three with all three students using the same manipulative material at the same time. This was done to ensure limited distractions for the students, and to avoid any disappointment when no manipulative materials were provided.

The hypothesis of Experiment 3 remained the same as that of Experiment 2. However, the method differed in the instructional material and the way groups were formed. It was hypothesized that a worked example using the child's primary skills of movement and grouping, with the abacus as the manipulative material, would be superior to that of using the MAB blocks as the manipulative material, in learning the secondary skills of place value and number knowledge, particularly when focusing on problems involving expanded notation and face value.

8.2 Method

8.2.1 Participants

Experiment 3 was conducted with forty-five Year 2 students, selected from three private schools in the Eastern and Western suburbs of Sydney. School 1 had a student population of 1206, with 46% of students having a language background other than English. My School (www.myschool.edu.au) classified 81% of the students as being in the top quartile of the Index of Community Socio-Educational Advantaged. It is therefore considered to be a very advantaged school. School 2 had a student population of 2036, with 21% of students having a language background other than English. My School classified 50% of the students as being in the top quartile of the Index of Community Socio-Educational Advantaged. It is therefore also considered to be a fairly advantaged school. The third school had a student population of 1012, with 68% of students having a language background other than English. My School classified 79% of the students as being in the top quartile of the Index of Community Socio-Educational Advantaged. It is therefore considered to be a very advantaged school. Students were all aged approximately 7-8 years, and were randomly assigned to three groups: Abacus worked example (15 students), MAB blocks worked example (15 students), No visuals worked example (15 students).

8.2.2 Materials and Procedure

Students were tested in groups of three, with each group of three students having access to either an abacus (Abacus worked example group), MAB blocks (MAB blocks worked example group), or no manipulative materials at all (No visuals worked example). In addition to having access to the physical materials, the

students' worked examples contained illustrations of the relevant manipulative materials for each of the examples. All the materials used as part of this experimental setup can be found in Appendix C: Experimental materials used in Experiment 3.

Prior to commencing the testing, students were given a pre-test, composed of 7 place value questions, to establish their current understanding of place value. Following this test, students were tested in two phases: the acquisition phase and the test phase. In the acquisition phase, students were presented with four sets of place value questions, each set consisting of a worked example followed by a similar question for the students to solve. Each student was given four minutes for each set of questions: two minutes to study the worked example and a further two minutes to solve the similar question. Therefore, for completing all four sets of problems, each student needed 16 minutes in total. Students in all of the three groups were given the same amount of time in the acquisition phase of the experiment. Students were asked to follow the worked example solution when solving the similar question, with the worked example available to students at all times in the acquisition phase. Students were allowed to have as many attempts as needed at a particular question, as long as their total time per question did not exceed two minutes.

The acquisition phase was followed by the test phase, in which a common test was used to examine the effect, if any, of the acquisition phase on learning the concept of place value. The test phase was composed of 7 questions, without any worked examples but similar to the problems used in the acquisition phase. Each student was given a maximum of one minute to solve each of the 7 questions, for a total of 7 minutes. If a question remained unsolved after 1 minute, the child was guided to the next question. The time taken by each student to solve the questions was measured. Each correct answer was worth 1 mark, resulting in a total possible score of 7 marks. Children did not have access to the worked examples during the testing phase.

8.3 Results

The score received by each of the students in the pre-test was measured and analysed to understand base-line knowledge in all the groups, and to ensure that there were no significant differences in the cohorts of students between the Abacus (n = 15, M = 51.67, SD = 14.07), MAB blocks (n = 15, M = 44.17, SD = 28.69), and Control (n = 15, M = 52.5, SD = 32.46) groups, F (2, 44) = 0.457, p = .636. Additionally, the interaction of the treatment and three different schools used was also found to be non-significant, F (2,44) = 1.127, p = .359).

In the acquisition phase, the variables under analysis were the learning times, the amount of time each student spent studying the worked examples in the acquisition phase, the number of attempts made per question by each of the three groups, and the learning scores. In the test phase, the variables under analysis were the time taken to complete the test and the total test scores. Table 6 shows the means and standard deviations of each of the variables.

	Abacus	MAB blocks	Control	Total
	(N=15)	(N=15)	(N=15)	(N=45)
Pre-test Score %	51.67 (14.07)	44.17 (28.69)	52.50 (32.46)	49.44 (25.97)
Learning Time (mins)	12.44 (1.39)	11.58 (1.30)	9.55 (0.71)	11.25 (1.69)
Acquisition Phase Score	91.67	91.67 (15.43)	100 (0.00)	94.44 (14.97)
%	(20.41)			
Testing Phase Time	2.33 (1.21)	1.63 (0.81)	2.22 (1.64)	2.06 (1.27)
(mins)				
Testing Phase Score %	72.5 (20.70)	65.0 (31.05)	86.31 (17.92)	74.60 (25.00)
Question 1 Attempts	1.2 (0.56)	1.27 (0.59)	1.07 (0.26)	1.18 (0.49)
Question 2 Attempts	1.87 (0.74)	1.40 (0.63)	1.27 (0.59)	1.51 (0.69)
Question 3 Attempts	1.07 (0.26)	1.27 (0.46)	1.27 (0.46)	1.20 (0.40)
Question 4 Attempts	1.27 (0.46)	1.33 (0.82)	1.33 (0.49)	1.31 (0.60)

Table 6: Descriptive Statistics for the variables

A one-way between groups analysis of variance was conducted to explore the impact of the use of manipulative materials on place value learning, as measured by the testing phase scores, testing times, and learning times and acquisition phase scores. Using the pre-test mark as a covariate, there were significant differences in the mean testing phase scores between the Abacus (n = 15, M = 72.5, SD = 20.7), MAB blocks (n = 15, M = 65.00, SD = 31.05), and Control (n = 15, M = 86.31, SD = 17.92) groups, F(2, 44) = 3.180, p = .05, $\eta_p^2 = .134$. A follow up Tukey post-hoc test was run to find that the significant difference was between the MAB and Control groups, with p = .049. Additionally, using the learning time as another covariate, there were significant differences in the mean test phase scores between the Abacus (n = 15, M = 72.5, SD = 20.7), MAB blocks (n = 12, M = 68.75, SD = 33.5), and Control (n = 13, M = 84.2, SD = 18.4) groups, F(2, 39) = 3.315, p < .05, $\eta_p^2 = .159$. A follow up post-hoc test using the LSD adjustment was run to discover the significant difference between the Abacus and Control group with p = .024, and MAB and Control group with p = .022.

There were no significant differences in the mean acquisition phase scores between the Abacus (n = 15, M = 91.67, SD = 20.41), MAB blocks (n = 15, M = 91.67, SD = 15.43), and Control (n = 15, M = 100, SD = 0) groups, F(2,44) = 1.424, p = .252, $\eta_p^2 = .065$.

Additionally, no significant differences were found in the mean testing phase time across the three different groups, Abacus (n = 15, M = 2.33, SD = 1.21), MAB blocks (n = 15, M = 1.63, SD = 0.81), and Control (n = 15, M = 2.22, SD = 1.64) groups, F(2,44) = 0.67, p = .201, $\eta_p^2 = .075$.

However, there was a statistically significant difference in the mean learning times, that is the time taken during acquisition phase to study the worked examples, between the three groups of Abacus (n = 15, M = 12.44, SD = 1.39), MAB blocks (n = 15, M = 11.58, SD = 1.30), and Control (n = 15, M = 9.55, SD = 0.71), F(2,44) = 21.207, p < .01, $\eta_{p^2} = .541$. A Tukey post-hoc test showed that the increase in learning times from 9.55 ± 0.71 in the Control group to, 11.58 ± 1.30 in the MAB

blocks group was statistically significant (p < .01). Additionally, the increase in learning times from the Control Group (9.55 ± 0.71) to 12.44 ± 1.39 in the Abacus group was also statistically significant (p < .01). This could indicate that it took less time to study the worked examples presented in the control group, than those groups that used manipulative materials.

8.4 Discussion

Experiment 3 was designed to replicate Experiment 2, with modifications to the learning materials given to the students in the acquisition phase. Students were required to solve place value problems using either an abacus, MAB blocks, or no manipulative materials for the control group. All students were first presented with worked examples of place value problems, that were also illustrated by using an abacus, MAB blocks or no manipulative materials, depending on the group they were assigned to. It was hypothesized that, with the reduced reading comprehension required, students would be able to use their limited cognitive resources to process the instructional place value material. This increase of working memory resources available to students was expected to be reflected in better post-test performance. Additionally, it was anticipated that due to the greater one-to-one correspondence of the abacus beads in relation to countable objects, and the intuitive grouping of the beads in tens, the abacus would make a superior reprepresention of place value in comparison to MAB blocks.

The results of Experiment 3 only partially supported this hypothesis. Firstly, the reduced time taken by the control group, in comparison to both MAB block and abacus groups, to study the worked examples, can be explained. Because the instructional material for the control group was not accompanied by the use of manipulative materials, or relevant illustrations, the group was required to process less information. Since children were only subjected to numerical information, they

were able to process the information faster due to an increase in their available working memory resources.

Secondly, whilst no differences were found between the use of the MAB blocks and an abacus in number decomposition tasks, a Tukey post-hoc test showed a significant difference in the post-test results between the group using MAB blocks and the control group. The students in the control group, who were given no access to physical manipulative material or their visual representation, performed better in the post-test than those students that had access to, and were given visual representations of, MAB blocks to represent place value problems. Such a difference could be due to the potential redundancy effect the MAB blocks may have caused to the student. The numerical information and the visual representation of the MAB blocks were each sufficient on their own to allow the student to understand the concept, yet they are often presented in conjunction with each other in the classroom and in instructional material. This leads to the basis of Experiment 4, which aims to explore whether the use of MAB blocks, both visually and physically, may lead to the redundancy effect, which in turn may have a negative impact on learning.

9 Experiment 4 – The Worked Example, Place Value and Manipulative Material Redundancy

9.1 Introduction

The results of Experiment 3 indicated that worked examples without MAB manipulatives may be beneficial to young primary-aged children establishing their early understanding of place value concepts. However, these worked examples needed to be sufficiently simple in terms of the language and reading comprehension skills required not to overwhelm the student's limited cognitive resources, and yet appropriately difficult to activate the learning pathways.

Students in the MAB block worked example group were provided with illustrations of blocks to demonstrate number deconstruction and were also allowed to use physical MAB blocks to help construct and manipulate given numbers. Similarly, children in the Abacus worked example group were provided with illustrations of an abacus to demonstrate number deconstruction and were also allowed to use the physical abacus to help construct and manipulate given numbers. The third group of children, the Control worked example group were provided with no illustrations or manipulative materials to assist with number construction and manipulation. The children in the Control worked example group performed better in the post-test, in comparison to the students in the MAB worked example group. There were no significant differences between the Abacus worked example group and the Control worked example group.

Since no significant differences in performance were found between the two types of manipulative materials used, it was decided that, in Experiment 4, only MAB blocks will be utilised. The MAB blocks were chosen due to their popularity and widespread use in Australian classrooms. It was also observed in Experiment 3, that the large quantity of blocks required to construct three- and four-digit numbers would often interfere with students being able to keep track of such numbers and their associated blocks. Students using blocks often made counting and handling errors. These errors were likely related to the amount of information that was required to be processed by their working memory at any one time. Keeping track of both the blocks necessary for the numerical deconstruction and also the concept of place value that the blocks represent, may have resulted in the complete exhaustion of their limited cognitive resources. The instructional material, in the form of worked examples, was therefore modified in Experiment 4, to include simpler numbers that did not require a large number of blocks in construction.

Experiment 4 aimed to explore the impact on children's learning and acquisition of the concept of place value, specifically number deconstruction, when using MAB blocks in worked examples, both illustratively and physically. It was hypothesized that children who did not have access to any manipulative materials would perform better in the testing phase, as measured by the post-test, post-test transfer and delayed test, than those children given access to MAB blocks.

9.2 Method

9.2.1 Participants

Experiment 4 was conducted with fifty-two Year 2 students, selected from two private schools in the Eastern suburbs of Sydney. According to the My School website (<u>www.myschool.edu.au</u>), the first school had a student population of 1029, with 14% of students having a language background other than English. My School classified 76% of the students as being in the top quartile of the Index of Community Socio-Educational Advantaged. It is therefore considered to be a very advantaged school. School 2 had a student population of 777, with 7% of students having a language

background other than English. My School classified 77% of the students as being in the top quartile of the Index of Community Socio-Educational Advantaged. It is therefore considered to be a very advantaged school, similarly to the first school. Students were all aged approximately 7-8 years, and were randomly assigned to two groups: MAB blocks worked example (24 students), No visuals worked example (28 students).

9.2.2 Materials and Procedure

Students were tested in groups of three, with each group of three students having access to either the MAB blocks (MAB blocks worked example group), or no manipulative materials at all (No visuals worked example). Students in the MAB blocks group, had access to the physical materials as well as illustrations of MAB blocks in the paper-based worked examples. The second group, No visuals worked example group, were provided with the same worked examples, except no access was given to physical blocks, and no illustrations were included with their examples. All the materials used as part of this experimental setup can be found in Appendix D: Experimental materials used in Experiment 4.

Prior to commencing the testing, students were given a pre-test, composed of 7 place value questions, to establish their current understanding of place value. Following this test, students participated in two phases: the acquisition phase and the test phase. In the acquisition phase, students were presented with 9 place value statements, each of which detailed how to deconstruct a number up to a hundred, increasing by tens, such as 19, 29, 39, and so on. There were no paired questions associated with these worked examples during the acquisition phase. Each student was given two minutes per statement to study the worked example, and to construct the number using MAB blocks, if they were in the MAB block group. Therefore, for completing all nine problems, each student needed 18 minutes in total. Students in all of the three groups were given the same amount of time in the acquisition phase of the experiment. Students were allowed to have as many attempts as needed to construct a particular numerical value within a question, so long as their total time per question did not exceed two minutes. At the conclusion of their allocated question time, if the student had not constructed the correct numerical value, a solution was demonstrated to them.

The acquisition phase was followed by the test phase, in which a common test was used to examine the effect, if any, of the acquisition phase on learning the concept of place value. The test phase was composed of 10 post-test questions, without any worked examples, but similar to the problems used in the acquisition phase using different numerical values. Each student was given a maximum of 1 minute to solve each of the 10 questions, for a total of 10 minutes. If a question remained unsolved after 1 minute, the child was guided to the next question. Each question was worth 1 mark, resulting in a total possible score of 10 marks. Children did not have access to the worked examples during the testing phase. The post-test was then followed up by a post-transfer test, where 6 questions needed to be answered. The questions were different in structure to the worked examples, but in answering them, a student may be able to use information that could have been learnt from the worked examples. Each student was again given 1 minute for each question, and were able to make as many attempts as needed, as long as the overall time did not exceed 1 minute.

One week later, a delayed test was given to the participating students to ascertain whether any delayed place value knowledge existed. The delayed test was composed of sixteen questions. The first 10 questions directly related to the worked examples that were given to the students a week earlier. The subsequent 6 questions did not directly relate to the worked examples, but required the knowledge from the worked examples to be extended. Each student was given 1 minute per question, for a total of sixteen minutes to be spent on the delayed test. The students were allowed to make as many attempts as they needed to respond to the questions, as long as their overall time did not exceed 1 minute per question, with sixteen minutes in total to complete the test.

9.3 Results

The score received by each of the students in the pre-test was measured and analysed to understand base-line knowledge in all the groups, and to ensure that there are no significant differences in the cohorts of students between the MAB blocks (n = 24, M = 69.79, SD = 27.783), and Control (n = 28, M = 79.76, SD = 16.582) groups, F(1, 51) = 2.551, p = .117, $\eta_p^2 = .049$.

The variables under analysis in the test phase were the scores for the posttest, the post-test transfer, and the delayed test scores. Table 7 shows the means and standard deviations of each of these variables.

Table 7: Descriptive Statistics for the variables

	MAB blocks	Control	Total
	(N=24)	(N=28)	(N=52)
Pre-test Mark %	69.79 (27.78)	79.76 (16.58)	75.16 (22.78)
Post-test Mark %	86.39 (30.07)	87.8 (20.18)	87.15 (24.98)
Post-test Transfer Mark %	73.04 (30.46)	77.19 (24.26)	75.27 (27.1)
Delayed Test Mark %	n = 19	n = 25	n = 44
	79.79 (29.98)	81.44 (18.33)	80.73 (23.75)

A one-way between groups analysis of variance was conducted to explore the impact of the redundancy effect on place value learning, as measured by the testing phase scores, including the post-test scores, post-test transfer scores and the delayed test scores. Using the pre-test mark as a covariate, there were no significant differences in the mean testing phase scores in the post-test between the MAB blocks (n = 24, M = 86.39, SD = 30.07), and Control (n = 28, M = 87.8, SD = 20.18) groups, F(1, 51) = 0.688, p = .411, $\eta_p^2 = .014$. Additionally, there were no significant differences in the post-test transfer results between the MAB blocks (n = 24, M = 30.46), and Control (n = 28, M = 77.19, SD = 24.26) groups, F(1, 51) = 0.688, p = .415, M = 77.19, SD = 24.26) groups, F(1, 51) = 0.688, p = .415, M = 77.19, SD = 24.26) groups, F(1, 51) = 0.688, p = .415, M = 77.19, SD = 24.26 groups, F(1, 51) = 0.688, p = .415, M = 77.19, SD = 24.26 groups, F(1, 51) = 0.688, p = .415, M = 77.19, SD = 24.26 groups, F(1, 51) = 0.688, p = .415, M = 77.19, SD = 24.26 groups, F(1, 51) = 0.688, p = .415, M = 77.19, SD = 24.26 groups, F(1, 51) = 0.688, p = .415, M = 77.19, SD = 24.26 groups, F(1, 51) = 0.688, P = .415, M = 77.19, SD = 24.26 groups, F(1, 51) = 0.688, P = .415, M = 77.19, SD = 24.26 groups, F(1, 51) = 0.688, P = .415, P = 0.688, P = .416, P = 0.688, P = 0.68
0.420, p = .520, $\eta_p^2 = .008$. The delayed test was completed the week after the posttest and the post-test transfer. Due to absences of some children on that day, the delayed test could only be given to 44 children, 19 of those in the MAB worked example group and 25 of those in the Control group. Similarly to the post-test and the post-test transfer results, no significant differences were found in the delayed test results between the MAB block group (n = 19, M = 79.79, SD = 29.98) and the Control group (n = 25, M = 81.44, SD = 18.32), F(1, 43) = 2.139, p = .151, $\eta_p^2 = .050$.

9.4 Discussion

Experiment 4 was designed to test the hypothesis that the utilisation of MAB blocks in worked examples in Year 2 place value material may induce a redundancy effect and could therefore lead to difficulties in learning the concept of place value effectively. Students were asked to study worked examples that demonstrated numbers up to the hundreds jumping in groups of tens. All students were instructed to say these number skips out loud and, if in the MAB blocks group, were also required to construct the numbers using the MAB blocks. It was expected that students not using the MAB blocks would perform better than those students that had access to MAB blocks, as measured by the post-test, the post-test transfer and the delayed test scores in the testing phase.

The results of Experiment 4 did not support this hypothesis. The high scores of the students in the pre-test, post-test, post-test transfer and delayed test in the testing phase of the experiment indicated a high level of prior knowledge. This demonstrates the presence of the ceiling effect, with students unable to acquire any further knowledge. The high level of student performance in the test phase meant that a better alignment between the difficulty of the material and the expertise of the students was necessary. The New South Wales Mathematics K-10 Syllabus (Board

of Studies Teaching and Educational Standards - NSW, n.d.), defines the ability of students to count collections to a hundred by partitioning the numbers using place value, as an outcome required from Stage 1.2 (Year 2). However, since the experimental data was collected in the later stages of Year 2 (at the conclusion of Term 3 and Term 4), students might have already built up enough prior knowledge to learn the concepts easily. This prior knowledge would have meant that they were able to complete the questions in the pre-test and the test phases without a a great deal of learning taking place in the acquisition phase. This was one possible reason for misjudging the prior knowledge base of the students.

It was expected that repeating this experiment with a different cohort of learners and using a more complex set of questions would yield significant differences between the MAB worked example group and the control group. By using learners who are at Stage 2.1 of schooling (end of Year 3), and increasing the complexity of the testing material to the expected learning outcome of Stage 2.2 (Year 4) to *'use place value to partition numbers of up to five digits and recognise this as 'expanded notation''* (Board of Studies Teaching and Educational Standards - NSW, n.d.; NESA, 2015), could circumvent the ceiling effect observed in Experiment 4. This hypothesis was tested in Experiment 5.

10 Experiment 5 - The Worked Example, Place Value and Manipulative Redundancy (increased expertise and complexity of instructional material)

10.1 Introduction

Experiment 4 demonstrated no statistical significance between the group using MAB blocks, illustratively and physically, and the control group. One possible explanation for this lack of significance could be a higher level of prior topic knowledge than was initially observed with children at this stage of schooling. This indicated that either the level of expertise of the students needed to be changed to include more novice learners, or the complexity of the material needs to be increased, so that students were provided with new material, of which they had little or no prior experience.

In Experiment 5, the complexity of the material was increased to include number decomposition into the tens of thousands. This level of number comprehension was associated with Stage 2.2 (end of Year 4) of learning, as prescribed by the New South Wales Mathematics K-10 Syllabus (NESA, 2015). Due to the extreme complexity of material, students were selected at the conclusion of Year 3 to participate in the experiments in the first two weeks of Term 1 of Year 4. This was done to ensure that numbers of such magnitude had not yet been covered in the classroom and that students had no prior knowledge of how to decompose and bridge such numbers into the thousands and the tens of thousands. This was to be established by the pre-test. The learning materials of Experiment 5 were modified not only to include the more complex number deconstructions, but also to add a paired problem to each of the worked examples to help with concept acquisition. It was hypothesized in Experiment 5 that the students who were presented with illustrations of MAB blocks and required to use physical MAB blocks, to assist with

number construction and decomposition, would be hindered in their learning by the redundancy effect caused by the blocks.

10.2 Method

10.2.1 Participants

Experiment 5 was conducted with thirty-eight Year 4 students, selected from two private schools in the Eastern suburbs of Sydney. According to the My School website (www.myschool.edu.au), the first school had a student population of 1029, with 14% of students having a language background other than English. My School classified 76% of the students as being in the top quartile of the Index of Community Socio-Educational Advantaged. It is therefore considered to be a very advantaged school. School 2 had a student population of 777, with 7% of students having a language background other than English. My School classified 77% of the students as being in the top quartile of the Index of Community as being in the top quartile of the Index of Community Socio-Educational Advantaged. It is therefore considered to be a very advantaged school. School 2 had a student population of 777, with 7% of students having a language background other than English. My School classified 77% of the students as being in the top quartile of the Index of Community Socio-Educational Advantaged. It is therefore considered to be a very advantaged school, similarly to the first school. Students were all aged approximately 8-9 years, and were randomly assigned to two groups: MAB blocks worked example (20 students), No visuals worked example (18 students).

10.2.2 Materials and Procedure

Students were tested in groups of three, with each group having access to either MAB blocks (MAB blocks worked example group), or no manipulative materials (No visuals worked example). The MAB blocks groups was given both illustrations and physical MAB blocks to manipulate. The Control group did not have any illustrations included with their acquisition text, but in all other respects the questions and text were identical to the MAB blocks group. All the materials used as part of this experimental setup can be found in Appendix E: Experimental materials used in Experiment 5.

Prior to commencing the testing, students were given a pre-test, comprised of 7 place value questions, in order to establish their current understanding of place value. Following this test, students participated in two phases: the acquisition phase and the test phase. In the acquisition phase, students were presented with 10 guestions based on number expansion in place value and the solutions thereto. Each guestion and solution detailed how to deconstruct a number up to tens of thousands. Each worked example was followed by a paired question, using different numerical values but otherwise the same as the worked example. Students were allowed to refer back to the worked example when solving the paired problem. Each student in the MAB blocks group was given 2 minutes per question to study the worked example and then to construct the illustrated number using MAB blocks. Students in the Control group were given 2 minutes to study each worked example. A further 1 minute was provided to students in both groups to solve the paired problem. Therefore, for completing all 10 paired sets of problems, each student needed thirty minutes in total. Students in the two groups were given the same amount of time in the acquisition phase of the experiment. Students were allowed to have as many attempts as needed at constructing a particular numerical value within a question, as long as their total time per question did not exceed two minutes. At the conclusion of their allocated question time, if the student had not constructed the correct numerical value, a solution was demonstrated to them.

A test phase followed the acquisition phase, in which a common test was used to examine the effect, if any, of the acquisition phase. The test phase was comprised of a post-test, with 10 questions, similar to the problems used in the acquisition phase, but using different numerical values and without any worked examples. Each student was given a maximum of 1 minute to solve each of the 10 questions, for a total test time of 10 minutes. If a question remained unsolved after 1 minute, the child was guided to the next question. Each question was worth one mark, resulting in a total possible score of 10 marks. Children did not have access to the worked examples during the testing phase. The post-test was then followed up by a post-transfer test, where 6 questions needed to be answered. The questions were different in structure to the worked examples, but in answering them, a student may be able to use information that could have been learnt from the worked examples. Each student was again given 1 minute for each question, and were able to make as many attempts as needed, as long as the overall time did not exceed 1 minute.

One week later, a delayed test was given to the participating students to ascertain whether any delayed place value knowledge existed. The delayed test was composed of sixteen questions. The first 10 questions directly related to the worked examples that were given to the students a week earlier. The subsequent 6 questions did not directly relate to the worked examples, but required the knowledge from the worked examples to be extended. Each student was given 1 minute per question, for a total of sixteen minutes to be spent on the delayed test. The students were allowed to make as many attempts as they needed to respond to the questions, as long as their overall time did not exceed 1 minute per question, with sixteen minutes in total to complete the test.

10.3 Results

The variables under analysis in the test phase were the scores for the post-test, the post-test transfer, and the delayed test scores. Table 8 shows the means and standard deviations of each of these variables.

	MAB blocks	Control	Total
	(N=20)	(N=18)	(N=38)
Post-test Mark %	74 (0.883)	91.1 (0.758)	82.1 (1.189)
Post-test Transfer Mark %	62.5 (1.803)	62.2 (2.045)	62.4 (1.895)
Delayed Test Mark %	79.0 (1.252)	95.6 (0.784)	8.68 (1.338)

Table 8: Descriptive Statistics for the variables

A one-way between groups analysis of variance was conducted to explore the impact of the redundancy effect on the use of manipulative materials in place value learning, as measured by the testing phase scores, including the post-test scores, post-test transfer scores and the delayed test scores. Using the pre-test mark as a covariate, there were statistically significant differences in the mean testing phase scores in the post-test between the MAB blocks (n = 20, M = 74.0, SD = 0.883), and Control (n = 18, M = 91.1, SD = 0.758) groups, F(1, 37) = 52.883, p < 0.01. There were no significant differences in the post-test transfer results between the MAB block group (n = 20, M = 62.5, SD = 1.803), and Control group (n = 18, M = 62.2, SD = 2.045), F(1, 37) = 0.005, p = .946. The delayed test was completed a week following the post-test transfer. There were no children absent in this instance and all children who participated the previous week were able to complete the delayed test. Statistically significant differences were found in the delayed test performance marks between the MAB block group (n = 20, M = 79.0, SD = 1.252) and the Control group (n = 18, M = 95.6, SD = 0.784), F(1, 37) = 25.872, p < 0.01.

10.4 Discussion

The purpose of Experiment 5 was to test whether the use of MAB blocks, both visually within worked examples, and physically when students are using these materials hands-on, could lead to a redundancy effect that could have a negative impact on learning outcomes. The method used in Experiment 5 repeated that of Experiment 4, but used testing material that was higher in complexity. This was done to ensure that the worked example effect could be obtained, which could only occur if the presented material was adequately complex for the learner.

The results of Experiment 5 showed improved performance of the control group in the testing phase, as measured by the mean post-test and the delayed test marks. Within the cognitive load theory framework, this finding could be explained by the redundancy effect (Sweller, 2012). The use of manipulative materials, in the

form of MAB blocks, to explain place value, could lead to an increased extraneous load and therefore could be detrimental to learning. According to the redundancy effect, if learners are presented with a diagram and text simultaneously that provide exactly the same information, the redundancy effect may be triggered, causing a lack of working memory resources to process the necessary information (P. Chandler & Sweller, 1991). Since Experiment 5 used illustrations and textual information, in addition to physical use of blocks, this could have triggered the redundancy effect and led to a high extraneous load on the student, based on the instructional material. Learners were not able to ignore the partly split information, potentially neutralizing the redundancy effect caused by the MAB block illustrations. because they were also required to construct the number decompositions with the physical blocks, using both the illustrative and written numerical information. Students were thus subjected to the potential redundancy of the blocks with each of the questions in the acquisition phase. The sources of information presented to the learner were self-contained, that is they could each be understood on their own, which would imply the presence of the redundancy effect. However, these types of questions are often presented to the student in this way in their normal learning environment, in an Australian classroom. In this instance, the use of MAB blocks in place value instructional material may have presented a redundancy to student learning. The objects increased the extraneous load by repeatedly drawing children's attention away from the task, thereby consuming limited working memory resources.

11 General Discussion

11.1 Summary of key findings

Mathematics curricula are generally designed on the assumption that instruction based on manipulative materials should precede instruction using mathematical symbols. Accordingly, formal manipulative materials are frequently employed in the classroom from the very early stages of education. Instructional techniques using manipulative materials give students the opportunity to interact with physical objects to learn more about abstract mathematical concepts. However, literature remains ambivalent on the advantages of such materials. Gürbüz (2010) and Sherman & Bisanz (2009) carried out studies which found the use of these materials to be an effective approach to improving student's achievement in mathematics. Other studies, however, have found that the use of manipulative materials either made no difference to students' performance or actually inhibited their performance (Canobi et al., 2003; Carbonneau, Marley & Selig 2013a; Uttal et al., 1997; Sowell 1989; McNeil et al., 2009; Amaya et al., 2008). These contradictions in research results may arise due to the variability in the types of manipulative materials used, the method of instructional guidance and in potential characteristics of the learning environment and the learner (Carbonneau, Marley & Selig 2013b; Carbonneau & Marley 2015).

One of the possible problems posed by the use of manipulative materials is their dual representation (Uttal et al., 1997). The materials not only have to be seen as objects in their own right, but also have to symbolise the concept or procedure they are attempting to demonstrate. Translating between these physical objects and the abstract concepts they demonstrate can be a difficult process, and one that requires intensive use of the limited cognitive resources that children possess. When children interact with manipulative materials, they are largely focused on the objects in their hands, thereby committing their cognitive resources to manipulating them, and understanding the abstract concept represented by the manipulative materials. This process may not leave sufficient cognitive resources for the processes required to understand the actual mathematical concepts (Boulton-Lewis 1998).

MAB blocks are one of the most common manipulative materials used in the classroom. These blocks are designed to assist in the teaching of place value, with blocks grouped into ones, tens, hundreds, thousands and so on. These groups are based on the size of the block, and are often referred to as a 'unit' for a single block, a 'long' for a group of ten blocks, and a 'flat' for a group of a hundred blocks. Students are not always aware that there are ten blocks stuck together in a long, and a hundred blocks stuck together in a flat or a thousand small cubes stuck together to make a thousand block. Using language such as ones, longs, flats is also not helpful in developing a link between the relevant block and the number that it is meant to represent. This can lead to a further disconnection between the manipulative material and the mathematical concept that it aims to represent.

Another example of a base-10 manipulative material is a Slavonic abacus. The Slavonic abacus is made up of ten rows of beads, with ten beads on each of the rows. The beads are coloured in such a way, that each row and each column is made up of five beads of one colour and five beads of another colour. Each row of beads corresponds to the fingers on the children's hands, that is, five beads of one colour can count as the fingers on one hand, and the second colour beads on the same row represent the fingers on the second hand. This ensures that children are easily able to subitise both in the ones and in the tens, up to one hundred. The Slavonic abacus can offer children a way to envision a number, as a whole, with

each beads correlating to one single count. This may provide a more intuitive way to count and to represent numbers up to one hundred.

This thesis proposes that MAB blocks are not intuitive to students' basic number schemas and can, therefore, lead to the redundancy effect. Students often learn to copy a particular procedure, such as number expansion, to reach a specific goal using MAB blocks, and do not relate these objects to the numbers themselves. With the aim to contribute further to the body of research on the use of manipulative materials in learning mathematical concepts, this study specifically focused on investigating the use of manipulative materials in learning the concept of place value, taking into consideration cognitive load theory and its associated effects (Sweller et al., 1998).

Five experiments were carried out with students in Years 2 to 4 as participants. Experiment 1 investigated the impact a gradual (one stage at a time) or an all-at-once (all stages at the same time) transition had with two types of manipulative materials, an abacus or MAB blocks, on learning the concept of place value. Students were randomly assigned to one of four groups, in a 2 (gradual vs. all-at-once transition) \times 2 (abacus vs. MAB blocks) mixed factorial design. It was proposed that using an abacus as a manipulative material, and using a gradual transition across the five stages of place value understanding, would be a more effective way to learn the concept. The proposition was based on the argument that the abacus would be a more intuitive material that could aid in schema construction, due to its easy representation of the base-10 system. In addition, it was proposed that the gradual transition would ensure no cognitive overload in learning. However, these arguments were not supported by the findings, showing that there were no significant differences in the results regardless of the variables under test.

Experiment 1 showed no significant differences between the gradual intervention and the all-at-once teaching intervention. Experiment 2, therefore, built on this finding by instead using the worked example effect to provide a consistent

intervention across the groups of students. Three groups of students were used in this experiment. The first group was assigned the abacus, the second group used MAB blocks and the control group was not given any manipulative materials. The test material focused on only the last stage of place value understanding, number expansion and decomposition, which was the area where students had the most difficulty in Experiment 1. Experiment 2 was designed to investigate whether using the abacus in learning place value would be a more effective method for schema construction due to its one-to-one count correspondence and easy division into groups of tens beads. The results of this experiment did not support the hypothesis, as no significant differences were found amongst the three groups.

Experiment 3 was based on the observation that the worked examples presented to the students in Experiment 2 were too cognitively demanding, in terms of the level of English comprehension required, thereby causing cognitive overload before any learning could take place. Accordingly, Experiment 3 repeated Experiment 2, but with worked examples that did not require a high level of English comprehension. As in Experiments 1 and 2, Experiment 3 was designed to compare the efficiency of using an abacus versus MAB blocks to learn the concept of place value. The results of the experiment showed that the control group (no manipulative materials) performed better than the group using an abacus and the group using MAB blocks. Additionally, no significant differences were observed between the group using an abacus and the control group. This result thus led to the proposition that the use of MAB blocks could potentially lead to the redundancy effect, thereby causing ineffective learning.

In order to further investigate the difference found in Experiment 3 between the group using MAB blocks and the control group, Experiment 4 was designed to include a delayed learning test in addition to an immediate post-test, given to the students a week after the intervention. Instructional material was modified to include more questions that focused on expanded number notation, thereby forcing the students to interact with the MAB blocks. The rationale of this experiment was based on the assumption that the use of MAB blocks led to the redundancy effect and therefore hindered effective learning of the concept of place value. Based on this, it was predicted that the students in the control group would perform better than those using the MAB blocks, leading to better outcomes in the post-test and the delayed learning test. The results did not support this argument, showing no significant differences between the group using MAB blocks and the control group. However, this result could be due to the ceiling effect observed throughout this experiment. Children scored very highly on both the pre-test and the post-test, thereby indicating that no actual learning had taken place. Due to the timing of this experiment at the conclusion of the school year, students had already become familiar with the concepts used in the acquisition phase, and so the worked examples did not display novel concepts for these students. Since the worked examples did not present novel information, no learning could take place.

Experiment 5 then built on this information by increasing the expertise of students and increasing the complexity of the instructional material. This was done by using more difficult worked examples, with numbers into the thousands and tens of thousands, and additionally breaking them up in unconventional ways to test for a deeper understanding of the concept of place value. Students were tested at the beginning of term 1 of the following school year and, therefore, had not yet learnt the material included in the worked examples. Two groups were used as part of this experiment, one group using MAB blocks, and the second group using no manipulative materials. The argument proposed by this experiment, similar to that of Experiment 4, was that MAB blocks, used both in worked examples and physically, could lead to the redundancy effect in the instructional material, resulting in ineffective learning. The findings of this final experiment showed significant differences in the performance of those students using MAB blocks and those not using any manipulative materials. Students performed better in the post-test and the delayed test when no manipulative materials were used to support the worked

examples. This could indicate that the use of MAB blocks, physically and as part of worked examples, might have led to the redundancy effect and to less effective instructional material. In addition, it was observed that MAB blocks were often used in a task-irrelevant way, distracting the student. This could also result in the redundancy effect being observed. This is the most significant finding to emerge from this study showing that learning to handle manipulative materials, specifically MAB blocks, when learning the concept of place value, may not lead to a deeper understanding of this concept and improvements in student performance. The finding is grounded in cognitive load theory, as the use of MAB blocks may result in the redundancy effect and cause difficulty for students translating between the blocks and the concept of place value they are designed to demonstrate, due to the limited cognitive resources available to the student.

11.2 Theoretical Implications

Recent work by Mix et al. (2016) found mixed results when using MAB blocks to help teach children the concept of place value. The study found that, for some children, there was a clear advantage using the blocks, whilst for others, the blocks were only beneficial once they had already learnt the concept of place. The study found that one factor influencing the benefits of using the blocks was the number of years that children had been trained in their use. The more training the chidren had had, the more beneficial it was to use the blocks. Another study (Mix et al., 2013), found the meaning of MAB blocks was not always obvious to children, requiring some prior knowledge of number to be explicit in the student's mind, before the blocks could be understood. This defeats the purpose of using blocks to teach the concept of place value, if children might need to have a basic understanding of place value in order to learn how to use the blocks.

The aim of this study was to consider the manipulative materials, specifically MAB blocks, from the viewpoint of cognitive load theory. Based on the findings of

this study, there are two implications arising from the use of MAB blocks in teaching mathematical concepts. Firstly, research has shown that children may find it very difficult to transfer knowledge from the manipulative material to its symbolic written form they are meant to represent. This is in line with the current research showing that knowledge gained in one context may not transfer to another context, because of the mismatch between the format in which the material was first encoded and the way in which it is needed to be recalled (Uttal et al., 2013). The difficulty of such a general transfer in memory is also linked to the well-documented difficulty of maintaining the dual representation of a concrete object and the abstract concept it is demonstrating (Uttal et al., 1997; DeLoache 2000; Uttal et al., 2013). It was observed throughout the experimentation phase that children would often not use the blocks for their intended purpose, to help them solve a particular question. In fact, the children were often observed noting an answer down and then, as an afterthought, because the question required them to do so, constructing the number expansion using blocks. Often, the block response would be correct, but would not match the abstract symbolic representation written on paper by the student. This indicated, firstly, that blocks and symbols were seen as concepts in their own right and not linked, and, secondly, that difficulties were experienced translating between blocks and their symbolic place value representation.

The results of this research also extend the findings of some recent studies on the efficacy of manipulative materials in teaching mathematical concepts at an early stage of education (Uttal et al., 2013; Kamina & Iyer, 2009; Carbonneau & Marley, 2015; McNeil & Jarvin 2007; Moyer & Jones, 2004; Uttal et al., 1997) by examining them in the context of cognitive load theory. Recent research (K J Carbonneau & Marley, 2015) indicates that the use of manipulative materials can also be ineffective due to problems with dual representation. In addition, the current study has shown that the use of MAB blocks to teach the concept of place value may also contribute to the redundancy effect, which may reduce their efficacy in teaching this concept. In particular, Experiment 5 indicated that the students in the control group (without any manipulative materials) performed better solving place value problems of number expansion and decomposition, due to the potential redundancy effect created by the MAB blocks in the second group. Therefore, cognitive load factors should be considered when selecting optimal instructional strategies in this area.

11.3 Educational Implications

The results of this study may challenge the well-established practice of using MAB blocks to demonstrate the concept of place value to children to help them understand the base-10 number system. Firstly, there was an issue with dual representation. Students perceived the blocks as objects in their own right, without having any association with the number concepts that the blocks were meant to demonstrate. Hence, a problem with translating from a concrete object to the abstract concept of number was observed. Construction with blocks including bundling and grouping is a primary skill, and the process of using the blocks in this way comes intuitively. However, in this study the blocks did not directly translate to the secondary skill of mastering the base-10 number system. Such a system needs to be explicitly taught as a concept in its own right.

Alternatively, teaching such a concept requires instructional material comprising three components: learning to play with the blocks, learning how the number system is structured and, thirdly, translating between these two representations. Indeed, when using either the blocks or the abacus, it was often observed that children engaged in task-irrelevant activities, such as using blocks for construction, and making music with an abacus. It could be beneficial to abandon the use of these manipulative materials and persevere with explicit instruction in how the number system works. From observations of students in Year 2, it would appear that they might not yet have the general schemas required to build any deep understanding of the base-10 number system. It could be appropriate for such concepts to be taught at a later stage and, in the meantime, there could be a greater

focus on creating better number schemas in memory, that could aid in the construction, manipulation and processing of numbers at later stages.

It was observed in the experiments of this study that children could successfully carry out routine tasks, such as pointing to a number in the tens or hundreds column, without actually understanding the concept of number. It was obvious that children had simply learnt by rote the processes required to complete such tasks, without showing any deep understanding of the concept of place value. For example, the link between MAB blocks and their direct numerical representation should not be assumed knowledge in young children. Learners did not understand the concept that the blocks were designed to illustrate, instead treating the blocks independently of the numbers. Students manipulated the blocks, based purely on their knowledge of blocks, as opposed to any relationship that these blocks might have had to the numbers.

The results of this study might also have more general educational implications, which leads to the second, and perhaps most important, point. This study has indicated that using MAB blocks with worked examples may lead to the redundancy effect, and, therefore, their use may be detrimental to learning. Worked examples are often presented in this way in instructional material for students. Although the redundancy effect and its repercussions are well researched in literature, there is not much data dealing with this effect in younger students. This study has indicated that using simple worked examples, requiring simple reading comprehension skills, and with no redundant illustrations, could be the most beneficial means for a young student learning the concept of place value. The need for simple instructions that require only basic comprehension skills can be attributed to young learners still developing their secondary skill of reading. Complicated instructions that require complex comprehension skills use up limited cognitive capacity that would otherwise be used in learning mathematical concepts. The efficacy of not using redundant illustrations or tools to teach place value concepts was demonstrated in Experiment 5. Students performed better in their post- and

delayed- tests, in the group that did not use MAB blocks. The improved performance in the delayed test could also be an indication of better schema construction, encouraged by the use of simple worked examples without the distraction of the MAB blocks. In this situation, students were able to focus all their cognitive resources on the concept of place value.

11.4 Limitations of Study and Future Research Directions

Some of the limitations of this research may be utilised to inform future research directions exploring the use of manipulative materials in teaching place value, from the perspective of cognitive load theory. Firstly, only two manipulative materials (the abacus and MAB blocks) were used in the experiments. Other materials, such as cuisenaire rods, place value mats, grouped paddle-pop sticks and others can also be used in order to manipulate the concepts of place value in a concrete way. However, these other materials were not used as part of this study. The results would have been more generalized if a greater variety of manipulative materials could have been used to teach the concept of place value.

Secondly, due to the time constraints imposed by school timetables, only a limited number of worked examples was presented to the students in each of the experimental situations. Therefore, advantage could not be taken of the fact that more worked examples could have been used for students to acquire more general and robust schemas.

Thirdly, it is important to consider that there are variations in school curriculum milestones. Therefore, when students from different schools were used as part of the same experimental group, it could have been possible that some of their prior learning outcomes had affected their ability to solve questions and understand the concepts presented in this study.

Fourthly, this study did not measure the cognitive load placed on each student, as it is very difficult to measure cognitive load in children under 15 years of

age (H. Lee, 2014; H. Lee, Plass, & Homer, 2006). Because the students in this experimental study were all so young, with the oldest aged just 8 years, no attempt were made to measure the levels of cognitive load using, for example, subjective rating scales (Paas, Tuovinen, et al., 2003). Some alternative measures could possibly be used with this category of students. The ability to provide a valid and reliable estimation of the amount of load experienced whilst completing a task is a complex undertaking, even more so when it concerns such a young cohort of students (H. Lee, 2014). A small number of studies have used a modified subjective scale with cartoon faces to help young students rate their cognitive load whilst solving questions (Hu et al., 2015; van Loon-Hillen et al., 2012). However, no known studies have used such scales with children as young as 6-7, as used in this series of experiments. From converstations with students after the experiments have taken place, it appeared that the young students were keen to label any task undertaken as 'easy'. This was observed even when it was clear that the students found the questions and the content to be difficult. Thus, it could be possible that measuring cognitive load, with the use of a subjective scale, in such a young cohort of students could be not possible. This could present an interesting area for further research.

Finally, only private school children were tested in Experiments 2, 3, 4 and 5. It is not possible to say whether these results can be generalized across the category of schools: private, public, catholic and independent schools.

Despite these limitations, the results do appear to have a solid theoretical grounding in already well-established cognitive load theory and, therefore, may have significant implications for the current widespread use of MAB blocks. Further work in the area could be undertaken both to reinforce, and to expand on, these findings.

Some research has already shown that objects high in perceptual detail are not effective in establishing deep mathematical understanding of concepts (Goldstone & Sakamoto 2003; Sloutsky et al., 2005). Children often use all of their limited cognitive resources to focus on the details of the manipulative material, with no resources left to establish the connection between the manipulative material and the concept it is attempting to represent. However, MAB blocks do not present a rich perceptual experience, as the blocks are simple cubes devoid of shape or interesting structure. McNeil (2007) proposed that this aspect of MAB blocks could make it easier for students to view them as mathematical tools, and to focus on the concepts that the blocks are trying to teach. However, the findings of this research do not indicate this to be the case. It was often observed that children would use these blocks in task irrelevant ways, using them to build and play with, instead of understanding that they represented mathematical concepts. A possible further area of research could examine whether the different types of blocks could make a difference. For example, there are coloured blocks available, or blocks that are able to divide into units and click together to make longs, flats, and so on, thereby potentially encouraging a more fluent translation between the dual representations. Additionally, it would be interesting to investigate whether using the physical blocks only or being given visual representations of the blocks, or a combination of both, would lead to a change in student performance and understanding. This could further lead into examination of whether using blocks as part of animated simulation is beneficial to student learning. Because simulations are able to break up the cubes and, therefore, show that a long is made of 10 small unit blocks, there is potential that it would aid in a better dual representation of the concept and not cause a cognitive overload.

This study examined the use of blocks and an abacus as manipulative materials with students in their second and fourth year of schooling. It would be beneficial to investigate whether the expertise of the student impacts the way MAB blocks and other manipulative materials are used and whether they provide more value at the beginning of schooling or at later stages. For example, the blocks could be used at later stages of schooling to expand upon a concept once the concept has already been learnt. This is in contrast to using the blocks at the beginning of schooling to teach the concept initially.

Lastly, a longitudinal study would be of benefit to examine whether long-term cognitive benefits can be obtained by the use of manipulative materials, or whether the concept of dual representation and its effect on basic and initial schema construction would hinder the learning process. It would be interesting to start at the beginning of a child's formal education, and then track the frequency of the use of such material and its effects. A control class would run in parrallel and be given no access to manipulative materials. Such a study would have to be carried out in a real-life context, with no formal interventions. Each class would be provided with worked examples of place value questions at varying stages of the year.

11.5 Conclusion

The present thesis argues that the use of manipulative materials (such as MAB blocks or an abacus) in the early primary school curriculum does not support the creation of number composition schemas in the young learners, and may, indeed, hinder learning. This has implications for the design of instructional materials that incorporate both illustrations and the physical use of MAB blocks and other base-10 material, when teaching individual components of number deconstruction.

The results suggest that the use of MAB blocks in worked examples, both illustratively and physically, may cause an increased cognitive load on the student, and generate a redundancy effect. The findings of the present research may thus lead instructional designers and primary school teachers to create learning activities that do not incorporate the use of MAB blocks, whether as illustrations or physical, into teaching the concept of place value.

The study demonstrated that cognitive load factors are essential to be considered while designing or selecting instructional materials for this area of education. Established cognitive load theory techniques could be used to improve learning outcomes.

References

- Aburime, F. E. (2007). How manipulatives affect the mathematics achievement of students in Nigerian schools. *Educational Research Quarterly*. Retrieved from http://search.proquest.com/openview/e4d43451a395a0ea75fe50875347801e/ 1?pq-origsite=gscholar
- Adams, J. W., Hitch, G. J., & Donlan, C. (1998). Children's Mental Arithmetic and Working Memory. In *The development of mathematical skills. Studies in developmental psychology.* (pp. 153–173). Psychology Press/Taylor {&} Francis (UK), Hove, England,.
- Adler, I. (1966). Mental Growth and the Art of Teaching. *The Mathematics Teacher*, *59*(8), 706–715. Retrieved from

http://www.jstor.org.wwwproxy0.library.unsw.edu.au/stable/27957462

- Alloway, T. P., Gathercole, S. E., Willis, C., & Adams, A.-M. (2004). A structural analysis of working memory and related cognitive skills in young children. *Journal of Experimental Child Psychology*, *87*(2), 85–106. https://doi.org/10.1016/j.jecp.2003.10.002
- Anderson, J. R., & Fincham, J. M. (1994). Acquisition of procedural skills from examples. *Journal of Experimental Psychology: Learning, Memory, and Cognition*, *20*(6), 1322–1340. Retrieved from http://psycnet.apa.org/psycinfo/1995-04357-001
- Anderson, J. R., Fincham, J. M., & Douglass, S. (1997). The Role of Examples and Rules in the Acquisition of a Cognitive Skill. *Journal of Experimental Psychology: Learning, Memory, and Cognition*, *23*(4), 932–945. Retrieved from http://psycnet.apa.org/journals/xlm/23/4/932/
- Antell, S. E. G., & Caron, A. J. (1985). Neonatal Perception of Spatial
 Relationships. *Infant Behavior and Development*, 8(1), 15–23. Retrieved from http://www.sciencedirect.com/science/article/pii/S0163638385800138

Antell, S. E., & Keating, D. P. (1983). Perception of numerical invariance in

neonates. *Child Development*, 695–701. Retrieved from http://www.jstor.org/stable/10.2307/1130057

- Ashcraft, M. H. (1992). Cognitive arithmetic: A review of data and theory. *Cognition*, 44(1), 75–106. Retrieved from http://www.sciencedirect.com/science/article/pii/0010027792900511
- Ashcraft, M. H., & Krause, J. A. (2007). Working memory, math performance, and math anxiety. *Psychonomic Bulletin {&} Review*, *14*(2), 243–248. Retrieved from http://link.springer.com/article/10.3758/BF03194059
- Atkinson, R. C., & Shiffrin, R. M. (1968). Human memory: A proposed system and its control processes. *Psychology of Learning and Motivation*, *2*, 89–195.
 Retrieved from

http://books.google.com/books?hl=en&lr=&id=SVxyXuG73wwC&oi=fnd&pg=P A89&dq=Human+memory+A+proposed+system+and+its+control+processes& ots=CxO7uwQR7v&sig=fYb8f6AYeU0SO_wxvDPC5SP7QX4

- Atkinson, R. K., Derry, S. J., Renkl, A., & Wortham, D. (2000). Learning from examples: Instructional principles from the worked examples research. *Review* of Educational Research, 70(2), 181–214. Retrieved from http://rer.sagepub.com/content/70/2/181.short
- Ayres, P. (2006). Impact of reducing intrinsic cognitive load on learning in a mathematical domain. *Applied Cognitive Psychology*, *20*(3), 287–298. https://doi.org/10.1002/acp.1245
- Baddeley, A. D. (1998). Recent developments in working memory. *Current Opinions in Neurobiology*, *8*, 234–238.
- Baddeley, A. D. (1999). *Essentials of human memory*. Taylor {&} Francis Routledge. Retrieved from http://psycnet.apa.org/psycinfo/1999-02729-000
- Baddeley, A. D. (2000). The episodic buffer: a new component of working memory? *Trends in Cognitive Sciences*. Retrieved from http://www.sciencedirect.com/science/article/pii/S1364661300015382

Baddeley, A. D., & Hitch, G. J. (1974). Working memory. Psychology of Learning

and Motivation, 8, 47-89. Retrieved from

https://books.google.com.au/books/about/Psychology_of_Learning_and_Motiv ation.html?id=o5LScJ9ecGUC

Bailey, J. (2015). Supporting lower-achieving seven- and eight-year-old children with place value understandings. *Australian Primary Mathematics Classroom*, *20*(3), 3. Retrieved from http://search.informit.com.au/documentSummary;dn=492343129971150;res=l ELHSS

- Baillargeon, R. (1991). Reasoning about the height and location of a hidden object in 4.5-and 6.5-month-old infants. *Cognition*, *38*(1), 13–42. Retrieved from http://www.sciencedirect.com/science/article/pii/001002779190021U
- Baillargeon, R., & Graber, M. (1987). Where's the rabbit? 5.5-month-old infants' representation of the height of a hidden object. *Cognitive Development*, *2*(4), 375–392. Retrieved from

http://www.sciencedirect.com/science/article/pii/S088520148780014X

- Barbarin, O. A., & Wasik, B. H. (2009). Handbook of Child Development and Early Education: Research to Practice. (O. A. Barbarin & B. H. Wasik, Eds.).
 Guilford Press.
- Barner, D., Alvarez, G., Sullivan, J., & Frank, M. C. (2016). Learning Mathematics in a Visuospatial Format: A Randomized , Controlled Trial of Mental Abacus Instruction, 87(4), 1146–1158. https://doi.org/10.1111/cdev.12515
- Baroody, A. J. (1986). Counting Ability of Moderately and Mildly Handicapped Children. *Education & Training of the Mentally Retarded*, 289–300. Retrieved from http://psycnet.apa.org/psycinfo/1987-23427-001
- Baroody, A. J. (1990). How and when should place-value concepts and skills be taught? *Journal for Research in Mathematics Education*, *21*(4), 281–286.
 Retrieved from http://www.jstor.org/stable/10.2307/749526
- Barrouillet, P., Thevenot, C., & Fayol, M. (2010). Evidence for Knowledge of the Syntax of Large Numbers in Preschoolers. *Journal of Experimental Child*

Psychology, 105(3), 264–271. https://doi.org/10.1016/j.jecp.2009.10.005

- Bartolini Bussi, M. G. (2011). Artefacts and utilization schemes in mathematics teacher education: place value in early childhood education. *Journal of Mathematics Teacher Education*, 14(2), 93–112. https://doi.org/10.1007/s10857-011-9171-2
- Berends, I. E., & van Lieshout, E. C. D. M. (2009). The effect of illustrations in arithmetic problem-solving: Effects of increased cognitive load. *Learning and Instruction*, 19(4), 345–353. https://doi.org/10.1016/j.learninstruc.2008.06.012
- Berg, D. H. (2008). Working memory and arithmetic calculation in children: The contributory roles of processing speed, short-term memory, and reading. *Journal of Experimental Child Psychology*, *99*(4), 288–308. https://doi.org/10.1016/j.jecp.2007.12.002
- Bernardo, A. B. I. (2001). Asymmetric activation of number codes in bilinguals:
 Further evidence for the encoding complex model of number processing.
 Memory {&} Cognition, 29(7), 968–976. Retrieved from http://link.springer.com/article/10.3758/BF03195759
- Bjorklund, D. F., & Pellegrini, A. D. (2002). *The Origins of Human Nature: Evolutionary Developmental Psychology*. American Psychological Association.
- Blankenberger, S., & Vorberg, D. (1997). The single-format assumption in arithmetic fact retrieval. *Journal of Experimental Psychology: Learning, Memory, and Cognition, 23*(3), 721. Retrieved from http://psycnet.apa.org/journals/xlm/23/3/721/
- Board of Studies Teaching and Educational Standards NSW. (n.d.). Syllabus Mathematics K-10. Retrieved from https://syllabus.bostes.nsw.edu.au/mathematics/mathematics
 - k10/content/1105/
- Bobis, J., Sweller, J., & Cooper, M. (1993). Cognitive load effects in a primaryschool geometry task. *Learning and Instruction*, *3*(1), 1–21. https://doi.org/10.1016/S0959-4752(09)80002-9

- Bokosmaty, S., Sweller, J., & Kalyuga, S. (2014). Learning Geometry Problem Solving by Studying Worked Examples: Effects of Learner Guidance and Expertise. *American Educational Research Journal*, 1–27. https://doi.org/10.3102/0002831214549450
- Bomba, P. C., & Siqueland, E. R. (1983). The nature and structure of infant form categories. *Journal of Experimental Child Psychology*, *35*(2), 294–328.
 Retrieved from

http://www.sciencedirect.com/science/article/pii/0022096583900851

- Boulton-Lewis, G. M. (1998). Children's strategy use and interpretations of mathematical representations. *The Journal of Mathematical Behavior*, *17*(2), 219–237. https://doi.org/10.1016/S0364-0213(99)80060-3
- Brannon, E. M. (2002). The development of ordinal numerical knowledge in infancy. *Cognition*, *83*(3), 223–240. Retrieved from http://www.sciencedirect.com/science/article/pii/S0010027702000057
- Brünken, R., Plass, J. L., & Leutner, D. (2004). Assessment of Cognitive Load in Multimedia Learning with Dual-Task Methodology: Auditory Load and Modality Effects. *Instructional Science*, *32*(1–2), 115–132. Retrieved from http://link.springer.com/article/10.1023/B:TRUC.0000021812.96911.c5
- Bull, R., & Scerif, G. (2001). Executive Functioning as a Predictor of Children's Mathematics Ability: Inhibition, Switching, and Working Memory. *Developmental Neuropsychology*, *19*(3), 273–293. https://doi.org/10.1207/S15326942DN1903_3
- Byrge, L., Smith, L. B., & Mix, K. S. (2013). Beginnings of Place Value: How Preschoolers Write Three-Digit Numbers. *Child Development*, *85*(2), 437–443. https://doi.org/10.1111/cdev.12162
- Campbell, J. I. D. (1994). Architectures for numerical cognition. *Cognition*, *53*(1), 1–44. Retrieved from

http://www.sciencedirect.com/science/article/pii/0010027794900752 Campbell, J. I. D., & Epp, L. J. (2004). An encoding-complex approach to numerical cognition in Chinese-English bilinguals. *Canadian Journal of Experimental Psychology/Revue Canadienne de Psychologie Exp{é}rimentale*, *58*(4), 229. Retrieved from http://psycnet.apa.org/journals/cep/58/4/229/

- Carbonneau, K. J., & Marley, S. C. (2015). Instructional guidance and realism of manipulatives influence preschool children's mathematics learning. *The Journal of Experimental Education*, *83*(4), 495–513. https://doi.org/10.1080/00220973.2014.989306
- Carbonneau, K. J., Marley, S. C., & Selig, J. P. (2013). A meta-analysis of the efficacy of teaching mathematics with concrete manipulatives. *Journal of Educational Psychology*, *105*(2), 380–400. https://doi.org/10.1037/a0031084
- Carlson, R., Chandler, P., & Sweller, J. (2003). Learning and understanding science instructional material. *Journal of Educational Psychology*, *95*(3), 629–640. https://doi.org/10.1037/0022-0663.95.3.629
- Chan, B. M., & Ho, C. S.-H. (2010). The cognitive profile of Chinese children with mathematics difficulties. *Journal of Experimental Child Psychology*, *107*(3), 260–279. https://doi.org/10.1016/j.jecp.2010.04.016
- Chan, W. W. L., Au, T. K., & Tang, J. (2014). Strategic counting: A novel assessment of place-value understanding. *Learning and Instruction*, *29*, 78–94. https://doi.org/10.1016/j.learninstruc.2013.09.001
- Chandler, C. C., & Kamii, C. (2009). Giving Change When Payment Is Made with a Dime: the Difficulty of Tens and Ones. *Journal for Research in Mathematics Education*, *40*(2), 97–118. Retrieved from http://www.jstor.org/stable/40539328
- Chandler, P., & Sweller, J. (1991). Cognitive Load Theory and the Format of Instruction. *Cognition and Instruction*, 8(4), 293–332. https://doi.org/10.1207/s1532690xci0804_2
- Chandler, P., & Sweller, J. (1992). The Split-Attention Effect as a Factor in the Design of Instruction. *British Journal of Educational Psychology*, *62*(2), 233–246. Retrieved from http://onlinelibrary.wiley.com/doi/10.1111/j.2044-

8279.1992.tb01017.x/full

- Chandler, P., & Sweller, J. (1996). Cognitive Load While Learning to Use a Computer Program. *Applied Cognitive Psychology*, *10*(2), 151–170. Retrieved from http://www.csd.uoc.gr/~hy564/files/material/17/Cognitive Load While Learning to Use Computer.pdf
- Chen, O., Kalyuga, S., & Sweller, J. (2015). The Worked Example Effect, the Generation Effect, and Element Interactivity. *Journal of Educational Psychology*, 1–17. https://doi.org/10.1037/edu0000018
- Chi, M. T. H., Feltovich, P. J., & Glaser, R. (1981). Categorization and Representation of Physics Problems by Experts and Novices. *Cognitive Psychology*, *5*, 121–152. Retrieved from http://www.sciencedirect.com/science/article/pii/S0364021381800298
- Chi, M. T. H., Glaser, R., & Rees, E. (1982). Expertise in Problem Solving. In R. J.
 Sternberg (Ed.), *Advances in the psychology of human intelligence* (pp. 1–35).
 Hillsdale, N.J.: Erlbaum.
- Clark, R. C., & Mayer, R. E. (2011). *e-Learning and the Science of Instruction*. John Wiley {&} Sons. Retrieved from http://books.google.com.au/books?id=MOutGGET2VwC&printsec=frontcover& dq=E+Learning+and+the+science+of+instruction&hl=&cd=1&source=gbs_api
- Clarke, T., Ayres, P., & Sweller, J. (2005). The Impact of Sequencing and Prior Knowledge on Learning Mathematics Through Spreadsheet Applications. *Educational Technology Research and Development*, *53*(3), 15–24. Retrieved from http://link.springer.com/article/10.1007/BF02504794
- Cobb, P. (1995). Cultural tools and mathematical learning: A case study. *Journal for Research in Mathematics Education*. Retrieved from http://psycnet.apa.org/psycinfo/1996-09556-001
- Cocchini, G., Logie, R. H., Sala, S. Della, MacPherson, S. E., & Baddeley, A. D. (2002). Concurrent performance of two memory tasks: Evidence for domainspecific working memory systems. *Memory {&} Cognition, 30*(7), 1086–1095.

https://doi.org/10.3758/BF03194326

- Cohen, J. (1988). *Statistical power analysis for the behavioral sciences*. Hillsdale, N.J.: L. Erlbaum Associates.
- Cohen, N. J. (1980). Preserved Learning and Retention of Pattern-Analyzing Skill in Amnesia: Dissociation of Knowing How and Knowing That. *Science*, *210*, 207–209.
- Cooper, G. A., & Sweller, J. (1987). Effects of Schema Acquisition and Rule Automation on Mathematical Problem-Solving Transfer. *Journal of Educational Psychology*, *79*(4), 347–362. Retrieved from http://psycnet.apa.org/journals/edu/79/4/347/
- Cooper, R. G. (1984). Early Number Development: Discovering Number Space with Addition and Subtraction . In C. Sophian (Ed.), *Origins of cognitive skills* (pp. 157–192). Hillsdale, N.J.: Erlbaum.
- Cornoldi, C., & Vecchi, T. (2004). *Visuo-spatial Working Memory and Individual Differences*. Psychology Press. Retrieved from http://books.google.com.au/books?id=rcp4AgAAQBAJ&printsec=frontcover&d q=Visuo+spatial+Working+Memory+and+Individual+Differences&hl=&cd=1&s ource=gbs_api
- Cowan, N. (2001). Metatheory of storage capacity limits. *Behavioral and Brain Sciences*, *24*(1), 154–176.
- Daraganova, G., & Ainley, J. (2012). The Longitudinal Study of Australian Children Annual Statistical Report 2011. In A. I. of Family Studies (Ed.), *Chapter 8: Children's numeracy skills* (pp. 1–114).
- De Groot, A. (1965). Thought and choice in chess. *The Hague Netherlands: Mouton*. Retrieved from http://scholar.google.com/scholar?q=related:d1buRDhlivAJ:scholar.google.co m/&hl=en&num=20&as_sdt=0,5
- Dehaene, S. (1992). Varieties of numerical abilities. *Cognition*, 44(1), 1–42. Retrieved from

http://www.sciencedirect.com/science/article/pii/001002779290049N

- Dehaene, S. (1997). *The Number Sense How the Mind Creates Mathematics*. Oxford University Press.
- Dehaene, S. (2004). Arithmetic and the brain. *Current Opinion in Neurobiology*, *14*(2), 218–224. https://doi.org/10.1016/j.conb.2004.03.008
- Dehaene, S., Izard, V., Pica, P., & Spelke, E. (2006). Core Knowledge of Geometry in an Amazonian Indigene Group. *Science*, *311*(5759), 381–384. https://doi.org/10.1126/science.1121739
- Dehaene, S., Spelke, E., Pinel, P., Stanescu, R., & Tsivkin, S. (1999). Sources of Mathematical Thinking: Behavioral and Brain-Imaging Evidence. *Science*, *284*(5416), 970–974. Retrieved from http://www.sciencemag.org/content/284/5416/970.short
- Dienes, Z. P. (1961). *Building Up Mathematics*. London Hutchinson Educational label: Humanities Press, New York.
- Dietrich, J. F., Huber, S., Dackermann, T., Moeller, K., & Fischer, U. (2016). Place-value understanding in number line estimation predicts future arithmetic performance. *British Journal of Developmental Psychology*, *34*(4), 502–517. https://doi.org/10.1111/bjdp.12146
- Driscoll, M. P. (2005). *Psychology of learning for instruction* (3rd ed.). Boston: Pearson. Retrieved from http://ocw.metu.edu.tr/pluginfile.php/9013/mod_resource/content/1/driscollch10 (1).pdf
- English, L. D., & Halford, G. S. (1995). *Mathematics Education: Models and Processes*. Mahwah, NJ: Lawrence Erlbaum Associates, Publishers.
- Ericsson, K. A., Chase, W. G., & Faloon, S. (1980). Acquisition of a Memory Skill. Science, 208(6), 1181–1182. Retrieved from http://www.cogpsy.info/images/5geheugen/Ericsson et al - 1980 - acquisition of a Memory Skill.pdf
- Ericsson, K. A., Delaney, P. F., Weaver, G., & Mahadevan, R. (2004). Uncovering the structure of a memorist's superior "basic" memory capacity. *Cognitive*

Psychology, 49(3), 191-237. https://doi.org/10.1016/j.cogpsych.2004.02.001

- Ericsson, K. A., & Kintsch, W. (1995). Long-term working memory. *Psychological Review*, *102*(2), 211–245. Retrieved from http://psycnet.apa.org/psycinfo/1995-24067-001
- Ericsson, K. A., Krampe, R. T., & Tesch-Römer, C. (1993). The role of deliberate practice in the acquisition of expert performance. *Psychological Review*, *100*(3), 363–406. https://doi.org/10.1037/0033-295X.100.3.363
- Feigenson, L., Carey, S., & Hauser, M. (2002). The representations underlying infants' choice of more: Object files versus analog magnitudes. *Psychological Science*, *13*(2), 150–156. Retrieved from http://pss.sagepub.com/content/13/2/150.short
- Fennema, E. (1972). Models and mathematics. *The Arithmetic Teacher*, *19*(8), 635–640.
- Field, A. (2007). *Discovering Statistics Using SPSS*. SAGE Publications. Retrieved from http://ebookcentral.proquest.com/lib/unsw/detail.action?docID=743649
- Florax, M., & Ploetzner, R. (2010). What contributes to the split-attention effect? The role of text segmentation, picture labelling, and spatial proximity. *Learning and Instruction*, *20*(3), 216–224.

https://doi.org/10.1016/j.learninstruc.2009.02.021

- Fosnot, C. T., & Dolk, M. (2001). *Young Mathematicians at Work*. Heinemann Educational Books.
- Friedman, M. (1978). The manipulative materials strategy: The latest pied piper? Journal for Research in Mathematics Education. Retrieved from http://www.jstor.org/stable/748965

Frith, C. D., & Frit, U. (1972). The solitaire illusion: An illusion of numerosity. *Perception {&} Psychophysics*. Retrieved from http://link.springer.com/article/10.3758/BF03206279

Fürst, A. J., & Hitch, G. J. (2000). Separate roles for executive and phonological components of working memory in mental arithmetic. *Memory {&} Cognition,*

28(5), 774-782. Retrieved from

http://link.springer.com/article/10.3758/BF03198412

- Fuson, K. C. (1990). Conceptual structures for multiunit numbers: Implications for learning and teaching multidigit addition, subtraction, and place value.
 Cognition and Instruction, 7(4), 343–403. Retrieved from http://www.tandfonline.com/doi/pdf/10.1207/s1532690xci0704_4
- Fuson, K. C. (1992). Handbook of Research on Mathematics Teaching and Learning. In D. A. Grouws (Ed.), *A project of the National Council of Teachers* of Mathematics (pp. 243–275). England: Macmillan Publishing Co. Inc.
- Fuson, K. C., & Briars, D. J. (1990). Using a Base-Ten Blocks Learning/Teaching Approach for First-and Second-Grade Place-Value and Multidigit Addition and Subtraction. *Journal for Research in Mathematics Education*, *21*(3), 180–206. Retrieved from http://www.jstor.org/stable/749373
- Fuson, K. C., Wearne, D., Hiebert, J. C., Murray, H. G., Human, P. G., Olivier, A. I.,
 ... Fennema, E. (1997). Children's conceptual structures for mulitdigit numbers and methods of multidigit addition and subtraction. *Journal for Research in Mathematics Education*, *28*(2), 130–162. Retrieved from http://psycnet.apa.org/psycinfo/1997-03871-001
- García-Orza, J., León-Carrión, J., & Vega, O. (2003). Dissociating Arabic Numeral Reading and Basic Calculation: A Case Study. *Neurocase*, *9*(2), 1–11. Retrieved from

http://www.tandfonline.com/doi/abs/10.1076/neur.9.2.129.15070

 Geary, D. C. (1990). A componential analysis of an early learning deficit in mathematics. *Journal of Experimental Child Psychology*, *49*(3), 363–383.
 Retrieved from

http://www.sciencedirect.com/science/article/pii/002209659090065G

Geary, D. C. (1994a). Children's Early Numerical Abilities. In *Children's mathematical development: Research and practical applications* (pp. 1–35).
 Washington DC, US: American Psychological Association.

https://doi.org/10.1037/10163-001

- Geary, D. C. (1994b). Children's Early Numerical Abilities. In *Children's mathematical development: Research and practical applications* (pp. 1–35).
 Washington DC, US: American Psychological Association. https://doi.org/10.1037/10163-001
- Geary, D. C. (1994c). *Children's mathematical development: Research and practical applications*. Washington DC, US: American Psychological Association.
- Geary, D. C. (1994d). Developing arithmetical skills. In *Children's Mathematical Development: Research and Practical Applications* (p. 327). APA Books.
 Retrieved from http://books.google.com.au/books?id=K5cucAAACAAJ&dq=Childrens+Mathe matical+Development&hl=&cd=1&source=gbs_api
- Geary, D. C. (1994e). Mathematical Disabilities. In *Children's mathematical development: Research and practical applications* (pp. 155–187). Washington DC, US: American Psychological Association. https://doi.org/10.1037/10163-001
- Geary, D. C. (1995). Reflections of evolution and culture in children's cognition: Implications for mathematical development and instruction. *American Psychologist*, *50*(1), 24–37.
- Geary, D. C. (1996). Biology, Culture, and Cross-National Differences in Mathematical Ability. In R. J. Sternberg & T. Ben-Zeev (Eds.), *The Nature of Mathematical Thinking*. Mahwah, NJ: Lawrence Erlbaum Associates, Publishers.
- Geary, D. C. (2000). From infancy to adulthood: the development of numerical abilities. *European Child Adolescent Psychiatry*, *9 Suppl 2*(22), II11-I16. Retrieved from http://www.ncbi.nlm.nih.gov/pubmed/11138899
- Geary, D. C. (2002). Principles of evolutionary educational psychology. *Learning and Individual Differences*, *12*(4), 317–345. Retrieved from

http://www.sciencedirect.com/science/article/pii/S1041608002000468

- Geary, D. C. (2005). Evolution and Development of Brain and Cognition. In *The* Origin of Mind: Evolution Of Brain, Cognition, And General Intelligence (p. 459). American Psychological Association (APA). Retrieved from http://books.google.com.au/books?id=a7ixQgAACAAJ&dq=The+origin+of+min d&hl=&cd=6&source=gbs_api
- Geary, D. C. (2007). Educating the Evolved Mind: Reflections and Refinements. In
 J. S. Carlson & J. R. Levin (Eds.), *Educating the Evolved Mind: Conceptual foundations for an evolutionary educational psychology*. Information Age Publishing, Inc.
- Geary, D. C. (2008a). An Evolutionarily Informed Education Science. *Educational Psychologist*, *43*(4), 179–195. https://doi.org/10.1080/00461520802392133
- Geary, D. C. (2008b). Whither Evolutionary Educational Psychology? *Educational Psychologist*, *43*(4), 217–226. https://doi.org/10.1080/00461520802392240
- Geary, D. C., & Bjorklund, D. F. (2000). Evolutionary Developmental Psychology. *Child Development*, 71(1), 57–65. Retrieved from http://www.jstor.org/stable/1132217
- Geary, D. C., Hoard, M. K., Byrd-Craven, J., & Catherine DeSoto, M. (2004).
 Strategy choices in simple and complex addition: Contributions of working memory and counting knowledge for children with mathematical disability. *Journal of Experimental Child Psychology*, *88*(2), 121–151.
 https://doi.org/10.1016/j.jecp.2004.03.002
- Geary, D. C., Hoard, M. K., Nugent, L., & Bailey, D. H. (2013). Adolescents' Functional Numeracy Is Predicted by Their School Entry Number System Knowledge. *PLoS ONE*, *8*(1), 1–8. https://doi.org/10.1371/journal.pone.0054651
- Gerjets, P., & Scheiter, K. (2003). Goal Configurations and Processing Strategies as Moderators Between Instructional Design and Cognitive Load: Evidence From Hypertext-Based Instruction. *Educational Psychologist*, *38*(1), 33–41.

https://doi.org/10.1207/S15326985EP3801_5

- Gerjets, P., Scheiter, K., & Catrambone, R. (2006). Can learning from molar and modular worked examples be enhanced by providing instructional explanations and prompting self-explanations? *Learning and Instruction*, *16*(2), 104–121. https://doi.org/10.1016/j.learninstruc.2006.02.007
- Ginns, P. (2005). Meta-analysis of the modality effect. *Learning and Instruction*, *15*(4), 313–331. https://doi.org/10.1016/j.learninstruc.2005.07.001
- Ginsburg, N. (1976). Effect of Item Arrangement on Perceived Numerosity:
 Randomness vs Regularity. *Perceptual and Motor Skills*, *43*(2), 663–668.
 Retrieved from http://www.amsciepub.com/doi/pdf/10.2466/pms.1976.43.2.663
- Gobet, F. (2000). Some shortcomings of long-term working memory. *British Journal of Psychology*, *91*, 551–570.
- Gobet, F., & Simon, H. A. (1996). Templates in Chess Memory: a Mechanism for Recalling Several Boards. *Cognitive Psychology*, *31*(1), 1–40. Retrieved from http://bura.brunel.ac.uk/handle/2438/1339
- Greenberg, D. L., & Verfaellie, M. (2010). Interdependence of episodic and semantic memory: Evidence from neuropsychology. *Journal of the International Neuropsychological Society*, *16*(5), 748–753. https://doi.org/10.1017/S1355617710000676
- Guida, A., Gobet, F., Tardieu, H., & Nicolas, S. (2012). Brain and Cognition. *Brain and Cognition*, *79*(3), 221–244. https://doi.org/10.1016/j.bandc.2012.01.010
- Harskamp, E. G., Mayer, R. E., & Suhre, C. (2007). Does the modality principle for multimedia learning apply to science classrooms? *Learning and Instruction*, *17*(5), 465–477. https://doi.org/10.1016/j.learninstruc.2007.09.010
- Hecht, S. A. (2002). Counting on working memory in simple arithmetic when counting is used for problem solving. *Memory {&} Cognition*, *30*(3), 447–455.
 Retrieved from http://link.springer.com/article/10.3758/BF03194945
- Hiebert, J., & Wearne, D. (1992). Links between teaching and learning place value with understanding in first grade. *Journal for Research in Mathematics*
Education, 98–122. Retrieved from http://www.jstor.org/stable/10.2307/749496

- Ho, C. S.-H., & Cheng, S.-F. (1997). Training in Place-Value Concepts Improves Children's Addition Skills. *Contemporary Educational Psychology*, *22*, 495– 506.
- Hoard, M. K., Geary, D. C., Byrd-Craven, J., & Nugent, L. (2008). Mathematical Cognition in Intellectually Precocious First Graders. *Developmental Neuropsychology*, *33*(3), 251–276. https://doi.org/10.1080/87565640801982338
- Holmes, J., & Adams, J. W. (2006). Working Memory and Children's Mathematical Skills: Implications for mathematical development and mathematics curricula. *Educational Psychology*, *26*(3), 339–366. https://doi.org/10.1080/01443410500341056
- Houdé, O. (1997). Numerical Development: From the Infant to the Child. Wynn's (1992) Paradigm in 2-and 3-Year Olds. *Cognitive Development*, *12*(3), 373–391. Retrieved from
- http://www.sciencedirect.com/science/article/pii/S0885201497900095 Hu, F.-T. (2014, November). *Getting the Point: Tracing Worked Examples*

Enhances Learning.

- Hu, F.-T., Ginns, P., & Bobis, J. (2015). Getting the point: tracing worked examples enhances learning. *Learning and Instruction*, *35*, 85–93.
- Huff, M., Bauhoff, V., & Schwan, S. (2012). Computers in Human Behavior. *Computers in Human Behavior*, *28*(4), 1254–1261.
 https://doi.org/10.1016/j.chb.2012.02.008
- Huff, M., & Schwan, S. (2010). Integrating information from two pictorial animations: Complexity and cognitive prerequisites influence performance. *Applied Cognitive Psychology*, *25*(6), 878–886. https://doi.org/10.1002/acp.1762
- Huttenlocher, J., Newcombe, N., & Sandberg, E. H. (1994). The coding of spatial location in young children. *Cognitive Psychology*, *27*(2), 115–147. Retrieved

from http://www.sciencedirect.com/science/article/pii/S0010028584710140

Hynes, M. C. (1986). Selection Criteria. *The Arithmetic Teacher*, *33*(6), 11–13. Retrieved from

http://www.jstor.org.wwwproxy0.library.unsw.edu.au/stable/41192834

Izard, V., Sann, C., Spelke, E. S., & Streri, A. (2009). Newborn Infants Perceive Abstract Numbers. *Proceedings of the National Academy of Sciences*, *106*(25), 10382–10385. Retrieved from http://www.pnas.org/content/106/25/10382.short

- Izard, V., & Spelke, E. S. (2009). Development of Sensitivity to Geometry in Visual Forms. *Human Evolution*, 23(3), 213. Retrieved from http://www.ncbi.nlm.nih.gov/pmc/articles/PMC3045057/
- Jeung, H. J., Chandler, P., & Sweller, J. (1997). The Role of Visual Indicators in Dual Sensory Mode Instruction. *Educational Psychology*, *17*(3), 329–345. https://doi.org/10.1080/0144341970170307

Jordan, N. C., Kaplan, D., Ramineni, C., & Locuniak, M. N. (2009). Early math matters: kindergarten number competence and later mathematics outcomes. *Developmental Psychology*, *45*(3), 850–67. https://doi.org/10.1037/a0014939

Kalyuga, S. (2010). Schema Acquisition and Sources of Cognitive Load. In J. L.
Plass, R. Moreno, & R. Brunken (Eds.), *Cognitive load theory* (pp. 48–64).
New York: Cambridge. Retrieved from
http://books.google.com/books?hl=en&lr=&id=mFJe8ZnAb3EC&oi=fnd&pg=P
R5&dq=Cognitive+mechanisms+of+schema+acquisition+and+transfer+from+c
onsciously+controlled+to+automatic+processing+are+the+major+learning+me
chanisms+and+foundation+of+our+intellectual+abili

- Kalyuga, S. (2011). Cognitive Load Theory: How Many Types of Load Does It Really Need? *Educational Psychology Review*, *23*(1), 1–19. https://doi.org/10.1007/s10648-010-9150-7
- Kalyuga, S., Ayres, P., Chandler, P., & Sweller, J. (2003). The Expertise Reversal Effect. *Educational Psychologist*, *38*(1), 23–31.

https://doi.org/10.1207/S15326985EP3801_4

- Kalyuga, S., Chandler, P., & Sweller, J. (1998). Levels of Expertise and Instructional Design. *Human Factors: The Journal of the Human Factors and Ergonomics Society*, 40(1), 1–17. Retrieved from http://hfs.sagepub.com/content/40/1/1.short
- Kalyuga, S., Chandler, P., & Sweller, J. (1999). Managing Split-Attention and Redundancy in Multimedia Instruction. *Applied Cognitive Psychology*, *13*(4), 351–371. Retrieved from http://visuallearningresearch.wiki.educ.msu.edu/file/view/Kalyuga,+Chandler,+ %26+Sweller+(1999).pdf/47290509/Kalyuga,+Chandler,+%26+Sweller+(1999)
 .pdf
- Kalyuga, S., Chandler, P., & Sweller, J. (2000). Incorporating learner experience into the design of multimedia instruction. *Journal of Educational Psychology*, *92*(1), 126–136. https://doi.org/10.1037//0022-0663.92.1.126
- Kalyuga, S., Chandler, P., Tuovinen, J., & Sweller, J. (2001). When problem solving is superior to studying worked examples. *Journal of Educational Psychology*, *93*(3), 579–588. https://doi.org/10.1037//0022-0663.93.3.579
- Kalyuga, S., & Sweller, J. (2004). Measuring Knowledge to Optimize Cognitive Load Factors During Instruction. *Journal of Educational Psychology*, *96*(3), 558–568. https://doi.org/10.1037/0022-0663.96.3.558
- Kamii, C. (1986). Place Value: An Explanation of Its Difficulty and Educational Implications for the Primary Grades. *Journal of Research in Childhood Education*, 1(2), 75–86. https://doi.org/10.1080/02568548609594909
- Kamii, C. (2000). Young Children Reinvent Arithmetic. Teachers College Press. Retrieved from http://books.google.com.au/books?id=k9aWSwAACAAJ&dq=Young+children+

reinvent+arithmetic&hl=&cd=1&source=gbs_api

Kamii, C., & Joseph, L. L. (2004). *Young Children Continue to Reinvent Arithmetic* - *2nd Grade*. Teachers College Press. Retrieved from http://books.google.com.au/books?id=WYsK1zyHqYAC&printsec=frontcover& dq=Young+children+reinvent+arithmetic&hl=&cd=2&source=gbs_api

- Kaminski, J. A., Sloutsky, V. M., & Heckler, A. F. (2008). LEARNING THEORY: The Advantage of Abstract Examples in Learning Math. *Science*, *320*(5875), 454–455. https://doi.org/10.1126/science.1154659
- Keil, F. C. (2008). Adapted Minds and Evolved Schools. *Educational Psychologist*, *43*(4), 196–202. https://doi.org/10.1080/00461520802392232
- Kellman, P. J., & Spelke, E. S. (1983). Perception of partly occluded objects in infancy. *Cognitive Psychology*, *15*(4), 483–524. Retrieved from http://www.sciencedirect.com/science/article/pii/0010028583900178
- Kennedy, L., Tipps, S., & Johnson, A. (2007). Guiding Children's Learning of Mathematics. Cengage Learning. Retrieved from http://books.google.com.au/books?id=mrFYTHOjHjYC&printsec=frontcover&d q=Guiding+Childrens+learning+of+mathematics&hl=&cd=1&source=gbs_api
- Kintsch, W. (1998). *Comprehension*. Cambridge University Press. Retrieved from http://books.google.com.au/books?id=LuycnLrY3k8C&printsec=frontcover&dq =Comprehension+A+paradigm+for+cognition&hl=&cd=1&source=gbs_api
- Kirschner, P., Sweller, J., & Clark, R. E. (2006). Why minimal guidance during instruction does not work: An analysis of the failure of constructivist, discovery, problem-based, experiential and inquiry-based teaching. *Educational Psychologist*, *41*, 75–86. Retrieved from http://www.ydae.purdue.edu/LCT/HBCU/documents/Analysis_of_the_Failure_of_Discovery_PBL__Experiential_Inquiry_Learning.pdf
- Kobayashi, T., Hiraki, K., Mugitani, R., & Hasegawa, T. (2004). Baby arithmetic: one object plus one tone. *Cognition*, *91*(2), B23--B34. https://doi.org/10.1016/j.cognition.2003.09.004
- Koechlin, E. (1997). Numerical Transformations in Five-Month-Old Human Infants.
 Mathematical Cognition, 3, 89–104. Retrieved from http://www.tandfonline.com/doi/abs/10.1080/135467997387425

- Kroesbergen, E. H., Van Luit, J. E. H., Van Lieshout, E. C. D. M., Van Loosbroek,
 E., & de Rijt, B. A. M. (2009). Individual Differences in Early Numeracy: The
 Role of Executive Functions and Subitizing. *Journal of Psychoeducational Assessment*, *27*(3), 226–236. https://doi.org/10.1177/0734282908330586
- Langley, P., Laird, J. E., & Rogers, S. (2009). Cognitive Architectures: Research Issues and Challenges. *Cognitive Systems Research*, *10*(2), 141–160. https://doi.org/10.1016/j.cogsys.2006.07.004
- Laski, E. V, Ermakova, A., & Vasilyeva, M. (2014). Early use of decomposition for addition and its relation to base-10 knowledge. *Journal of Applied Developmental Psychology*, *35*(5), 444–454.
 https://doi.org/10.1016/j.appdev.2014.07.002
- Leahy, W., & Sweller, J. (2011). Cognitive load theory, modality of presentation and the transient information effect. *Applied Cognitive Psychology*, *25*(6), 943– 951. https://doi.org/10.1002/acp.1787
- Lee, H. (2014). Measuring cognitive load with electroencephalography and selfreport: focus on the effect of English-medium learning for Korean students. *Educational Psychology*, 1–11. https://doi.org/10.1080/01443410.2013.860217
- Lee, H., Plass, J. L., & Homer, B. D. (2006). Optimizing cognitive load for learning from computer-based science simulations. *Journal of Educational Psychology*, *98*(4), 902–913. https://doi.org/10.1037/0022-0663.98.4.902
- Lee, K., Ng, E. L., & Ng, S. F. (2009). The contributions of working memory and executive functioning to problem representation and solution generation in algebraic word problems. *Journal of Educational Psychology*, *101*(2), 373– 387. https://doi.org/10.1037/a0013843
- LeFevre, J., DeStefano, D., Coleman, B., & Shanahan, T. (2005). Mathematical Cognition and Working Memory. In J. I. D. Campbell (Ed.), *Handbook of mathematical cognition* (pp. 361–378). New York: Psychology Press. Retrieved from

http://books.google.com/books?hl=en&lr=&id=AQZ5jmmpaDAC&oi=fnd&pg=P

A361&dq=Mathematical+cognition+and+working+memory&ots=sNtu8s8_He& sig=lmekaN1hKI_e6F-U98mz_Vjw4e4

- Leppink, J., Paas, F., Gog, T. Van, Vleuten, C. P. M. Van Der, & Merriënboer, J. J.
 G. Van. (2014). Effects of pairs of problems and examples on task performance and different types of cognitive load. *Learning and Instruction*, 30, 32–42. https://doi.org/10.1016/j.learninstruc.2013.12.001
- Leslie, K. C., Low, R., Jin, P., & Sweller, J. (2011). Redundancy and expertise reversal effects when using educational technology to learn primary school science. *Educational Technology Research and Development*, *60*(1), 1–13. https://doi.org/10.1007/s11423-011-9199-0
- Lipton, J. S., & Spelke, E. S. (2004). Discrimination of Large and Small Numerosities by Human Infants. *Infancy*, *5*(3), 271–290. Retrieved from http://www.tandfonline.com/doi/abs/10.1207/s15327078in0503_2
- Logie, R. H., Gilhooly, K. J., & Wynn, V. (1994). Counting on Working Memory in Arithmetic Problem Solving. *Memory* {&} *Cognition*, *22*(4), 395–410. Retrieved from http://link.springer.com/article/10.3758/BF03200866

Lourenco, S. F., & Huttenlocher, J. (2008). The Representation of Geometric Cues in Infancy. *Infancy*, *13*(2), 103–127. https://doi.org/10.1080/15250000701795572

- Mandler, G., & Shebo, B. J. (1982). Subitizing: an analysis of its component processes. *Journal of Experimental Psychology: General*, *111*(1), 1. Retrieved from http://psycnet.apa.org/journals/xge/111/1/1/
- Mann, A., Moeller, K., Pixner, S., Kaufmann, L., & Nuerk, H.-C. (2011). Attentional Strategies in Place-Value Integration. *Zeitschrift F{ü}r Psychologie*, *219*(1), 42–49. https://doi.org/10.1027/2151-2604/a000045
- Mann, A., Moeller, K., Pixner, S., Kaufmann, L., & Nuerk, H.-C. (2012). On the development of Arabic three-digit number processing in primary school children. *Journal of Experimental Child Psychology*, *113*(4), 594–601. https://doi.org/10.1016/j.jecp.2012.08.002

- Marley, S. C., & Carbonneau, K. J. (2014). Future Directions for Theory and Research with Instructional Manipulatives:Commentary on the Special Issue Papers. *Educational Psychology Review*, *26*(1), 91–100. https://doi.org/10.1007/s10648-014-9259-1
- Marshall, L., & Swan, P. (2013). Exploring the Use of Mathematics Manipulative Materials: Is It What We Think It Is? In *EDU-COM International Conference Sustainability in Higher Eduation Directions for Change* (pp. 1–14). Perth, Western Australia.

Mayer, R. E. (2005). Principles for Managing Essential Processing in Multimedia Learning: Segmenting, Pretraining, and Modality. In R. E. Mayer (Ed.), *The Cambridge Handbook of Multimedia Learning* (pp. 169–182). Cambridge University Press. Retrieved from http://books.google.com.au/books?id=SSLdo1MLlywC&pg=PA169&dq=Princi ples+for+managing+essential+processing+in+multimedia+learning+segmentin g+pretraining+and+modality&hl=&cd=1&source=gbs_api

- Mayer, R. E., Bove, W., Bryman, A., Mars, R., & Tapangco, L. (1996). When Less Is More: Meaningful Learning From Visual and Verbal Summaries of Science Textbook Lessons. *Journal of Educational Psychology*, *88*(1), 64. Retrieved from http://psycnet.apa.org/journals/edu/88/1/64/
- Mayer, R. E., Mathias, A., & Wetzell, K. (2002). Fostering understanding of multimedia messages through pre-training: Evidence for a two-stage theory of mental model construction. *Journal of Experimental Psychology: Applied*, 8(3), 147–154. https://doi.org/10.1037//1076-898X.8.3.147
- Mayer, R. E., & Moreno, R. (1998). A Split-Attention Effect in Multimedia Learning: Evidence for Dual Processing Systems in Working Memory. *Journal of Educational Psychology*, *90*(2), 312. Retrieved from http://psycnet.apa.org/journals/edu/90/2/312/
- Mayer, R. E., & Moreno, R. (2003). Nine Ways to Reduce Cognitive Load in Multimedia Learning. *Educational Psychologist*, *38*(1), 43–52.

https://doi.org/10.1207/S15326985EP3801_6

- McCloskey, M., Caramazza, A., & Basili, A. (1985). Cognitive mechanisms in number processing and calculation: Evidence from dyscalculia. *Brain and Cognition*, 4(2), 171–196. Retrieved from http://www.sciencedirect.com/science/article/pii/0278262685900697
- McCrink, K., & Wynn, K. (2004). Large-Number Addition and Subtraction by 9-Month-Old Infants. *Psychological Science*, *15*(11), 776–781. https://doi.org/10.1111/j.0956-7976.2004.00755.x
- McCrink, K., & Wynn, K. (2007). Ratio abstraction by 6-month-old infants. *Psychological Science*, 18(8), 740–745. Retrieved from http://pss.sagepub.com/content/18/8/740.short
- McGurk, H. (1972). Infant discrimination of orientation. *Journal of Experimental Child Psychology*, *14*(1), 151–164. Retrieved from http://www.sciencedirect.com/science/article/pii/0022096572900409
- McNeil, N. M., & Jarvin, L. (2007). When Theories Don't Add Up: Disentangling the Manipulatives Debate. *Theory into Practice*, *46*(4), 309–316. https://doi.org/10.1080/00405840701593899
- Meyer, M. L., Salimpoor, V. N., Wu, S. S., Geary, D. C., & Menon, V. (2010). Differential contribution of specific working memory components to mathematics achievement in 2nd and 3rd graders. *Learning and Individual Differences*, *20*(2), 101–109. https://doi.org/10.1016/j.lindif.2009.08.004
- Miller, G. A. (1956). The magical number seven, plus or minus two: some limits on our capacity for processing information. *Psychological Review*, *63*(2), 81–97. https://doi.org/10.1037/h0043158
- Miura, I. T., & Okamoto, Y. (1989). Comparisons of US and Japanese First Graders' Cognitive Representation of Number and Understanding of Place Value. *Journal of Educational Psychology*, *81*(1), 109–113. Retrieved from http://psycnet.apa.org/journals/edu/81/1/109/

Miura, I. T., Okamoto, Y., Kim, C. C., Steere, M., & Fayol, M. (1993). First graders'

cognitive representation of number and understanding of place value: Crossnational comparisons: France, Japan, Korea, Sweden, and the United States. *Journal of Educational Psychology*, *85*(1), 24. Retrieved from http://psycnet.apa.org/journals/edu/85/1/24/

- Mix, K. S., Huttenlocher, J., & Levine, S. C. (2002). Quantitative development in infancy and early childhood. Oxford University Press. Retrieved from http://www.worldcat.org/title/math-without-words-quantitative-development-ininfancy-and-early-childhood/oclc/46601951
- Mix, K. S., Prather, R. W., Smith, L. B., & Stockton, J. D. (2014). Young Children's Interpretation of Multidigit Number Names: From Emerging Competence to Mastery. *Child Development*, *85*(3), 1306–1319. https://doi.org/10.1111/cdev.12197
- Mix, K. S., Smith, L. B., Stockton, J. D., Cheng, Y.-L., Barterian, J. A., Cheng, Y.-L., & Barterian, J. A. (2016). Grounding the Symbols for Place Value:
 Evidence From Training and Long-Term Exposure to Base-10 Models. *Journal of Cognition and Development*, *0*(0), 1–23.
 https://doi.org/10.1080/15248372.2016.1180296
- Moeller, K., Pixner, S., Zuber, J., Kaufmann, L., & Nuerk, H.-C. (2011). Early placevalue understanding as a precursor for later arithmetic performance - A longitudinal study on numerical development. *Research in Developmental Disabilities*, *32*(5), 1837–1851. https://doi.org/10.1016/j.ridd.2011.03.012
- Mousavi, S. Y., Low, R., & Sweller, J. (1995). Reducing cognitive load by mixing auditory and visual presentation modes. *Journal of Educational Psychology*, *87*(2), 319. Retrieved from http://psycnet.apa.org/journals/edu/87/2/319/
- Muller, K. (2010). Evolutionary Educational Psychology : The Disparity Between
 How Children Want to Learn and How They Are Being Taught. *EvoS Journal: The Journal of the Evolutionary Studies Consortium*, 2(1), 12–23.
- Mwangi, W., & Sweller, J. (1998). Learning to solve compare word problems: The effect of example format and generating self-explanations. *Cognition and*

Instruction, 16(2), 173–199. https://doi.org/10.1207/s1532690xci1602_2

NAP - Results. (n.d.). Retrieved from

http://reports.acara.edu.au/Home/Results#results

- NAPLAN. (2016). NAP Minimum standards numeracy. Retrieved December 10, 2016, from https://www.nap.edu.au/naplan/numeracy/minimum-standards
- Nataraj, M. S., & Thomas, M. O. J. (2007). Developing the Concept of Place Value, 2, 523–532.
- National Curriculum Board. (2009). *Shape of the Australian Curriculum: Mathematics*. Commonwealth of Australia.
- NESA. (2015). NSW Education Standards Authority (Mathematics K-12). Retrieved January 5, 2017, from

http://syllabus.nesa.nsw.edu.au/mathematics/mathematics-k10/

Newcombe, N., Huttenlocher, J., & Learmonth, A. (1999). Infants' Coding of Location in Continuous Space. *Infant Behavior and Development*, *22*(4), 483– 510. Retrieved from

http://www.sciencedirect.com/science/article/pii/S0163638300000114

- Ngan Ng, S. S., & Rao, N. (2010). Chinese Number Words, Culture, and Mathematics Learning. *Review of Educational Research*, *80*(2), 180–206. https://doi.org/10.3102/0034654310364764
- Nuerk, H.-C., Moeller, K., Klein, E., Willmes, K., & Fischer, M. H. (2011). Extending the Mental Number Line. *Zeitschrift F{ü}r Psychologie*, *219*(1), 3–22. https://doi.org/10.1027/2151-2604/a000041
- Olive, T. (2003). Working Memory in Writing: Empirical Evidence From the Dual-Task Technique. *European Psychologist*, *9*(1), 32–42. Retrieved from http://www.psycontent.com/index/A2T8481T0054982K.pdf
- Paas, F., Renkl, A., & Sweller, J. (2003). Cognitive load theory and instructional design: Recent developments. *Educational Psychologist*, *38*(1), 1–4. Retrieved from http://www.tandfonline.com/doi/pdf/10.1207/S15326985EP3801_1

Paas, F., & Sweller, J. (2011). An Evolutionary Upgrade of Cognitive Load Theory:

Using the Human Motor System and Collaboration to Support the Learning of Complex Cognitive Tasks. *Educational Psychology Review*, *24*(1), 27–45. https://doi.org/10.1007/s10648-011-9179-2

- Paas, F., Tuovinen, J. E., Tabbers, H., & Van Gerven, P. W. M. (2003). Cognitive Load Measurement as a Means to Advance Cognitive Load Theory. *Educational Psychologist*, *38*(1), 63–71.
- Passolunghi, M. C., Mammarella, I. C., & Altoè, G. (2008). Cognitive Abilities as Precursors of the Early Acquisition of Mathematical Skills During First Through Second Grades. *Developmental Neuropsychology*, *33*(3), 229–250. https://doi.org/10.1080/87565640801982320
- Passolunghi, M. C., & Siegel, L. S. (2001). Short-Term Memory, Working Memory, and Inhibitory Control in Children with Difficulties in Arithmetic Problem Solving. *Journal of Experimental Child Psychology*, *80*(1), 44–57. https://doi.org/10.1006/jecp.2000.2626
- Passolunghi, M. C., & Siegel, L. S. (2004). Working memory and access to numerical information in children with disability in mathematics. *Journal of Experimental Child Psychology*, *88*(4), 348–367. https://doi.org/10.1016/j.jecp.2004.04.002
- Passolunghi, M. C., Vercelloni, B., & Schadee, H. (2007). The precursors of mathematics learning: Working memory, phonological ability and numerical competence. *Cognitive Development*, *22*(2), 165–184. https://doi.org/10.1016/j.cogdev.2006.09.001
- Peucker, S., & Weißhaupt, S. (2013). Development of Numerical Concepts. *South African Journal of Childhood Education*, *3*(1), 21–38.
- Pollock, E., Chandler, P., & Sweller, J. (2002). Assimilating complex information. *Learning and Instruction*, *12*, 61–86.
- Raghubar, K. P., Barnes, M. A., & Hecht, S. A. (2010). Learning and Individual Differences. *Learning and Individual Differences*, *20*(2), 110–122. https://doi.org/10.1016/j.lindif.2009.10.005

- Reif, F., & Heller, J. I. (1982). Knowledge structure and problem solving in physics. *Educational Psychologist*, *17*(2), 102–127. Retrieved from http://www.tandfonline.com/doi/abs/10.1080/00461528209529248
- Renkl, A., & Atkinson, R. K. (2003). Structuring the Transition From Example Study to Problem Solving in Cognitive Skill Acquisition: A Cognitive Load Perspective. *Educational Psychologist*, *38*(1), 15–22. https://doi.org/10.1207/S15326985EP3801_3
- Resnick, L. B. (1984). A developmental theory of number understanding. The development of mathematical thinking.
- Ross, S. H. (1989). Parts, Wholes, and Place Value: A Developmental View. *Arithmetic Teacher, 36*(6), 47–51. Retrieved from http://www.eric.ed.gov/ERICWebPortal/recordDetail?accno=EJ389487
- Rouder, J. N., & Geary, D. C. (2014). Children's cognitive representation of the mathematical number line. *Developmental Science*, *17*(4), 525–536. https://doi.org/10.1111/desc.12166
- Rourke, A., & Sweller, J. (2009). The Worked-Example Effect Using III-Defined
 Problems: Learning to Recognise Designers' Styles. *Learning and Instruction*, *19*(2), 185–199. https://doi.org/10.1016/j.learninstruc.2008.03.006
- Rumelhart, D. E. (1980). Schemata: The Building Blocks of Cognition. In R. Anderson & R. J. Spiro (Eds.), *Theoretical Issues in Reading Comprehension* (pp. 33–58). Hillsdale, N.J.: Lawrence Erlbaum Associates. Retrieved from http://scholar.google.com/scholar?q=related:NAnfmf5Ga_0J:scholar.google.co m/&hl=en&num=20&as_sdt=0,5
- Rumelhart, D. E., & Ortony, A. (1977). The Representation of Knowledge in Memory. In R. Anderson, R. J. Spiro, & W. E. Montague (Eds.), *Schooling and the Acquisition of Knowledge* (p. 61). Hillsdale, N.J.: Lawrence Erlbaum Associates. Retrieved from
 - http://www.cs.northwestern.edu/~ortony/Andrew_Ortony_files/Rumelhart and Ortony.pdf

- Schmittau, J., & Vagliardo, J. J. (2006). Using concept mapping in the development of the concept of positional system. *Concept Maps: Theory, Methodology, Technology, CMC 2006*, 590–597. Retrieved from http://cmc.ihmc.us/cmc2006papers/cmc2006-p91.pdf
- Schneider, W., & Shiffrin, R. M. (1977). Controlled and automatic human information processing: I. Detection, search, and attention. *Psychological Review*, 84(1), 1. Retrieved from http://psycnet.apa.org/journals/rev/84/1/1/
- Schwartz, M., Day, R. H., & Cohen, L. B. (1979). Visual Shape Perception in Early Infancy. *Monographs of the Society for Research in Child Development*, 44(7), 1–63. Retrieved from http://psycnet.apa.org/psycinfo/1980-05200-001
- Sciama, S. C., Semenza, C., & Butterworth, B. (1999). Repetition priming in simple addition depends on surface form and typicality. *Memory {&} Cognition, 27*(1), 116–127. Retrieved from http://link.springer.com/article/10.3758/BF03201218
- Siegel, L. S., & Ryan, E. B. (1989). The Development of Working Memory in Normally Achieving and Subtypes of Learning Disabled Children. *Child Development*, *60*(4), 973–980. Retrieved from http://www.jstor.org/stable/10.2307/1131037
- Siegler, R. S., & Opfer, J. E. (2003). The Development of Numerical Estimation: Evidence for Multiple Representations of Numerical Quantity. *Psychological Science*, 14(3), 237–243. https://doi.org/10.1111/1467-9280.02438
- Simon, H. A., & Chase, W. G. (1973). Skill in chess: Experiments with chessplaying tasks and computer simulation of skilled performance throw light on some human perceptual and memory processes. *American Scientist*. Retrieved from http://www.jstor.org/stable/10.2307/27843878
- Simon, H. A., & Gilmartin, K. (1973). A simulation of memory for chess positions. Cognitive Psychology, 5, 29–46. Retrieved from http://www.sciencedirect.com/science/article/pii/0010028573900248
- Simon, T. J., Hespos, S. J., & Rochat, P. (1995). Do infants understand simple arithmetic? A replication of Wynn (1992). *Cognitive Development*, *10*(2), 253–

269. Retrieved from

http://www.sciencedirect.com/science/article/pii/088520149590011X

Slater, A., Mattock, A., & Brown, E. (1990). Size constancy at birth: Newborn infants' responses to retinal and real size. *Journal of Experimental Child Psychology*, 49(2), 314–322. Retrieved from http://www.sciencedirect.com/science/article/pii/002209659090061C

Slater, A., Mattock, A., Brown, E., & Bremner, J. G. (1991). Form Perception at

Birth: Revisited. *Journal of Experimental Child Psychology*, *51*(3), 395–406. Retrieved from

http://www.sciencedirect.com/science/article/pii/0022096591900846

 Slater, A., Morison, V., Somers, M., Mattock, A., Brown, E., & Taylor, D. (1990).
 Newborn and older infants' perception of partly occluded objects. *Infant Behavior and Development*, *13*(1), 33–49. Retrieved from http://www.sciencedirect.com/science/article/pii/016363839090004R

Smith, S. S. (2008). *Early Childhood Mathematics* (4th ed.). Pearson.

- Sowell, E. J. (1989). Effects of Manipulative Materials in Mathematics Instruction. *Journal for Research in Mathematics Education*, *20*(5), 498–505. Retrieved from http://www.jstor.org.wwwproxy0.library.unsw.edu.au/stable/749423
- Spanjers, I. A. E., Gog, T., & van Merriënboer, J. J. G. (2011). Segmentation of Worked Examples: Effects on Cognitive Load and Learning. *Applied Cognitive Psychology*, *26*(3), 352–358. https://doi.org/10.1002/acp.1832
- Spelke, E., Lee, S. A., & Izard, V. (2010). Beyond Core Knowledge: Natural Geometry. *Cognitive Psychology*, *34*(5), 863–884. https://doi.org/10.1111/j.1551-6709.2010.01110.x
- Spelke, E. S. (2005). Sex Differences in Intrinsic Aptitude for Mathematics and Science?: A Critical Review. *American Psychologist*, 60(9), 950–958. https://doi.org/10.1037/0003-066X.60.9.950
- St Clair Thompson, H., Stevens, R., Hunt, A., & Bolder, E. (2010). Improving children's working memory and classroom performance. *Educational*

Psychology, 30(2), 203-219. https://doi.org/10.1080/01443410903509259

Starkey, P., & Cooper, R. G. (1980). Perception of numbers by human infants. *Science*. Retrieved from http://psycnet.apa.org/psycinfo/1981-27833-001

- Starkey, P., Spelke, E., & Gelman, R. (1983). Detection of Intermodal Numerical Correspondences by Human Infants. *Science*, *222*, 179–181.
- Steffe, P. L., Cobb, P., & von Glasersfeld, E. (1988). *Construction of arithmetical meanings and strategies*. New York: Springer-Verlag.
- Stein, M. K., & Bovalino, J. W. (2001). Reflections on Practice: Manipulatives: One Piece of the Puzzle. *Mathematics Teaching in the Middle School*, *6*(6), 356–359. Retrieved from

http://www.jstor.org.wwwproxy0.library.unsw.edu.au/stable/41180973

Sternberg, R. J., & Ben-Zeev, T. (2012). *The Nature of Mathematical Thinking*. Taylor & Francis. Retrieved from https://books.google.com.au/books?id=q7F777rDI1AC

Stevenson, A. (Ed.). (2010). Oxford Dictionary of English. OUP Oxford. Retrieved

from https://books.google.com.au/books?id=anecAQAAQBAJ

Stevenson, H. W., Chen, C., & Lee, S. Y. (1993). Mathematics Achievement of Chinese, Japanese and American Children: Ten Years Later. *Current Directions in Psychological Science*, *259*, 53–58.

Strauss, M. S., & Curtis, L. E. (1981). Infant perception of numerosity. *Child Development*, 1146–1152. Retrieved from http://www.jstor.org/stable/10.2307/1129500

Swan, P., & Marshall, L. (2010). Revisiting Mathematics Manipulative Materials. *Australian Primary Mathematics Classroom*, *15*(2), 13–19.

Swanson, H. L., & Beebe-Frankenberger, M. (2004). The Relationship Between Working Memory and Mathematical Problem Solving in Children at Risk and Not at Risk for Serious Math Difficulties. *Journal of Educational Psychology*, *96*(3), 471–491. https://doi.org/10.1037/0022-0663.96.3.471

Swanson, H. L., & Sachse-Lee, C. (2001). Mathematical Problem Solving and

Working Memory in Children with Learning Disabilities: Both Executive and Phonological Processes Are Important. *Journal of Experimental Child Psychology*, *79*(3), 294–321. https://doi.org/10.1006/jecp.2000.2587

Sweller, J. (1988). Cognitive Load During Problem Solving: Effects on Learning. *Cognitive Psychology*, *12*(2), 257–285. Retrieved from http://www.sciencedirect.com/science/article/pii/0364021388900237

- Sweller, J. (1989). Cognitive Technology: Some Procedures for Facilitating Learning and Problem Solving in Mathematics and Science. *Journal of Educational Psychology*, *81*(4), 457–466.
- Sweller, J. (1999). *Instructional Design in Technical Areas*. Camberwell, Victoria: ACER.
- Sweller, J. (2002). Cognitive Load Theory, Learning Difficulty, and Instructional Design, 1–18.
- Sweller, J. (2003). Evolution of Human Cognitive Architecture. *Psychology of Learning and Motivation, 43*, 216–266. Retrieved from http://books.google.com/books?hl=en&Ir=&id=o-IXJUHzDFsC&oi=fnd&pg=PA215&dq=EVOLUTION+OF+HUMAN+COGNITIV E+ARCHITECTURE&ots=C_OPOgVbfa&sig=NxztytUXEu2_d97mv6NLqRXI1 8Y

Sweller, J. (2004). Instructional Design Consequences of an Analogy between Evolution by Natural Selection and Human Cognitive Architecture. *Instructional Science*, *32*, 9–31. Retrieved from

http://www.springerlink.com/index/t42495336q677725.pdf

Sweller, J. (2008). Instructional Implications of David C. Geary's Evolutionary Educational Psychology. *Educational Psychologist*, *43*(4), 214–216. https://doi.org/10.1080/00461520802392208

Sweller, J. (2010). Element Interactivity and Intrinsic, Extraneous, and Germane Cognitive Load. *Educational Psychology Review*, *22*(2), 123–138. https://doi.org/10.1007/s10648-010-9128-5 Sweller, J. (2011). *Cognitive Load Theory* (Vol. 55). Elsevier Inc. https://doi.org/10.1016/B978-0-12-387691-1.00002-8

- Sweller, J. (2012). Human cognitive architecture: Why some instructional procedures work and others do not. In *APA educational psychology handbook, Vol 1: Theories, constructs, and critical issues.* (pp. 295–325). Washington: American Psychological Association. https://doi.org/10.1037/13273-011
- Sweller, J., Ayres, P., & Kalyuga, S. (2011). *Cognitive Load Theory*. Springer. Retrieved from

http://books.google.com.au/books?id=sSAwbd8qOAAC&printsec=frontcover& dq=Cognitive+Load+Theory&hl=&cd=1&source=gbs_api

- Sweller, J., & Chandler, P. (1994). Why Some Material Is Difficult to Learn. Cognition and Instruction, 12(3), 185–233. https://doi.org/10.1207/s1532690xci1203_1
- Sweller, J., & Cooper, G. A. (1985). The Use of Worked Examples as a Substitute for Problem Solving in Learning Algebra. *Cognition and Instruction*, *2*(1), 59–89. https://doi.org/10.1207/s1532690xci0201_3
- Sweller, J., van Merriënboer, J. J. G., & Paas, F. (1998). Cognitive architecture and instructional design. *Educational Psychology Review*, *10*(3), 251–296. Retrieved from http://www.lavoisier.fr/livre/notice.asp?ouvrage=1103657
- Takir, A. (2012). The Effect of an Instruction Designed by Cognitive Load Theory Principles on 7th Grade Students' Achievement in Algebra Topics and Cognitive Load. *Creative Education*, *3*(2), 232–240. https://doi.org/10.4236/ce.2012.32037
- Tarmizi, R. A., & Sweller, J. (1988). Guidance During Mathematical Problem Solving. *Journal of Educational Psychology*, *80*(4), 424–436. Retrieved from http://psycnet.apa.org/journals/edu/80/4/424/
- Thevenot, C., & Barrouillet, P. (2006). Encoding numbers: Behavioral evidence for processing-specific representations. *Memory* {&} Cognition, 34(4), 938–948. Retrieved from http://link.springer.com/article/10.3758/BF03193439

Thorndyke, P. W., & Hayes-Roth, B. (1977). The Use of Schemata in the Acquisition and Transfer of Knowledge. *Cognitive Psychology*, *11*, 82–106. Retrieved from

http://www.sciencedirect.com/science/article/pii/0010028579900057

- Tindall-Ford, S., Chandler, P., & Sweller, J. (1997). When Two Sensory Modes Are Better Than One. *Journal of Experimental Psychology: Applied*, *3*(4), 257–287. Retrieved from http://psycnet.apa.org/journals/xap/3/4/257/
- Tulving, E. (1972). Episodic and Semantic Memory . In E. Tulving (Ed.), Organization of Memory (pp. 381–403). New York: Academic Press. Retrieved from http://web.media.mit.edu/~jorkin/generals/papers/Tulving_memory.pdf
- Uttal, D. H., Scudder, K. V, & DeLoache, J. S. (1997). Manipulatives as symbols: A new perspective on the use of concrete objects to teach mathematics. *Journal* of Applied Developmental Psychology, 18(1), 37–54. https://doi.org/10.1016/S0193-3973(97)90013-7
- van der Sluis, S., van der Leij, A., & de Jong, P. F. (2005). Working Memory in Dutch Children with Reading- and Arithmetic-Related LD. *Journal of Learning Disabilities*, *38*(3), 207–221. https://doi.org/10.1177/00222194050380030301
- van Gog, T., Kester, L., & Paas, F. (2011). Contemporary Educational Psychology. *Contemporary Educational Psychology*, *36*(3), 212–218. https://doi.org/10.1016/j.cedpsych.2010.10.004
- van Gog, T., & Rummel, N. (2010). Example-Based Learning: Integrating Cognitive and Social-Cognitive Research Perspectives. *Educational Psychology Review*, *22*(2), 155–174. https://doi.org/10.1007/s10648-010-9134-7
- van Loon-Hillen, N., van Gog, T., & Brand-Gruwel, S. (2012). Effects of worked examples in a primary school mathematics curriculum. *Interactive Learning Environments*, *20*(1), 89–99. https://doi.org/10.1080/10494821003755510
- Van Loosbroek, E., & Smitsman, A. W. (1990). Visual perception of numerosity in infancy. *Developmental Psychology*, *26*(6), 916. Retrieved from http://psycnet.apa.org/journals/dev/26/6/916/

van Merriënboer, J. J. G., & Ayres, P. (2005). Research on Cognitive Load Theory and Its Design Implications for E-Learning. *Educational Technology Research and Development*, *53*(3), 5–13. Retrieved from http://link.springer.com/article/10.1007/BF02504793

van Merriënboer, J. J. G., Schuurman, J. G., De Croock, M. B. M., & Paas, F. (2002). Redirecting Learners' Attention During Training: Effects on Cognitive Load, Transfer Test Performance and Training Efficiency. *Learning and Instruction*, *12*(1), 11–37. Retrieved from

http://www.sciencedirect.com/science/article/pii/S0959475201000202

- van Merriënboer, J. J. G., & Sweller, J. (2005). Cognitive Load Theory and Complex Learning: Recent Developments and Future Directions. *Educational Psychology Review*, *17*(2), 147–177. https://doi.org/10.1007/s10648-005-3951-0
- van Merriënboer, J. J. G., & Sweller, J. (2010). Cognitive Load Theory in Health Professional Education: Design Principles and Strategies. *Medical Education*, *44*(1), 85–93. https://doi.org/10.1111/j.1365-2923.2009.03498.x
- Verhoeven, L., Schnotz, W., & Paas, F. (2009). Cognitive load in interactive knowledge construction. *Learning and Instruction*, *19*(5), 369–375. https://doi.org/10.1016/j.learninstruc.2009.02.002
- Von Aster, M. G., & Shalev, R. S. (2007). Number development and developmental dyscalculia. *Developmental Medicine {&} Child Neurology, 49*(11), 868–873. https://doi.org/10.1111/j.1469-8749.2007.00868.x
- Ward, M., & Sweller, J. (1990). Structuring Effective Worked Examples. *Cognition and Instruction*, *7*(1), 1–39. https://doi.org/10.1207/s1532690xci0701_1
- Wright, R. (2013). Assessing early numeracy: Significance, trends, nomenclature, context, key topics, learning framework and assessment tasks. *South African Journal of Childhood Education*, *3*(2), 21–40.
- Wynn, K. (1992). Addition and subtraction by human infants. *Nature*, *358*(6389), 749–750. Retrieved from

http://www.nature.com/nature/journal/v358/n6389/abs/358749a0.html

- Wynn, K., & Bloom, P. (1999). Individuation Without Objects: Implications for Developmental Psychology . In Paper presented at the Biennial Meeting of the Society for Research in Child Development. Albuquerque, NM.
- Xu, F., & Arriaga, R. I. (2010). Number discrimination in 10-month-old infants.
 British Journal of Developmental Psychology, 25(1), 103–108.
 https://doi.org/10.1348/026151005X90704
- Xu, F., & Spelke, E. S. (2000). Large number discrimination in 6-month-old infants.
 Cognition, 74(1), B1--B11. Retrieved from http://www.sciencedirect.com/science/article/pii/S0010027799000669
- Yeung, A. S., Jin, P., & Sweller, J. (1997). Cognitive Load and Learner Expertise: Split-Attention and Redundancy Effects in Reading with Explanatory Notes. *Contemporary Educational Psychology*, *23*(1), 1–21. Retrieved from http://www.sciencedirect.com/science/article/pii/S0361476X97909516
- Young-Loveridge, J. M. (2002). Early childhood numeracy: Building an understanding of part-whole relationships. *Australian Journal of Early Childhood*, *27*(4), 36–42. Retrieved from http://search.informit.com.au.wwwproxy0.library.unsw.edu.au/ODMzODgyOS4 2ODM3Mjc%3D/aeipt/122410.pdf
- Zhang, J., & Norman, D. A. (1995). A Representational Analysis of Numeration Systems. *Cognition*, *57*(3), 271–295. Retrieved from http://www.sciencedirect.com/science/article/pii/0010027795006743
- Zheng, X., Swanson, H. L., & Marcoulides, G. A. (2011). Journal of Experimental Child Psychology. *Journal of Experimental Child Psychology*, *110*(4), 481– 498. https://doi.org/10.1016/j.jecp.2011.06.001
- Zhu, X., & Simon, H. A. (1987). Learning Mathematics From Examples and by Doing. *Cognition and Instruction*, 4(3), 137–166. https://doi.org/10.1207/s1532690xci0403_1

Appendices

Appendix A: Experimental materials used in Experiment 1

Pre-test

Please complete the following counting sequences: D^{-} 1, 2, 3, 4, 5, ____, ____ 7, 8, 9, ____, ____, ____ 27, 28, 29, ____, ____, ____ 10, 20, 30, 40, ____, ____, ____ 23, 33, 43, ____, ____, ____ 145, 155, 165, ____, ____, ____ 2) Please answer the following questions: 7 + 3 = |5 - 3 = 8 + 6 = 12 + 10 = 22 + 8 = 10 + 4 = 35 - 31 = 35 - 15 = 10 + 25 =31 + 40 = 3) Please write these numbers as numerals:

Twenty three

Two hundred and three

Ninety nine

Seventeen

Seventy one
Seven hundred and twenty two
T en
Fifteen
4) Please write these numerals in words: 9
19
99
91
138
831
402
5) Please complete the following: 55 = - + 5 60 = 60 + -

102 = _ + _ + 2

- $230 = 200 + _ + _$ 6) What is the total value of the highlighted number? $93 = _$ $59 = _$ $102 = _$ $370 = _$ 7) What is the number of tens in 342? _ 8) What is the number of hundreds in 587? _
- 9) What is the number of ones in 324?

Post-test

The following numbers are to be written down to dictation: 6, 8, 3, 9, 7, 13, 28, 32, 57, 12, 21, D^{-} 65, 42, 88, 24, 301, 147, 741, 205, 502 2) Please complete the following counting sequences: 3, 4, 5, _, _, _ 17, 18, 19, _, _, _ 37, 38, 39, _, _, _ 20, 30, 40, _, _, _ 123, 133, 143, _, _, _ 175, 185, 195, _, _, _ 3) Please answer the following questions: 6 + 4 = 17 - 3 = 5 + 6 = 22 + 10 =52 + 8 = 10 + 9 = 37 - 32 = 38 - 18 = 10 + 15 =28 - 17 =

41 + 20 =4) Please write these numbers as numerals: Thirty three Four hundred and three _____ One hundred and ninety nine Nineteen Ninety one Four hundred and one Seven hundred and fifty five Two hundred and twenty two Twenty Sixteen 5) Please write these numerals in words: 9 _____ 19 _____ 99 91 _____ 15 _____ 51 | 27 _____ 721 502 _____ 808 _____ 6) Please complete the following: 77 = _ + 7 20 = 20 +_ |3 = _ + 3 104 = _ + _ + 4 100 + _ + _ 7) What is the total value of the highlighted number? 73 = _

- 56 = _ 306 = _ 52 = _ 220 = _ 8) What is the number of tens in 152? 9) What is the number of hundreds in 227?
- 10) What is the number of ones in 362?

Appendix B: Experimental materials used in Experiment 2

Pre-test

- 1) Write the following using expanded notation: $25 = \Box + \Box$ 2) Write the following using expanded notation: $905 = \Box + \Box + \Box$ 3) What is the face value of the digit 5 in the number 25? 4) Circle the numbers that have a face value of 6 in the tens. 106 6 63 66 616 675) What is the face to the face value of 6 in the tens.
- 5) What is the total value of the digit 9 in the number 39?
- 6) What is the total value of the digit 5 in the number 507?
- 7) What number is made up of 300, and 20 and 2?
- 8) I have 9 hundreds, and 9 ones, what number am 1?

Acquisition phase – abacus

Background:

Use a one hundred—frame with ten lines and ten beads on each of the lines. The abacus has all its beads in the 'off position on the right. To switch the beads on we will move them from the right to the left. Notice that one line has IO beads on it.

 $36 = \Box + \Box$



Question:

1) Please write the following using expanded notation:

Solution:



Question:

2) Now, use the worked example to solve this question, by writing the following using expanded notation:

3) Create a three-digit number that has a hundreds face value of 1, tens face value of 7, and a ones face value of 5





Question:

4) Now, use the worked example to solve this question. Create a three-digit number that has a ones face value of 9, a tens face value of 4, and a hundreds face value of 2.

5) What is the total value of the digit 5 in the number **259**?

Solution:



6) Now, use the worked example to solve this question. What is the total value of the digit 4 in the number 843?

Question:

7) I have I hundreds and 6 tens, what number am 1?

Solution:

Use place value to construct numbers. I hundreds means that there are I lots of 100, which is equal to 100 ones. In addition 6 tens means that there are 60 ones.	
So a number that has I hundreds and 6 tens, is made up 100 + 60, which is 160	

Question:

8) Now, use the worked example to solve this question. I have 8 tens and 2 ones, what number am 1?

Acquisition phase – MAB

Background:



1) Please write the following using expanded notation:

$$36 = \Box + \Box$$

Solution:



Question:

2) Now, use the worked example to solve this question, by writing the following using expanded notation:

Question:

3)

Create a three—digit number that has a hundreds face value of 1 , tens face value of 7 , and a ones face value of 5 .

Solution:



Question:

4) Now, use the worked example to solve this question. Create a three-digit number that has a ones face value of 9, a tens face value of 4, and a hundreds face value of 2.

Question:

5) What is the total value of the digit 5 in the number **259**?



Question:

6) Now, use the worked example to solve this question. What is the total value of the digit 4 in the number 843?

Question: 7)

I have I hundreds and 6 tens, what number am I?

Solution:



Question:

8) Now, use the worked example to solve this question. I have 8 tens and 2 ones, what number am 1?

Acquisition phase - no visuals

```
      Question:
      1)
      Please write the following using expanded notation:

      36 = \Box + \Box

      Solution:

      Use place value to construct numbers.

      The number 36 is made up of 3 tens which is 30

      ones. In addition there are 6 ones.

      So to expand 36, we will write:

      36 = 30 + 6
```

2)

Now, use the worked example to solve this question, by writing the following using expanded notation:

47 = 🗌 + 🗌

Question:

3) Create a three-digit number that has a hundreds face value of 1, tens face value of 7, and a ones face value of 5.

Solution:

Use place value to construct numbers. We need a number that has a hundreds face value of 1, which is 100 ones. In addition we need a tens face value of 7, which is 70 ones, and a ones face value of 5.	
So a number that has a hundreds face value of 1, a tens face value of 7, and a ones face value of 5, is a number that is made of 100, 70 and 5, equal to 175.	

Question:

4) Now, use the worked example to solve this question. Create a three-digit number that has a ones face value of 9, a tens face value of 4, and a hundreds face value of 2.

Question:

5) What is the total value of t	the digit 5 in the number 259 ?
---------------------------------	--

Solution:

Use place value to construct numbers.	
The number 259 is made up of 2 hundreds, which is equal to 200 ones; 5 tens equal to 50 ones, and 9 ones.	
So the total value of 5 in the number 259, is 5 tens, which is equal to 50.

Question:

6) Now, use the worked example to solve this question. What is the total value of the digit 4 in the number 843?

Question:

7)

I have I hundreds and 6 tens, what number am 1?

Solution:

Use place value to construct numbers.	
which is equal to 100 ones. In addition 6 tens means that there are 60 ones.	
So a number that has I hundreds and 6 tens, is made up 100 + 60, which is 160	

Question:

8) Now, use the worked example to solve this question. I have 8 tens and 2 ones, what number am 1?

Post-test

- 1) Write the following using expanded notation: $25 = \Box + \Box$
- 2) Write the following using expanded notation:
 905 = □ + □ + □
- 3) What is the face value of the digit 5 in the number 25?
- 4) Circle the numbers that have a face value of 6 in the tens. 106
 6
 63
 66
 616
 67
- 5) What is the total value of the digit 9 in the number **39**?

- 6) What is the total value of the digit 5 in the number $\mathbf{507}$?
- 7) What number is made up of 300, and 20 and 2?
- 8) I have 9 hundreds, and 9 ones, what number am 1?

Appendix C: Experimental materials used in Experiment 3

Pre-test

- 1) Expand:
 - 25 = ____ + ____
 - 905 = ____ + ____ + ____
- 2) State the value of the bold number in 25

3) Circle the numbers that have 6 in the tens (there can be more than one number!).

106 6 63 66 616 67

4) State the value of the bold number in 939

5) State the value of the bold number in 507

- 6) What number is made up of 300, and 20 and 2?
- 7) I have 9 hundreds, and 9 ones, what number am 1?

Acquisition phase – abacus

1) Expand the numbers. The first one has been done for you.

36 = 30 + 6



 $2\,)$ Guess what the numbers are. The first one has been done for you.





 $3\,)$ State the value of the bold number. The first one has been done for you

259

The value of the number 5 is <u>50</u>



| have | hundreds and 6 tens, what number am 1? | am <u>| 60</u>



Acquisition phase – MAB





 $2\,)$ Guess what the numbers are. The first one has been done for you.

What is a three-digit number that has a Hundreds face value of 1: Tens face value of 7: Ones face value of 5

The number is 175



What is a three-digit number that has a **Ones** face value of 9: **Tens** face value of 4: **Hundreds** face value of 2



3) State the value of the bold number. The first one has been done for you

259

The value of the number 5 is 50





 $4\,)$ Guess what the numbers are. The first one has been done for you.





Acquisition phase - no visuals

1) Expand the numbers. The first one has been done for you.

36 = 30 + 6 47 = ____ + ____

 $2\,)$ Guess what the numbers are. The first one has been done for you.

What is a three-digit number that has a Hundreds face value of 1: Tens face value of 7: Ones face value of 5

The number is 175

What is a three-digit number that has a **Ones** face value of 9; **Tens** face value of 4: **Hundreds** face value of 2

The number is _____

3) State the value of the bold number. The first one has been done for you

259

The value of the number 5 is 50

342

The value of the number 4 is _____

4) Guess what the numbers are. The first one has been done for you.

I have I hundreds and 6 tens, what number am 1?

1 am <u>160</u>

I have 3 hundreds and 7 ones, what number am 1?

1 am _____

Post-test

1) Expand:

72 = ____ + ____

- 380 = ____ + ____ + ____
- 2) State the value of the bold number in $\mathbf{37}$

- 3) Circle the numbers that have 3 in the ones (there can be more than one number!).
 - 136 36 63 273 532 13
- 4) State the value of the bold number in 249
- 5) State the value of the bold number in ${\it q}$ 3.2
- 6) What number is made up of 500, and 60 and 7?
- 7) I have 3 hundreds, and 4 ones, what number am 1?

Appendix D: Experimental materials used in Experiment 4

Pre-test

9) Circle the numbers that have 6 in the tens (there can be more than one number!).										
	106	6	63	66	616	67				
10)	Circle the nu	mbers that have I	in the hundreds	(there can be more t	han one number!).					
	106	1016	16	2166	616	61				
11)	1)What number is made up of 300, and 20 and 2?									

[2) | have 9 hundreds, and 9 ones, what number am [?

13)Expand:

25 = ____ tens + ____ ones

14)Expand:

931 = ____ + ____ +

Acquisition phase - MAB

1) What number am 1? I have a 5 in the ones column, the number of tens is 7 and there are 2 in the hundreds column.

I have 2 in the hundreds,

7 in the tens,



2) Expand the number 95:



95 = 90 + 5

3) Expand the number 342:

342 = 3 hundreds







342 = 300 + 40 + 2

4) What number am 1? I have a 3 in the ones column. The number in the tens column is 4. I have I hundreds.

I have I in the hundreds,



5) What number am 1? I have a 2 in the tens column, 7 in the ones column and I have 5 in my hundreds column.

I have 5 in the hundreds,

· · ·	_	_	_	_	_	_	_	n .	_	_		_	_	_
								М						
								M						
								M						
								Ш						
								N.						
								LI.						
								Ш			Γ			
Т								LI.			Г			Γ
								LI.						
								U						Γ
								_						

	_	_	_	_	_	_	_
Г							
Г		П					
Г		П					
Г		П	Т				
Г							
Г		П					
Г							

....

_	_	_			n						_	_	_	_	
				г	U										
-	-		-	⊢.	n	⊢	-	-	-	\vdash	-		-		-
				L	Ы										
					Ľ										
_	_		-	÷	h	-	-	-	-		-		-		-
				Γ.	IJ										
				-	п										
			_	L .	к	_			_		_		_		-
				Г	U										
			-	÷-	п	-	-	-	-		-				_
				L .	N										_
				г.	U										
-	-	-	-	⊢.	h	-	-	-	-	-	-	-	-	-	-
				L	ų										
				Γ.	Ľ										-
				_	-16										

2 in the tens

......

R



6) Expand the number 263:



263 = 200 + 60 + 3

Acquisition phase - no visuals

1) What number am 1? I have a 5 in the ones column, the number of tens is 7 and there are 2 in the hundreds column.

I have 2 in the hundreds, 7 in the tens 5 in the ones So I am 275

2) Expand the number 95:

95 has 9 tens and 5 ones, so

95 = 9 tens + 5 ones

3) Expand the number 342:

342 has 3 hundreds, 4 tens and 2 ones, so

342 = 3 hundreds + 4 tens + 2 ones

4) What number am 1? I have a 3 in the ones column. The number in the tens column is 4. I have I hundreds.

| have | in the hundreds 4 in the tens 3 in the ones So | am |43

5) What number am 1? I have a 2 in the tens column, 7 in the ones column and I have 5 in my hundreds column.

I have 5 in the hundreds, 2 in the tens 7 in the ones So I am 527 6) Expand the number 263:

263 has 2 hundreds, or 200, 6 tens or 60 and 3 ones or 3, so

263 = 200 + 60 + 3

Post-test

1) What number am 1? I have a 9 in the ones column, the number of tens is 2 and there are 8 in the hundreds column.

l am _____

- 2) What number am 1? I have a 7 in the ones column. The number in the tens column is 2. I have 2 hundreds.
 - 1 am _____

- 3) What number am 1? I have a 1 in the ones column. My hundreds column has 6 and 1 have 3 in my tens column.
- l am _____ 4) What number am 1? I have a 4 in my hundreds column, an 8 in my tens column and my ones column has 2. l am _____ 5) Expand: 34 = _____ + ____ 6) Expand: 4/6 = _____ + _____ + _____ 7) Expand: 999 = + + Post-test transfer 1) Expand: |7 = _____ tens + _____ ones 2) Expand: 8/9 = _____ hundreds + _____ tens + _____ ones 3) Expand: | 252 = _____ + ____ + ____ + ____ 4) Expand: 3/24 = _____ + ____ + ____ + ____ 5) Circle the numbers that have 2 in the hundreds (there can be more than one number!). 23 1226 142 219 2052 202

6) What number am 1? I have a 7 in my hundreds column. My tens column has 3 less than the hundreds. And my ones column has 2 less than the hundreds column.

l am _____

Delayed test

- What number am 1? I have a 4 in my hundreds column, an 8 in my tens column and my ones column has
 2.
- 1 am _______
 2) Expand:
 417 = ______ hundreds + ______ tens + _____ ones
 3) Expand:
 904 = ______ hundreds + ______ tens + _____ ones
 4) Expand:
 3419 = ______ + _____ + _____ + _____
 5) Expand:
 9999 = ______ + _____ + _____ + _____
 6) Circle the numbers that have 6 in the hundreds (there can be more than one number!).
 - 6266 602 1623 6006 4056 6606
- 7) What number am 1? I have an 8 in my hundreds column. My tens column has 2 less than the hundreds. And my ones column has 4 less than the hundreds column.
 - l am _____

Appendix E: Experimental materials used in Experiment 5

Pre-test

- 1) Circle the numbers that have 6 in the tens (there can be more than one number!).
 - 106 1 663 616 67

2) Circle the numbers that have 1 in the hundreds (there can be more than one number!).

1 016 2 166 611 161

3) Circle the numbers that have 7 in the thousands (there can be more than one number!).

7 3 1 6 7 1 777 1 732 1 7 3 5 6

- 4) What number is made up of 3 000, and 10 and 2?
- 5) I have 9 thousands, and 9 ones, what number am 1?
- 6) Arrange the following numbers in order from largest to smallest: 1 346, 942, 1 326, 924, 10 002, 9 009
- 7) Arrange the following numbers in order from smallest to largest: 10 987, 10 978, 1 724, 724, 6 002, 6 020

Acquisition phase – MAB

Expand the number 4 307:
 4 307 has

4 thousands (4 000)



and 7 ones (7)



2) Now solve a similar problem

Expand the number 8 031:

- 3) Expand the number 3 584
 - 3 584 has
 - 3 thousands (3 000)



5 hundreds (500)

Æ	11111	17	7
H		-	H
H		H	H
H		H	Ħ
T			
+			H
H		H	H
LT:		tt	Ħ

a –	4	2	<i>z _ z</i>	-		- 2	-	2	л.
4						т	L		И.
4				1		1	L		И.
4				-	_	+	1		и.
4	Ц			-	4	+	1		и.
4	Ц			-	-	+	-		и.
4	н			-	-	+	-		и.
1	н				Ц	+	1		и.
1	н				н	+	⊢	-	И.
1	Н	H	++	++	н	÷	÷	H	н
1					_	_	1		1

		_	e
i D		г	
i D			
i E			
i E			
i E			
i D			
i D			
i D			
1 0		1	
/ F			

- 4	c_{2}	-	<u> </u>	2	Z		
-[Т						L
-[1
-1							
- [
-[
- [
- [h
-[Т						_
- [_
- [7



8 tens (80)



0 0 0 0

This means: 3 584 = 3 000 + 500 + 80 + 4

4) Now solve a similar problem

Expand the number 9 929:

5) Expand the number 9 075 9 075 has 9 thousands (9 000)



6) Now solve a similar problem

Expand the number 7 380:

7) Expand the number 6 444

6 444 has

6 thousands (6 000)



This means: 6 + 4 + 4 = 6 000 + 400 + 40 + 4

8) Now solve a similar problem

Expand the number 3 888:

Acquisition phase - no visuals

- 1) Expand the number 4 307:
 - 4 307 has
 - 4 thousands (4 000)
 - 3 hundreds (300)
 - 0 tens (0)
 - and 7 ones (7)
 - This means: $4\ 307 = 4\ 000 + 300 + 0 + 7$
- Now solve a similar problem Expand the number 8 O31:
- 3) Expand the number 3 584
 - 3 584 has
 - 3 thousands (3 000)
 - 5 hundreds (500)
 - 8 tens (80)
 - and 4 ones (4)

This means: 3 584 = 3 000 + 500 + 80 + 4

- 4) Now solve a similar problem Expand the number 9 929:
- 5) Expand the number 9 075
 - 9 075 has
 - 9 thousands (9 000)

0 hundreds (0)
7 tens (70)
and 5 ones (5)
This means: 9 075 = 9 000 + 0 + 70 + 5
6) Now solve a similar problem Expand the number 7 380:
7) Expand the number 6 444
6 444 has
6 thousands (6 000)
4 hundreds (400)
4 tens (40)
and 4 ones (4)
This means: 6 444 = 6 000 + 400 + 40 + 4

 Now solve a similar problem Expand the number 3 888:

Post-test

1) Expand: 934 = _____ + ____ + ____ 2) Expand: 1 416 = _____ + ____ + ____ + ____ 3) Expand: 1 999 = _____ + ____ + ____ + ____ 4) Expand: 9 499 = _____ + ____ + ____ + ____

5)	Expand:					
	7 351 =	_ +	_ +	+		
6)	Expand:					
	4 099 =	_ +	_ +	+		
7)	Expand:					
	5 930 =	_ +	_ +	+		
8)	Expand:					
	2 024 =	_ +	_ +	+		
9)	Expand:					
	6 004 =	_ +	_ +	+		
10) Expand:					
	9 999 =	_ +	_ +	+		
Post-	test transfer					
1)	Expand:					
	5 8 9 =	_ thousands + .		_ hundreds +	tens +	ones
2)	Expand:					
	3 24 =	+	+	+	+	
3)	9 000 + 600 +	0 + 4 =				
4)	4 + 500 + 80 +	3 000 =				
5)	700 + 0 + 9 +	7 000 =				
6)	A number has 13 te	ns and 6 ones.	What is	the number?		

7) 5 thousands + 32 tens + 8 ones = _____

- 8) || thousands + 8 hundreds + 37 ones = _____
- 9) Fill in the missing number:

2088 + ____ = 2188

10) Fill in the missing number:

6 109 + ____ = 6 149

Delayed test

